Registration No. 333-

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-1 REGISTRATION STATEMENT

UNDER
THE SECURITIES ACT OF 1933

Monte Rosa Therapeutics, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

2836

(Primary Standard Industrial Classification Code Number)

84-3766197 (I.R.S. Employer Identification No.)

Monte Rosa Therapeutics, Inc. 645 Summer Street, Suite 102 Boston, MA 02210 (617) 949-2643

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Markus Warmuth, M.D.
President and Chief Executive Officer
Monte Rosa Therapeutics, Inc.
645 Summer Street, Suite 102
Boston, MA 02210
(617) 949-2643

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Robert E. Puopolo Marishka DeToy Goodwin Procter LLP 100 Northern Avenue Boston, Massachusetts 02210 (617) 570-1000 Nathan Ajiashvili Alison Haggerty Latham & Watkins LLP 885 Third Avenue New York, New York 10022 (212) 906-1200

Approximate date of commencement of proposed sale to the public: As soon as practicable after the effective date of this registration statement

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, as amended, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Securities Exchange Act of 1934.

 Large Accelerated filer
 □

 Non-accelerated filer
 □

 Smaller reporting company
 □

 Emerging growth company
 □

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

CALCULATION OF REGISTRATION FEE

	Proposed	
	maximum	
Title of each class of	aggregate	Amount of
securities to be registered	offering price(1)	registration fee
Common stock, \$0.0001 par value per share	\$	\$

(1) Estimated solely for the purpose of computing the registration fee in accordance with Rule 457(o) under the Securities Act. Includes the aggregate offering price of shares that the underwriters have the option to purchase to cover over-allotments, if any.

The registrant hereby amends this registration statement on such date or dates as may be necessary to delay its effective date until the registrant files a further amendment which specifically states that this registration statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act, or until the registration statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to said Section 8(a), may determine.

Explanatory note

This Amendment No. 1 ("Amendment No. 1") to the Draft Registration Statement ("Draft Registration Statement") is being filed solely for the purpose of filing Exhibits 10.14, 10.15, 10.16 and 10.17. This Amendment No. 1 does not modify any provisions of the prospectus that forms a part of the Draft Registration Statement and accordingly, such prospectus has been omitted.

Part II

Information not required in prospectus

Item 13. Other expenses of issuance and distribution

The following table sets forth the costs and expenses, other than underwriting discounts and commissions, to be paid by us in connection with the sale of the shares of common stock being registered hereby. All amounts shown are estimates except for the SEC registration fee, the FINRA filing fee and the Nasdaq Global Market initial listing fee.

SEC registration fee	\$ *
FINRA filing fee	*
Nasdaq listing fee	*
Printing and engraving expenses	*
Legal fees and expenses	*
Accounting fees and expenses	*
Blue Sky fees and expenses (including legal fees)	*
Transfer agent and registrar fees and expenses	*
Miscellaneous	*
Total	*

To be provided by amendment.

Item 14. Indemnification of directors and officers

Section 145 of the Delaware General Corporation Law (the DGCL) authorizes a corporation to indemnify its directors and officers against liabilities arising out of actions, suits and proceedings to which they are made or threatened to be made a party by reason of the fact that they have served or are currently serving as a director or officer to a corporation. The indemnity may cover expenses (including attorneys' fees) judgments, fines and amounts paid in settlement actually and reasonably incurred by the director or officer in connection with any such action, suit or proceeding. Section 145 permits corporations to pay expenses (including attorneys' fees) incurred by directors and officers in advance of the final disposition of such action, suit or proceeding. In addition, Section 145 provides that a corporation has the power to purchase and maintain insurance on behalf of its directors and officers against any liability asserted against them and incurred by them in their capacity as a director or officer, or arising out of their status as such, whether or not the corporation would have the power to indemnify the director or officer against such liability under Section 145.

We have adopted provisions in our certificate of incorporation and bylaws to be in effect upon the effectiveness of this registration statement that limit or eliminate the personal liability of our directors to the fullest extent permitted by the DGCL, as it now exists or may in the future be amended. Consequently, a director will not be personally liable to us or our stockholders for monetary damages or breach of fiduciary duty as a director, except for liability for:

- · any breach of the director's duty of loyalty to us or our stockholders;
- · any act or omission not in good faith or that involves intentional misconduct or a knowing violation of law;
- any unlawful payments related to dividends or unlawful stock purchases, redemptions or other distributions; or

any transaction from which the director derived an improper personal benefit.

These limitations of liability do not alter director liability under the federal securities laws and do not affect the availability of equitable remedies such as an injunction or rescission.

In addition, our bylaws to be in effect upon the effectiveness of this registration statement provide that:

- we will indemnify our directors, officers and, in the discretion of our board of directors, certain employees to the fullest extent permitted by the DGCL, as it now exists or may in the future be amended; and
- we will advance reasonable expenses, including attorneys' fees, to our directors and, in the discretion of our board of directors, to our officers and certain employees, in connection with legal proceedings relating to their service for or on behalf of us, subject to limited exceptions.

We intend to enter into indemnification agreements with each of our directors and executive officers. These agreements provide that we will indemnify each of our directors, certain of our executive officers and, at times, their affiliates to the fullest extent permitted by Delaware law. We will advance expenses, including attorneys' fees (but excluding judgments, fines and settlement amounts), to each indemnified director or executive officer in connection with any proceeding in which indemnification is available and we will indemnify our directors and officers for any action or proceeding arising out of that person's services as a director or officer brought on behalf of us or in furtherance of our rights. Additionally, certain of our directors or officers may have certain rights to indemnification, advancement of expenses or insurance provided by their affiliates or other third parties, which indemnification relates to and might apply to the same proceedings arising out of such director's or officer's services as a director referenced herein. Nonetheless, we have agreed in the indemnification agreements that our obligations to those same directors or officers are primary and any obligation of such affiliates or other third parties to advance expenses or to provide indemnification for the expenses or liabilities incurred by those directors are secondary.

We also maintain general liability insurance which covers certain liabilities of our directors and officers arising out of claims based on acts or omissions in their capacities as directors or officers, including liabilities under the Securities Act.

The underwriting agreement filed as Exhibit 1.1 to this registration statement provides for indemnification of us and our directors and officers by the underwriters against certain liabilities under the Securities Act and the Exchange Act.

Item 15. Recent sales of unregistered securities

In the three years preceding the filing of this registration statement, we have issued the following securities that were not registered under the Securities Act:

(a) Issuance of convertible promissory notes

In December 2019, we issued a convertible promissory note to an accredited investor in the principal amount of \$750,000.

No underwriters were involved in the foregoing sales of securities. The sales of securities described above were deemed to be exempt from registration pursuant to Section 4(a)(2) of the Securities Act, including Regulation D and Rule 506 promulgated thereunder, as transactions by an issuer not involving a public offering. All of the purchasers in this transaction represented to us in connection with their purchase that they were acquiring the securities for investment and not distribution, that they could bear the risks of the investment and could hold the securities for an indefinite period of time. Such purchasers received written disclosures that the securities

had not been registered under the Securities Act and that any resale must be made pursuant to a registration or an available exemption from such registration. All of the foregoing securities are deemed restricted securities for the purposes of the Securities Act.

(b) Issuances of capital stock

In April 2020 and September 2020, in connection with two separate Contribution and Exchange Agreements with the shareholders of Monte Rosa Therapeutics AG, we issued an aggregate of 5,000,000 shares of our common stock, 612,705 shares of our common stock in the form of restricted stock and 19,250,000 shares of our Series A convertible preferred stock to the shareholders of Monte Rosa Therapeutics AG, which included accredited investors, directors and employees.

Concurrent with the execution of the April 2020 Contribution and Exchange Agreement, we converted the entire principal amount of our outstanding convertible promissory note issued in December 2019 to an accredited investor, plus interest, into 754,280 shares of our Series A convertible preferred stock (for an aggregate issuance of 20,004,280 shares of Series A convertible preferred stock).

In April 2020, accredited investors purchased an aggregate of 9,627,234 shares of our Series A-2 convertible preferred stock at a price per share of \$1.2984.

In September 2020 and in February 2021, accredited investors purchased an aggregate of 48,000,000 shares of our Series B convertible preferred stock at a price per share of \$2.00.

In March 2021, accredited investors purchased an aggregate of 32,054,521 shares of our Series C convertible preferred stock at a price per share of \$2.9637.

No underwriters were involved in the foregoing sales of securities. The sales of securities described above were deemed to be exempt from registration pursuant to Section 4(a)(2) of the Securities Act, including Regulation D and Rule 506 promulgated thereunder, as transactions by an issuer not involving a public offering. All of the purchasers in these transactions represented to us in connection with their purchase that they were acquiring the securities for investment and not distribution, that they could bear the risks of the investment and could hold the securities for an indefinite period of time. Such purchasers received written disclosures that the securities had not been registered under the Securities Act and that any resale must be made pursuant to a registration or an available exemption from such registration. All of the foregoing securities are deemed restricted securities for the purposes of the Securities Act.

(c) Grants and exercises of stock options and restricted stock

As of March 31, 2021, we have granted stock options to purchase an aggregate of 7,786,146 shares of our common stock, with exercise prices ranging from \$0.32 to \$0.62 per share, to employees, directors and consultants. pursuant to 2020 Plan, and no shares of common stock have been issued upon the exercise of stock options pursuant to the 2020 Plan.

As of March 31, 2021, we have granted an aggregate of 1,250,446 shares of restricted stock to employees and consultants under the 2020 Plan and an additional 1,470,588 outside of the 2020 Plan.

The issuances of the securities under the 2020 Plan described above were deemed to be exempt from registration pursuant to Section 4(a)(2) of the Securities Act or Rule 701 promulgated under the Securities Act as transactions pursuant to compensatory benefit plans. The shares of common stock issued upon the exercise of options are deemed to be restricted securities for purposes of the Securities Act.

The issuance of securities described above to employees and consultants outside of the 2020 Plan were deemed exempt from registration pursuant to Section 4(a)(2) of the Securities Act as transactions by an issuer not involving a public offering.

Item 16. Exhibits and financial statement schedules

(a) Exhibits.

Exhibit number	Exhibit table
1.1*	Form of Underwriting Agreement
3.1**	Third Amended and Restated Certificate of Incorporation of the Registrant, as currently in effect
3.2*	Form of Fourth Amended and Restated Certificate of Incorporation of the Registrant (to be effective upon the closing of this offering)
3.3**	By-laws of the Registrant, as currently in effect
3.4*	Form of Amended and Restated By-laws (to be effective upon the closing of this offering)
4.1**	Second Amended and Restated Investors' Rights Agreement among the Registrant and certain of its stockholders, dated March 11, 2021
4.2*	Form of Common Stock Certificate
5.1*	Opinion of Goodwin Procter LLP
10.1**#	2020 Stock Option and Grant Plan, as amended, and forms of award agreements thereunder
10.2*#	2021 Stock Option and Incentive Plan and forms of award agreements thereunder
10.3*#	2021 Employee Stock Purchase Plan
10.4*#	Senior Executive Cash Incentive Bonus Plan
10.5*#	Form of Officer Indemnification Agreement
10.6*#	Form of Director Indemnification Agreement
10.7*#	Employment Agreement between the Registrant and Markus Warmuth, to be in effect upon the closing of this offering
10.8*#	Employment Agreement between the Registrant and Ajim Tamboli, to be in effect upon the closing of this offering
10.9*#	Employment Agreement between the Registrant and Owen Wallace, to be in effect upon the closing of this offering
10.10*#	Employment Agreement between the Registrant and Sharon Townson, to be in effect upon the closing of this offering
10.11*#	Employment Agreement between the Registrant and John Castle, to be in effect upon the closing of this offering
10.12*	Contribution and Exchange Agreement, dated April 14, 2020, between certain shareholders of Monte Rosa Therapeutics AG and the Registrant
10.13*	Contribution and Exchange Agreement, dated September 1, 2020, between certain shareholders of Monte Rosa Therapeutics AG and the Registrant
10.14	Services Agreement, dated as of April 10, 2018, between Ridgeline Therapeutics GmbH and Monte Rosa Therapeutics AG
10.15	Services Agreement, dated as of December 29, 2020, between Monte Rosa Therapeutics AG and the Registrant
10.16	License Agreement, dated as of April 10, 2018, among Cancer Research Technology Limited, The Institute of Cancer Research: Royal Cancer Hospital and Monte Rosa Therapeutics AG

Exhibit number	Exhibit table
10.17	Collaboration and Option Agreement, among Cancer Research Technology Limited, The Institute of Cancer Research: Royal Cancer Hospital and Monte Rosa Therapeutics AG, as amended on February 25, 2019, January 20, 2020 and June 18, 2020
10.18**	Lease Agreement, dated September 23, 2020, between OPG MP Parcel Owner (DE) LLC and the Registrant
21.1**	Subsidiaries of the Registrant
23.1*	Consent of Deloitte & Touche LLP, Independent Registered Public Accounting Firm
23.2*	Consent of Goodwin Procter LLP (included in Exhibit 5.1)
24.1*	Power of Attorney (included on signature page to this registration statement)

- * To be filed by amendment.
- ** Previously filed.
- # Indicates a management contract or any compensatory plan, contract or arrangement.
- † Portions of this exhibit (indicated by asterisks) will be omitted in accordance with the rules of the SEC.
- (b) Financial Statement Schedules.

None.

Item 17. Undertakings

The undersigned Registrant hereby undertakes to provide to the underwriters at the closing specified in the Underwriting Agreement certificates in such denominations and registered in such names as required by the underwriters to permit prompt delivery to each purchaser.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

The undersigned Registrant hereby undertakes that:

- (i) For purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the Registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective.
- (ii) For the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Signatures

Pursuant to the requirements of the Securities Act, Monte Rosa Therapeutics, Inc. has duly caused this registration statement on Form S-1 to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Boston, Commonwealth of Massachusetts, on the day of , 2021.

Monte	Rosa	Therapeu	itics	Inc
MOHILE	rusa	HIICIANCU	เนษอะ	IIIC.

Ву:

Markus Warmuth
President and Chief Executive Officer

Signatures and power of attorney

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints Markus Warmuth and Ajim Tamboli, and each of them, either of whom may act without the joinder of the other, as his true and lawful attorneys-in-fact and agents with full power of substitution and re-substitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign any and all amendments (including post-effective amendments) to this registration statement, and to sign any registration statement for the same offering covered by the registration statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act, and all post-effective amendments thereto, and to file the same, with all exhibits thereto and all documents in connection therewith, with the SEC, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or any of them, or his or their substitute or substitutes, may lawfully do or cause to be done or by virtue hereof.

Pursuant to the requirements of the Securities Act, this registration statement has been signed by the following persons in the capacities indicated on the day of , 2021.

Signature	Title
Markus Warmuth	President, Chief Executive Officer and Director (Principal Executive Officer)
Ajim Tamboli	Chief Financial Officer (Principal Financial Officer and Principal Accounting Officer)
	Director
Alexander Mayweg	
	Director
Bradley J. Bolzon	
	Director
Ali Behbahani	

Signature	Title
Kimberly L. Blackwell	Director
	Director
Andrew Schiff	Director
Chandra P. Leo	Director
Christine Siu	

Certain identified information has been excluded from this exhibit because it is both not material and is the type that the registrant treats as private or confidential. Information that was omitted has been noted in this document with a placeholder identified by the mark "[***]".

SERVICES AGREEMENT

THIS SERVICES AGREEMENT (this "Agreement"), effective as of 10 April, 2018 (the "Effective Date"), is by and between RIDGELINE THERAPEUTICS GMBH, a Basel Switzerland corporation ("Ridgeline"), and MONTE ROSA THERAPEUTICS AG (in formation), a corporation to be registered in Switzerland (the "Company").

WHEREAS, Ridgeline is engaged in the business of facilitating the start-up, funding and ongoing operation of new biotechnology companies and provides management, scientific, business development, clinical development and other operational services to startup companies on a contract basis; and

WHEREAS, the Company is a drug discovery and development company and desires to engage Ridgeline to provide the Company various services and make available to the Company certain resources of Ridgeline on the terms set forth herein.

NOW, THEREFORE, in consideration of the above premises and for other good and valid consideration, the receipt and adequacy of which are hereby acknowledged, the parties, intending to be legally bound, agree as follows:

ARTICLE 1

DEFINED TERMS

- **1.1 "Company Confidential Information"** shall mean (a) all Company Work Product and (b) any and all other data, information, technology, samples and specimens relating to the Company or any other person or entity with which Company has a commercial relationship (other than Ridgeline) or their respective products, product concepts, technologies, businesses, financial, marketing, clinical or regulatory affairs, manufacturing processes and procedures, or those of any other third party, whether written, graphic or oral, and whether or not furnished to or obtained by Ridgeline, either directly or indirectly, during the course of performing Services hereunder; but excluding, in any event, Ridgeline Work Product.
- 1.2 "Company Work Product" shall mean any and all results (including data) and products (interim and/or final) of the Services performed by Ridgeline, whether tangible or intangible, including, without limitation, each and every business, financial or other plan, computation, compilation of information, invention (whether or not patentable), discovery, design, drawing, protocol, process, technique, formula, trade secret, device, compound, substance, material, pharmaceutical, method, software program (including without limitation, object code, source code, flow charts, algorithms and related documentation), listing, routine, manual and specification, whether or not patentable or copyrightable, that are made, developed, perfected, designed, conceived or first reduced to practice by Ridgeline, either solely or jointly with others, in the course of the Services.

- 1.3 "Ridgeline Confidential Information" shall mean all data, information, technology, samples and specimens relating to Ridgeline or any other person or entity with which Ridgeline has a commercial relationship (other than the Company) or their respective products, technologies, businesses, financial, marketing, clinical or regulatory affairs, manufacturing processes and procedures, or those of any other third party, from whom Ridgeline receives information on a confidential basis, whether written, graphic or oral, furnished to or obtained by the Company, either directly or indirectly, other than during the course of providing Services hereunder, including, without limitation, Ridgeline Work Product, but excluding Company Work Product.
 - **1.4 "Confidential Information"** shall mean the Ridgeline Confidential Information or the Company's Confidential Information, as applicable.
- **1.5 "Ridgeline Key Team"** shall mean those individuals set forth on **Exhibit A** and such other individuals as may be agreed to between Ridgeline and the Company from time to time.
- 1.6 "Ridgeline Work Product" shall mean any and all results (including data) and products (interim and/or final) of any activities or services performed by Ridgeline on behalf of itself or any third party, other than in the course of performing the Services, whether tangible or intangible, including, without limitation, each and every invention (whether or not patentable), discovery, design, drawing, protocol, process, technique, formula, trade secret, device, compound, substance, material, pharmaceutical, method, software program (including without limitation, object code, source code, flow charts, algorithms and related documentation), listing, routine, manual and specification, whether or not patentable or copyrightable, that are made, developed, perfected, designed, conceived or first reduced to practice by Ridgeline, either solely or jointly with others, whether before, during or after the Term.
 - **1.7 "Term"** shall have the meaning provided in Section 2.6.

ARTICLE 2

SERVICES

- **2.1 Services.** Subject to the terms of this Agreement, for the Term determined pursuant to Section 2.6(a) hereof, Ridgeline shall provide or cause to be provided to the Company such services, in the nature of those described on Exhibit A, as may reasonably be requested by the Company and reasonably approved by Ridgeline from time to time following the date hereof (the *"Services"*).
- **2.2 Charges and Payment.** As compensation for its Services hereunder, the Company shall pay Ridgeline in accordance with the provisions of **Exhibit A** attached hereto. In addition, the Company shall reimburse Ridgeline for its actual expenses (including travel expenses) as reasonably incurred by Ridgeline or its employees and/or consultants in the course of performing Services. Ridgeline shall invoice the Company on a quarterly basis for all charges pursuant to this Agreement in accordance with the provisions of **Exhibit A** attached hereto.

2.3 General Obligations; Standard of Care.

- (a) **Performance Requirements**. Ridgeline shall use commercially reasonable efforts to provide Services subject to the terms of this Agreement and in accordance with its policies, procedures and practices then in effect, and shall exercise substantially the same care and skill as it exercises in performing similar services for itself.
- **(b) Changes.** The parties acknowledge that Ridgeline may make changes from time to time in the manner of performing the Services (e.g., if Ridgeline is making substantially similar changes in performing similar services for itself or its affiliates). To the extent they materially affect the Services, such changes shall be made in consultation with the Company.
- **(c) Compliance**. Ridgeline agrees to perform the Services in accordance with the terms and conditions contained in this Agreement and in compliance with all applicable federal, state and local laws and regulations.
- **(d) Communication**. On a regular basis during the Term, the parties shall conduct meetings, either in person or by telephone or video conference, to discuss the progress and results of the Services.

2.4 Confidentiality.

(a) Confidentiality. Except to the extent expressly authorized by this Agreement or otherwise agreed in writing by the parties, the parties agree that, during the Term of this Agreement and for [***] thereafter, the receiving party shall keep confidential and shall not publish or otherwise disclose and shall not use for any purpose other than as expressly provided for in this Agreement any Confidential Information of the other party. Each party may use the other party's Confidential Information only to the extent required to accomplish the purposes of this Agreement, consistent with any restrictions on the use of Confidential Information received by a third party and communicated by the party disclosing such Confidential Information. To the extent that any such restrictions on the use of Confidential Information set forth in this' Agreement, the parties each hereby agree to be bound by such restrictions. The parties agree and acknowledge that certain Confidential Information may be required for submission to the U.S. Food and Drug Administration and/or federal or state regulatory bodies. The parties acknowledge and agree that such submissions, to the extent required by applicable law, shall not constitute a violation of the terms of this Agreement if permitted under any applicable agreement with a third party for whom the disclosing party obtained the Confidential Information. Each party will use at least the same standard of care as it uses to protect proprietary or confidential information of its own (but in no event less than reasonable care) to ensure that its employees, agents, consultants and other representatives do not disclose or make any unauthorized use of such Confidential Information. Each party will promptly notify the other upon discovery of any unauthorized use or disclosure of such Confidential Information.

- **(b) Limitations**. Confidential Information shall not include any information that the receiving party can prove by competent evidence: (i) was already known to the receiving party without any obligations of confidentiality prior to receipt from the other party; (ii) was generally available to the public or otherwise part of the public domain at the time of its disclosure to the receiving party; (iii) became generally available to the public or otherwise part of the public domain after its disclosure, other than through any act or omission of the receiving party in breach of any obligation of confidentiality; (iv) was disclosed to the receiving party, other than under an obligation of confidentiality, by a third party who had no obligation not to disclose such information to others; or (v) was independently discovered or developed by the receiving party without the use of Confidential Information; *provided*, *however*, that any combination of features or disclosures shall not be deemed to fall within the foregoing exclusions solely because certain individual features are
- **(c) Authorized Disclosure**. Notwithstanding Section 2.4(a), a party may disclose Confidential Information of the other party, without violating the obligations of this Agreement, to the extent the disclosure is required by a valid order of a court or other governmental body having jurisdiction, provided that such party gives reasonable prior written notice to the other party of such required disclosure and makes a reasonable effort to obtain, or to assist the other party in obtaining, a protective order preventing or limiting the disclosure and/or requiring that the Confidential Information so disclosed be used only for the purposes for which the law or regulation requires, or for which the order was issued.
- **(d)** Use of Name/Publicity. Neither party shall use the other party's name in connection with any publication or promotion without the other party's consent, except as required by federal, state or local laws, rules and regulations. Neither party shall disclose the specific content or terms of this Agreement without the prior written consent of the other party.

2.5 Intellectual Property Rights

(a) Ownership. The Company shall own all right, title and interest in and to all Company Work Product, including, without limitation, all patent, copyright or other intellectual property rights therein, that is conceived or first reduced to practice by Ridgeline, either solely or jointly with others, in the course of performing the Services (collectively, the "Company Intellectual Property"), and neither this Agreement, nor the provision of the Services hereunder, shall give Ridgeline any right, title or interest in or to any Company Intellectual Property. The Company shall be responsible for all costs and expenses associated with such Company Work Product. The Company hereby grants to Ridgeline a non-exclusive, worldwide, fully-paid, royalty-free license, without the right to sublicense, to use the Company's technology solely as necessary or appropriate to perform Services under this Agreement during the Term. Ridgeline shall retain all right, title and interest in and to any and all Ridgeline Work Product, including, without limitation, all patent, copyright or other intellectual property rights therein (collectively, the "Ridgeline Intellectual Property"), and neither this Agreement, nor the provision of the Services hereunder, shall give the Company any right, title or interest in or to any Ridgeline Intellectual Property.

(b) Assignment; Assistance. Ridgeline hereby assigns all of Ridgeline's right, title and interest in and to any Company Intellectual Property to the Company without royalty or any other consideration and agrees to execute all applications, assignments or other instruments reasonably requested by the Company in order for the Company to establish its ownership of such Company Intellectual Property and to obtain whatever protection for such Company Intellectual Property, including copyright and patent rights in any and all countries designated by the Company on such Company Intellectual Property as the Company shall determine. Ridgeline agrees to assist the Company, or its designee, in every reasonable way (but at the Company's expense) to secure the Company's rights in Company Intellectual Property and any copyrights, patents or other intellectual property rights relating to all Company Intellectual Property in any and all countries designated by the Company, including the disclosure to the Company of all pertinent information and data with respect to all Company Intellectual Property, the execution of all applications, specifications, oaths, assignments and all other instruments that the Company may deem necessary in order to apply for and obtain such rights and in order to assign and convey to the Company, its successors, assigns and nominees the sole and exclusive right, title and interest in and to all Company Intellectual Property. Ridgeline also agrees that its obligation to execute or cause to be executed any such instrument or papers shall continue after the expiration or termination of this Agreement. Ridgeline agrees that, if the Company is unable because of Ridgeline's unavailability, dissolution, or otherwise, to secure Ridgeline's signature for the purpose of applying for or pursuing any application for any United States or foreign patents or copyright registrations covering the Company Intellectual Property assigned to the Company herein, then, until such time Ridgeline becomes available it hereby designates and appoints the Company and its duly authorized officers and agents as Ridgeline's agent and attorney-in-fact, to act for and on Ridgeline's behalf to execute and file any such applications and to do all other lawfully permitted acts only to further the prosecution and issuance of patents and copyright registrations with the same legal force and effect as if executed by Ridgeline.

2.6 Term; Termination.

- **(a) Term**. The term of this Agreement (the "**Term**") shall commence on the Effective Date and shall remain in effect until terminated in accordance with this Section 2.6.
- **(b) Election to Terminate.** The Company may terminate this Agreement either with respect to all, or with respect to any one or more, of the Services provided hereunder (including, without limitation, terminating the provision of Services by any member or members of the Ridgeline Key Team) at any time and from time to time, for any reason or no reason, by giving written notice to Ridgeline at least [***] prior to the date of such termination. Ridgeline may terminate this Agreement either with respect to all, or with respect to any one or more, of the Services provided hereunder at any time and from time to time, for any reason or no reason, by giving written notice to the Company at least [***] prior to the date of such termination. In addition, the parties may at any time agree in writing to terminate this Agreement with respect to some or all of the Services, effective immediately or as indicated in such writing. In the event of any termination with respect to one or more, but less than all, Services, this Agreement shall continue in full force and effect with respect to any Services not terminated hereby.
- **(c) Payment Upon Early Termination**. In the event of termination of this Agreement or any Services hereunder, Ridgeline shall be paid for all work completed through the date of termination in accordance with this Agreement, including reasonable and documented out-of-pocket expenses and any non-cancelable commitments reasonably incurred by Ridgeline in accordance with this Agreement. Ridgeline shall refund to the Company any prepaid amounts not earned by Ridgeline prior to the date of such termination, including as set forth in Section 2.2 hereof.

(d) Survival Upon Termination. Expiration or termination of this Agreement will not relieve either party of any obligation accruing prior to such expiration or termination. Article 1, Sections 2.2, 2.4, 2.5, 2.6(c), 2.6(d), 3.3, 3.4, and Articles 4 and 5 will survive expiration or termination of this Agreement.

ARTICLE 3

REPRESENTATIONS AND WARRANTIES; DISCLAIMER; LIMITATION OF LIABILITY

- **3.1 Mutual Representations and Warranties**. Each party represents and warrants to the other that: (a) it has full power and authority to enter into this Agreement and to perform its obligations hereunder; (b) this Agreement is legally binding upon it, enforceable against it in accordance with its terms, and does not conflict with any charter or constituting document, or any agreement, instrument or understanding, oral or written, to which it is a party or by which it may be bound, nor violate any material law or regulation of any court, governmental body or administrative or other agency having jurisdiction over it; and (c) such party is not under any pre-existing obligation inconsistent with the provisions of this Agreement.
- **3.2 Ridgeline Representations and Warranties**. Ridgeline hereby represents and warrants to the Company that the Services shall be performed by qualified personnel in a good, timely, efficient and professional manner.
- 3.3 Disclaimer of Warranties. EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, EACH PARTY DISCLAIMS ALL WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, AND EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT, RIDGELINE MAKES NO REPRESENTATIONS OR WARRANTIES AS TO THE QUALITY, SUITABILITY OR ADEQUACY OF THE SERVICES FOR ANY PURPOSE OR USE.
- **3.4 Limitation of Liability**. NEITHER PARTY SHALL BE ENTITLED TO RECOVER FROM THE OTHER PARTY ANY SPECIAL, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES IN CONNECTION WITH THIS AGREEMENT; *provided*, *however*, that this Section 3.4 shall not be construed to limit either party's indemnification obligations under Article 4.

ARTICLE 4

INDEMNIFICATION

4.1 By the Company. The Company hereby agrees to save, defend, indemnify and hold harmless Ridgeline, its affiliates and their respective officers, directors, employees, consultants and agents (each, a "*Ridgeline Party*") from and against any and all losses, damages, liabilities, expenses and costs, including reasonable legal expense and attorneys' fees "*Losses*"), to which any Ridgeline Party may become subject as a result of any claim, demand, action or other proceeding by any third party to the extent such Losses arise directly or indirectly out of (a) the performance of the Services, (b) the development, manufacture, use, handling, storage, sale or other disposition of any product by the Company, or (c) the gross negligence or willful misconduct of any Company Party or the breach by the Company of any warranty, representation, covenant or agreement made by the Company in this Agreement, except, in each case, to the extent such Losses result from the gross negligence or willful misconduct of any Ridgeline Party or the breach by Ridgeline of any warranty, representation, covenant or agreement made by Ridgeline in this Agreement.

- **4.2** By Ridgeline. Ridgeline hereby agrees to save, defend, indemnify and hold harmless the Company, its affiliates and their respective officers, directors, employees, consultants and agents (each, a "Company Party") from and against any and all Losses to which any Company Party may become subject as a result of any claim, demand, action or other proceeding by any third party to the extent such Losses arise directly or indirectly out of the gross negligence or willful misconduct of any Ridgeline Party or the breach by Ridgeline of any warranty, representation, covenant or agreement made by Ridgeline in this Agreement, except, in each case, to the extent such Losses result from the gross negligence or willful misconduct of any Company Party or the breach by the Company of any warranty, representation, covenant or agreement made by the Company in this Agreement.
- **4.3 Control of Defense.** In the event a party seeks indemnification under Section 4.1 or Section 4.2, it shall inform the other party (the "*Indemnifying Party*") of a claim as soon as reasonably practicable after it receives notice of the claim, shall permit the Indemnifying Party to assume direction and control of the defense of the claim (including the right to settle the claim solely for monetary consideration with no admission of fault), and shall cooperate as requested (at the expense of the Indemnifying Party) in the defense of the claim.
- **4.4 Liability Insurance**. Each party agrees to maintain during the Term usual and customary liability and workers compensation insurance in amounts consistent with industry standards and to provide a certificate of insurance evidencing such coverage to the other party upon request.

ARTICLE 5

MISCELLANEOUS

- **5.1 Taxes**. Ridgeline will pay any and all taxes levied on account of any payments made to it under this Agreement. If any taxes are required to be withheld by the Company from any payment to Ridgeline, the Company shall (a) deduct such taxes from the payment, (b) timely pay the taxes to the proper taxing authority, and (c) send proof of payment to Ridgeline and certify its receipt by the taxing authority within [***] following such payment.
- **5.2 Relationship of Parties.** Nothing in this Agreement shall be deemed or construed by the parties or any third party as creating the relationship of principal and agent, partnership or joint venture between the parties, it being understood and agreed that no provision contained herein, and no act of the parties, shall be deemed to create any relationship between the parties other than the relationship of independent contractor nor be deemed to vest any rights, interest or claims in any third parties.

- **5.3 Integration**. This Agreement (including the Exhibits hereto) contains the complete, final and exclusive agreement of the parties relating to the subject matter hereof, and supersedes all prior and contemporaneous oral and written agreements or arrangements between the parties. To the extent this Agreement conflicts with any other agreements, written or oral, between the parties, this Agreement controls.
 - **5.4 Modification and Amendment.** This Agreement may not be modified or amended except in a writing signed by the parties.
 - **5.5 Governing Law**. This Agreement shall be governed by and construed in accordance with the laws of Switzerland
- **5.6 No Implied Licenses.** No right or license is granted under this Agreement by either party to the other, either expressly or by implication, except those specifically set forth herein.
- **5.7 Severability**. If any provision of this Agreement should be held invalid or unenforceable, the remaining provisions shall be unaffected and shall remain in full force and effect, to the extent consistent with the intent of the parties as evidenced by this Agreement as a whole.
- **5.8 Assignment**. Except as expressly provided hereunder, neither this Agreement nor any rights or obligations hereunder may be assigned or otherwise transferred by either party without the prior written consent of the other party (which consent shall not be unreasonably withheld); *provided*, *however*, that the Company may assign this Agreement and its rights and obligations hereunder without Ridgeline's consent in connection with the transfer or sale of all or substantially all of the Company's business to which this Agreement relates to a third party, whether by merger, sale of stock, sale of assets or otherwise. The rights and obligations of the parties under this Agreement shall be binding upon and inure to the benefit of the successors and permitted assigns of the parties. Any assignment not in accordance with this Agreement shall be void.
 - 5.9 Headings. Section headings are for convenience of reference only and shall not be considered in the interpretation of this Agreement.
- **5.10 Force Majeure**. In the event of a delay caused by inclement weather, fire, flood, strike or other labor dispute, act of God, act of governmental officials or agencies, or any other cause beyond the control of the parties, the party or parties so affected shall be excused from performance hereunder for the period of time attributable to such delay, which may extend beyond the time lost due to one or more of the causes mentioned above. In the event of any such delay, the parties may, in their sole discretion, amend this Agreement, as appropriate, by mutual written agreement.
- **5.11 Notices**. Any notices required or permitted hereunder shall be given by overnight courier to the appropriate party at the address specified below or at such other address as the party shall specify in writing.

If to Ridgeline: Ridgeline Therapeutics GmbH

Aeschenvorstadt 36, 4051 Basel, Switzerland

Attn: Dr. Alexander Mayweg

If to the Company: Monte Rosa Therapeutics AG

Aeschenvorstadt 36, 4051 Basel, Switzerland

Attn: Chief Executive Officer

All notices shall be deemed made upon receipt by the addressee as evidenced by the applicable written receipt.

5.12 Counterparts. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.

5.13 Non-Waiver. No failure or delay of one of the parties to insist upon strict performance of any of its rights or powers under this Agreement shall operate as a waiver thereof, nor shall any other single or partial exercise of such right or power preclude any other further exercise of any rights or remedies provided by law.

5.14 Waiver of Corporate Opportunity. In the event that either party to this Agreement or any director, officer, employee or representative of such party (the "*Primary Party*") acquires knowledge of a potential transaction or other matter (including, but not limited to, any compounds or other assets or the opportunity to acquire interests thereof) and that may be an opportunity of interest (a "*Corporate Opportunity*") for the other party to this Agreement (the "*Other Party*"), then the Other Party (i) renounces any expectancy that the Primary Party offer an opportunity to participate in such Corporate Opportunity to the Other Party and (ii) to the fullest extent permitted by law, waives any claim that such opportunity constituted a Corporate Opportunity that should have been presented by the Primary Party to the Other Party or any of its affiliates.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties have executed this Services Agreement as of the date first above written.

RIDGELINE THERAPEUTICS GMBH

By: /s/ Alexander Mayweg

Name: Alexander Mayweg

Title: CSO

MONTE ROSA THERAPEUTICS AG

(in formation)

By: <u>/s/ Illegible</u>

Name: Illegible

By: /s/ Illegible
Name: Illegible

By: /s/ Illegible

Name: Illegible

Title: Company Secretary, Cancer Research Technology

Limited

By: /s/ Illegible
Name: Illegible
Title: Director, ICR

Versant Venture Capital VI, L.P. By: Versant Ventures VI GP, L.P. By: Versant Ventures VI GP-GP, LLC

By: /s/ Bradley BolzonName: Bradley J. BolzonTitle: Managing Director

EXHIBIT A

SERVICES

- **1.** For purposes of this Agreement, the "*Ridgeline Key Team*" shall initially mean Alexander Mayweg, CSO of Ridgeline, [***], and additional team members for scientific or laboratory expertise and work as required or required from time to time in consultation with the Company. Subject to the provisions of Section 2.1 hereof, Ridgeline shall use commercially reasonable efforts to provide, among others, the following Services as may be requested from time to time by the Company:
 - **A. Set up services**. Ridgeline shall or may procure any necessary set up services such as company foundation, drafting or creation or refinement of business plan, capital equipment and infrastructure to perform Services for Company, including office equipment, IT systems and furniture.
 - **B. Research and Development Services.** Ridgeline shall or may provide to the Company general research and development services pursuant to a research and development plan to be mutually agreed by the parties, including [***]
 - **C. Management and Administrative Services**. Ridgeline shall or may provide management, strategic and administrative services, as mutually agreed by the parties, including:
 - i. [***];ii. [***];iii. [***].
 - D. Other. Ridgeline shall provide such other Services as mutually agreed between Ridgeline and the Company.
- 2. The Company shall pay Ridgeline per calendar quarter (or any partial quarter on a pro rata basis) for Services to be performed pursuant to this Agreement. Unless otherwise mutually agreed to by Ridgeline and the Company in writing prior to the commencement of Services in any calendar quarter, Ridgeline shall invoice the Company for the actual amounts incurred set forth in the table below (current estimates), on or about the [***] following each calendar quarter for the Services performed during the prior quarter, together with the amount of reimbursable costs and expenses incurred by Ridgeline on behalf of the Company pursuant to Section 2.2 hereof prior to such invoice. The Company agrees to pay all amounts due to Ridgeline arising under this Agreement within [***] of receipt of any such invoice.
- **3.** In connection with the Services to be provided hereunder, the Company will issue to Ridgeline an allocation (to be agreed by the Company board) of the Company's Common Stock/Options Pool. Ridgeline may allocate and transfer such shares of the Company's Common Stock to members of the Ridgeline Key Team from time to time according to an agreed vesting schedule, and the Company agrees to use commercially reasonable efforts to provide all necessary consents and to facilitate any such transfer of such shares.

4. The Company and Ridgeline acknowledge that the fees payable for Services have been set by reference to the costs expected to be incurred by Ridgeline in the provision of the Services to the Company, plus [***]. The currently projected cost estimates for Ridgeline set up, R&D (including drug discovery and design), management and administrative services (excluding the [***]) are as follows, shown in millions CHF. These costs and timelines are subject to approval by the Company's BOD and may be altered at the BOD level.

[***]
[***]
[***]
[***]
[***]
[***]

- **5.** The Services fees and expense reimbursements shall be payable in CHF (Swiss Francs) (unless mutually agreed by the parties) and shall be subject to all applicable government regulations and rulings.
- **6.** The Company expressly acknowledges that Ridgeline is engaged in the business of facilitating the start-up, funding and ongoing operation of multiple biotechnology companies and providing management, scientific, business development, financial and other operational services to those companies and that neither Ridgeline nor any other company to which Ridgeline provides services shall have any exclusivity or similar obligation to the Company, including without limitation any corporate opportunity obligation or any obligation to disclose or make available to the Company any information, potential transaction or other matter of which any such Ridgeline Party becomes aware otherwise than solely in the course of performing Services under this Agreement on behalf of the Company.

Certain identified information has been excluded from this exhibit because it is both not material and is the type that the registrant treats as private or confidential. Information that was omitted has been noted in this document with a placeholder identified by the mark "[***]".

SERVICES AGREEMENT

THIS SERVICES AGREEMENT (the "<u>Agreement</u>") is made as of December 29, 2020, by and between Monte Rosa Therapeutics Ag ("<u>Recipient</u>"), and Monte Rosa Therapeutics, Inc. ("<u>Service Company</u>") and shall be effective as of January 1, 2020.

WITNESSETH:

WHEREAS, Service Company is the parent company of the Recipient (together with the Service Company, the "Group");

WHEREAS, Recipient is, for tax purposes, the owner of the intellectual property for the Group;

WHEREAS, Service Company agrees to provide or cause to be provided to Recipient certain research and development services and management, administrative and support services for Recipient's business operations (the "Business") on the terms set forth in this Agreement, including Appendix A attached hereto.

NOW, THEREFORE, subject to the terms, conditions, covenants and provisions of this Agreement, Recipient and Service Company each mutually covenant and agree as follows:

ARTICLE I SERVICES PROVIDED

- 1.1 <u>Services</u>. Upon the terms and subject to the conditions set forth in this Agreement, Service Company will provide each of those services (hereinafter referred to individually as a "<u>Service</u>", and collectively as the "<u>Services</u>") set forth in <u>Appendix A</u> attached hereto (which is incorporated herein and made a part of this Agreement) to Recipient, as such Services are needed during the term of this Agreement.
- 1.2 <u>Personnel</u>. In providing the Services, Service Company may, as it deems necessary or appropriate, (i) use the personnel of Service Company or any affiliate thereof, and (ii) employ the services of reputable and qualified third parties.
 - 1.3 Level of Services. The Services will be provided and utilized in good faith and in a reasonable manner by the parties hereto.
- 1.4 <u>Service Company Access</u>. To the extent reasonably required for personnel of Service Company to perform the Services, Recipient shall provide personnel of Service Company or its affiliates with any reasonably necessary access during normal business hours (to the extent practicable) to its equipment, office space, plants, telecommunications and computer equipment and systems, and any other areas and equipment.

ARTICLE II COMPENSATION

2.1 <u>Invoices/Payment</u>. For the initial period from January 1, 2020 through December 31, 2020 and thereafter at the end of each calendar month during the term hereof, Service Company, and/or its affiliates, will submit a single itemized invoice to Recipient for all Services provided to such Recipient during the calendar month just ended in accordance with the pricing of such Services set forth on <u>Appendix A</u>. Payment of all undisputed invoiced amounts shall be made by check or electronic funds transmission in U.S. Dollars within [***] of the invoice date unless otherwise agreed to by the parties. Invoices not paid by the due date shall bear interest at the Applicable Federal Rate from the due date until payment is received. All payments shall be made to the account designated by Service Company.

ARTICLE III CONFIDENTIALITY

- 3.1 <u>Confidential Information</u>. All information that is disclosed or provided by one party (the "<u>Disclosing Party</u>") to the other party (the "<u>Recipient</u>") pursuant to this Agreement, whether in oral, written, graphic, electronic, or any other form, shall be the "Confidential Information" of the disclosing party, except that all deliverables or results of Services, except as provided in <u>Section 4.3</u>, shall be deemed the Confidential Information of Recipient. Except to the extent expressly authorized by this Agreement or by the other party in writing, during the term of this Agreement and for five (5) years thereafter, each party shall maintain in strict trust and confidence and shall not disclose to any third party or use for any purpose other than as provided for in this Agreement any Confidential Information of the other party. Service Company may use Recipient's Confidential Information only to the extent required to perform the Services, and for no other purpose. Each party agrees that it shall not use the other party's Confidential Information for any purpose or in any manner that would constitute a violation of applicable laws.
- 3.2 Exceptions. The obligations of confidentiality and nonuse set forth in Section 3.1 shall not apply to any specific portion of information that the Recipient can demonstrate by competent written proof: (a) is in the public domain or comes into the public domain through no fault of the Recipient; (b) is furnished to the Recipient by a third party rightfully in possession of such information and not subject to a duty of confidentiality with respect thereto; (c) is already known by the Recipient at the time of receiving such Confidential Information from the Disclosing Party as evidenced by the Recipient's prior written records; or (d) is independently developed by the Recipient without access to the Disclosing Party's Confidential Information.
- 3.3 <u>Authorized Disclosure</u>. Notwithstanding the foregoing in this ARTICLE III, the Recipient may disclose certain Confidential Information of the Disclosing Party to the extent such disclosure is required by law or regulation, or pursuant to a valid order of a court or other governmental body having jurisdiction, provided that the Recipient provides the Disclosing Party with reasonable prior written notice of such disclosure and reasonable assistance in obtaining a protective order or confidential treatment preventing or limiting the disclosure and/or requiring that the Confidential Information so disclosed be used only for the purposes for which the law or regulation required, or for which the order was issued.
- 3.4 <u>Return of Confidential Information</u>. Upon termination or expiration of the Agreement, or upon written request of the Disclosing Party, the Recipient shall promptly return to the Disclosing Party or destroy all documents, notes and other tangible materials representing the Disclosing Party's Confidential Information and all copies thereof; provided, however, that each party may retain a single archival copy of the other party's Confidential Information for the sole purpose of facilitating compliance with the surviving provisions of this Agreement.

ARTICLE IV INTELLECTUAL PROPERTY

- 4.1 <u>Work for Hire</u>. Except as provided in <u>Section 4.3</u>, all deliverables resulting from the Services provided under this Agreement are "Works Made for Hire" as defined in the U.S. Copyright Act and other copyrightable works will be deemed, upon creation, to be assigned to Recipient.
- 4.2 <u>Inventions and Assignment</u>. Except as provided in <u>Section 4.3</u>, any materials, data, processes, documents, deliverables, information (including Confidential Information), discoveries, inventions, know-how and the like developed or generated by or on behalf of Service Company during the course of performing Services, whether or not patentable, and all related patent, copyright and other intellectual property rights in any of the foregoing (collectively the "<u>Inventions</u>") shall be the sole and exclusive property of Recipient. Service Company hereby assigns, and to the extent it cannot presently assign, agrees to assign, to Recipient all of Service Company's worldwide right, title and interest in and to such Inventions. Service Company shall assist Recipient in securing for Recipient any patents, copyrights or other proprietary rights in such Inventions, and shall take such actions and execute such documents as Recipient may reasonably request in connection with providing such assistance or otherwise to vest in Recipient all right, title and interest in and to such Inventions, including without limitation any and all applications, assignments or other instruments. Service Company shall be compensated for all of its reasonable out-of-pocket costs and expenses associated with such requested assistance. To the extent Inventions cannot be assigned to Recipient under this ARTICLE IV, Service Company grants to Recipient an exclusive perpetual, irrevocable, transferable, fully paid-up, worldwide license, with the right to grant sublicenses, under such Inventions for any and all purposes.
- 4.3 Service Company Property. Any (i) processes or process improvements developed by Service Company related to Service Company's pre-existing technology that are general in nature and are not unique or specific to the targets or work performed for Recipient and as to which Recipient has not specifically funded development under this Agreement, and (ii) pre-existing patents, know-how or other technology or information owned or controlled by Service Company prior to the effective date of this Agreement and that are incorporated into or embodied in any Inventions or deliverables provided by Service Company under this Agreement will be owned by Service Company. For clarity, such pre-existing technology or pre-existing patents, know-how or other technology or information owned or controlled by Service Company shall not include any such technology, patents, know-how or information which has been assigned or exclusively licensed to Recipient or any third party. Service Company hereby grants to Recipient a perpetual, irrevocable, non-exclusive, worldwide, royalty-free, fully paid-up license (with a right to grant sublicenses) under Service Company's intellectual property rights solely to the extent necessary for Recipient to utilize the Inventions and the other deliverables of Services for any purpose. The foregoing license may be sublicensed by Recipient in connection with the transfer by Recipient of the deliverables or Inventions to which the license relates.
- 4.4 No Other License Grant. Except as expressly set forth in this Agreement, nothing in this Agreement, nor the delivery of any information or materials to Service Company by Recipient (or any third party acting on its behalf) in connection with Service Company's performance of Services under this Agreement shall be deemed to grant to either party any right or license under any patents, patent applications, know-how, technology, inventions or other intellectual property of the other party. Notwithstanding anything in this Agreement to the contrary, Recipient shall own all right, title and interest in and to all inventions, know-how, information and materials, and all related intellectual property rights, that arise from Recipient's use of Inventions and the other deliverables and results of Services arising from this Agreement.
- 4.5 <u>Subcontractors</u>. Service Company will ensure that its agreement with any permitted subcontractor includes the assignment of any Inventions to Recipient or to the Service Company, with the right of further transfer to Recipient.

ARTICLE V TERM

- 5.1 <u>Term</u>. This Agreement shall become effective on the date hereof and shall remain in force until the parties hereto mutually agree to terminate. The date of mutually agreed upon termination, is referred to as the "<u>Expiration Date</u>."
- 5.2 <u>Effects; Survival of Certain Obligations</u>. In the event of a termination or expiration of this Agreement in accordance with the terms hereof, this Agreement shall immediately become null and void and have no effect, and none of the parties shall have any liability of any nature whatsoever hereunder, or in connection with the transactions contemplated hereby, except that this <u>Section 5.2</u> and all other obligations of the parties specifically intended to be performed after the termination of this Agreement shall survive any termination of this Agreement. All obligations of Recipient and Service Company under this Agreement that arose prior to its termination or expiration and that have not been fully performed in accordance with the terms of this Agreement prior to such termination or expiration shall survive any such termination or expiration of this Agreement.

ARTICLE VI MISCELLANEOUS

- 6.1 <u>Complete Agreement</u>. This Agreement, together with <u>Appendix A</u>, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties hereto, or any of them, with respect to the subject matter hereof.
- 6.2 <u>Counterparts</u>. This Agreement may be executed in counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each of the parties, it being understood that all parties need not sign the same counterpart.
- 6.3 <u>Amendments; Waivers</u>. The parties may mutually amend or waive any provision of this Agreement at any time. No amendment or waiver of any provision of this Agreement shall be valid unless the same shall be in writing and signed by each of the parties. No waiver by any party of any default, misrepresentation or breach of warranty or covenant hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.
- 6.4 <u>Assignment; Reliance of Other Parties</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto in whole or in part (whether by operation of law or otherwise) without the prior written consent of the other parties and any attempt to make any such assignment without such consent shall be null and void; <u>provided</u>, that Recipient may, without the consent of Service Company, assign this Agreement to any acquirer of all or substantially all of the business and assets of Recipient, whether by acquisition of assets, merger or otherwise. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns. This Agreement (including the documents and instruments referred to herein) is not intended to confer upon any person or entity other than the parties hereto any rights or remedies under or by reason of this Agreement.
- 6.5 <u>Governing law; Jurisdiction and Venue</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Massachusetts without regard to its rules of conflict of laws.
- 6.6 <u>Severability</u>. In the event that any one or more provisions of this Agreement shall for any reason be held invalid, illegal or unenforceable in any respect, by any court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provisions of this Agreement and the parties shall use their reasonable best efforts to substitute a valid, legal and enforceable provision which, insofar as practicable, implements the original purposes and intents of this Agreement.

6.7 <u>Relationship of Parties</u>. Nothing in this Agreement is intended or should be construed to create a partnership, joint venture, or employer-employee relationship between Recipient and any of Service Company's employees or agents. Service Company is not the agent of Recipient and is not authorized, and, except as otherwise provided herein with respect to performance of the Services, must not represent to any third party that it is authorized, to make any commitment or otherwise act on behalf of Recipient.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed the day and year first above written.

MONTE ROSA THERAPEUTICS AG

By: /s/ Markus Warmuth

Name: Markus Warmuth, M.D. Title: Director, Authorized Signatory

MONTE ROSA THERAPEUTICS, INC.

By: /s/ Ajim Tamboli

Name: Ajim Tamboli

Title: Chief Financial Officer

Services Agreement Signature Page

APPENDIX A: SERVICES; PRICING

I. Research and Development Services

Service Company shall perform such research and development activities, as mutually agreed to with Recipient, during the term of this agreement, to support the development, maintenance and support of Recipient's intellectual property portfolio.

II. Management, Administrative and Support Services

Service Company shall make available to Recipient, during the term of this agreement, certain management, administrative and support services, as mutually agreed to with Recipient, to assist Recipient in conducting its business operations. The Services may include, but are not limited to, management, financial, investment, human resource services, information systems, and other services as the parties may agree to from time to time.

III. PRICING

A. Services

[***]

B. Out-of-Pocket Expenses

Service Company shall invoice Recipient for all direct expenses incurred or paid by Service Company in the performance of the Services under this Agreement.

Certain identified information has been excluded from this exhibit because it is both not material and is the type that the registrant treats as private or confidential. Information that was omitted has been noted in this document with a placeholder identified by the mark "[***]".
Annex 2.7: Background IP License Agreement
[separate Document]

LICENSE AGREEMENT

(1) CANCER RESEARCH TECHNOLOGY LIMITED

AND

(2) THE INSTITUTE OF CANCER RESEARCH

AND

(3) MONTE ROSA THERAPEUTICS AG

April 10, 2018

Table of Contents

		rage
1.	INTERPRETATION	2
2.	LICENCES	ϵ
3.	PERFORMANCE	8
4.	CONSIDERATION	8
5.	INTELLECTUAL PROPERTY PROTECTION, PROCEEDINGS AND COSTS	8
6.	REPRESENTATIONS AND WARRANTIES	10
7.	INDEMNITY	12
8.	INSURANCE	13
9.	LIMITATION OF LIABILITY	13
10.	CONFIDENTIALITY	14
11.	PUBLICATIONS	14
12.	TERM AND TERMINATION	14
13.	EFFECTS OF TERMINATION	15
14.	FORCE MAJEURE	15
15.	ASSIGNMENT	15
16.	NOTICES	16
17.	VARIATION	16
18.	ENTIRE AGREEMENT	16
19.	FURTHER ASSURANCE	17
20.	WAIVER	17
21.	SEVERABILITY	17
22.	LAW AND JURISDICTION	18
23.	EXECUTION	18
	ANNOUNCEMENTS AND USE OF NAMES	18
	DISPUTE RESOLUTION	18
	NO THIRD PARTY BENEFICIARIES	19
27.	LANGUAGE: AUTHORSHIP	19

LICENSE AGREEMENT

This License Agreement, dated as of the 10th day of April, 2018 (the "Execution Date"), is made by and among:

- **(1) CANCER RESEARCH TECHNOLOGY LIMITED**, a company registered in England and Wales under number 1626049 with registered office at Angel Building, 407 St John Street, London. EC IV 4AD. England ("*CRT*");
- (2) THE INSTITUTE OF CANCER RESEARCH: ROYAL CANCER HOSPITAL, a company limited by guarantee and registered in England (registered number 00534147) and registered charity whose office is at 123 Old Brompton Road, London, SW17 9LN, England ("ICR"); and
- **(3) MONTE ROSA THERAPEUTICS AG,** a company organized under the laws of Switzerland under number Switzerland ("*Company*"). with registered office at Aeschenvorstadt 36, 4051 Basel,

Whereas:

- **(A)** CRT is an oncology focused technology transfer and development company, wholly owned by Cancer Research UK ("*CRUK*"), a company limited by guarantee (registered in England and Wales under number 4325234) and a charity (registered in England under number 1089464 and registered in Scotland under number SC041666) of Angel Building, 407 St John Street, London, EC IV 4AD, United Kingdom;
- **(B)** The ICR is a registered charity and a college of the University of London, and is dedicated to the pursuit of scientific and clinical research into the understanding, diagnosis and treatment of cancer;
- **(C)** Pursuant to an agreement dated [***] between CRT and the ICR. CRT and the ICR jointly owns certain intellectual property generated at the ICR using funding from CRUK related to the field of cereblon (CRBN)-mediated protein degradation;
 - (D) The Company is engaged in the research and development of protein degradation-based therapeutics;
- **(E)** As a condition to and concurrent with the execution and delivery' of this Agreement, the Parties are entering into that certain (a) Formation and Investment Agreement (as may be amended, the "Formation and Investment Agreement"), and the related Shareholders' Agreement (as may be amended, the "Shareholders' Agreement"), by and among the Company and its shareholders, including CRT and the ICR. pursuant to which the Company issues common shares of the Company as consideration for the rights granted to it under this Agreement, and (b) Collaboration and Option Agreement (as may be amended, the "Collaboration and Option Agreement"), pursuant to which the Company engages the ICR to conduct certain research and development services and grants CRT and the ICR certain rights as set forth therein; and
- **(F)** As a condition to the execution and delivery of the Formation and Investment Agreement and the Collaboration and Option Agreement, the Company desires to obtain, and CRT and the ICR are willing to grant, certain exclusive and non-exclusive licenses under CRT's and the ICR's intellectual property rights in the field of cereblon (CBRN) mediated protein degradation, on the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the foregoing, the covenants and premises contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

1. INTERPRETATION

- 1.1 In this Agreement the following words and expressions shall have the following meanings:
- "*Affiliate*" means, with respect to a Party, any person that, whether now or in the future, Controls, is Controlled by or is under common Control with a Party. For the avoidance of doubt, Versant Ventures shall not be deemed an Affiliate of Company.
- "Agreement" means this License Agreement and each of the Schedules attached hereto, as amended from time to time in accordance with the provisions of this License Agreement.
- "Business Day" means a day other than a Saturday, Sunday, or any public holiday in England or any date on which commercial banks located in Switzerland, and/or New York. New- York, U.S.A, are authorized or required by law to close.
 - "Claims" has the meaning given in Clause 7.1.
 - "Collaboration and Option Agreement" has the meaning given in the recitals.
- "Commencement" means, in relation to a clinical trial, the date upon which administration of a Licensed Product to the first human subject has occurred, whether such subject is a healthy volunteer or a patient.
 - "Commercially Reasonable Efforts" means [***].
- "Competent Authority" means any local or national agency, authority, department, inspectorate, minister, ministry' official or public or statutory person (whether autonomous or not) of or of any government of any country having jurisdiction over the Agreement or any of the Parties or over the development or marketing of medicinal products including the FDA, the European Medicines Agency, the European Commission and the European Court of Justice.
- "Compound Intellectual Property" means that part of the CRT/ICR Existing Intellectual Property that (i) consists of the CRT /ICR Existing Compound Library, (ii) consists of, contains or relates directly and solely to any CRT/ICR Existing Compound and/or (iii) is set out in Schedule 1A part (i), but for the avoidance of doubt excludes the CRT/ICR Existing Intellectual Property' referred to in paragraph (b) of the definition of Non-Compound Intellectual Property.
- "Control" means the possession (directly or indirectly) of fifty percent (50%) or more of the voting stock or other equity' interest of a subject entity with the power to vote, or the power in fact to control the management decisions of such entity through the ownership of securities or by contract or otherwise and "Controlling", "Controls", "Controlled by" and "under common Control with" as used with respect to any Party shall be construed accordingly.

"CRBN" means cereblon.

"CRT/ICR Background Hit Intellectual Property" means [***].

"CRT/ICR Existing Compound Library" means [***].

"CRT/ICR Existing Compounds" means any and all compounds listed in Schedule IC.

"CRT/ICR Existing Intellectual Property" means [***].

"CRT/ICR Existing Know How" means [***].

"CRT/ICR Existing Materials" means [***].

"CRT/ICR Existing Patents" means those Patents that claim inventions within the CRT/ICR Existing Compounds, CRT/ICR Existing Compound Library, ICR/CRT Existing Know How, and/or ICR/CRT Existing Materials.

"CRT/ICR Existing Screening Platform" means the [***].

"CTU" means the ICR's Cancer Research UK Cancer Therapeutics Unit, which is led as of the Effective Date by Professor Rajesh Chopra.

"Effective Date" has the meaning given in Clause 12.1.

"Execution Date" has the meaning given in the preamble.

"Executive Officers" means an authorised executed officer of the Company, the Chief Executive Officer of CRT and the Director of Enterprise of the ICR or such other authorised officer of a Party' as may be substituted from time to time upon the giving of written notice to the other Party.

"FDA" means the United States Food and Drug Administration or any successor to it.

"Field" means the treatment, prevention and/or diagnosis of any and all diseases, disorders or conditions.

"Force Majeure" means in relation to a Party any event or circumstance which is beyond the reasonable control of that Party, which event or circumstance that Party could not reasonably be expected to have taken into account at the Effective Date and which results in or causes the failure of that Party' to perform any or all of its obligations under this Agreement including act of God, lightning, fire, storm, flood, earthquake, strike, lockout or other industrial disturbance, war, terrorist act, blockade, revolution, riot, insurrection, civil commotion, public demonstration, sabotage, act of vandalism, explosion, provided that lack of funds shall not be interpreted as a cause beyond the reasonable control of that Party.

- "Formation and Investment Agreement" has the meaning given in the recitals.
- "Indemnified Parties" means CRT, CRUK, the ICR and their respective officers, employees and agents (each, an "Indemnified Party" or "Indemnitee").
 - "JSC" has the meaning set forth in the Collaboration and Option Agreement.
- "Know How" means any technical, business or financial information and other information, of any type whatsoever, in any tangible or intangible form, which is not in the public domain as of the Effective Date, including, know-how, trade secrets, ideas, concepts, inventions, discoveries, data, formulae, specifications, information relating to Materials (including biological and chemical structures and functions as well as methods for synthesising chemical compounds), procedures for experiments and tests, results of experimentation and testing (including pharmacological, biological, chemical, biochemical, clinical test data and data resulting from non-clinical studies), results of research and development including laboratory' records and data analyses. Information in a compilation or a compilation of information may be Know How notwithstanding some or all of its individual elements are in the public domain.
- "Licensed Product" means any product: (i) containing or comprising any CRT/ICR Existing Compound; and/or (ii) discovered, developed and/or generated using or incorporating any part of the CRT/ICR Existing Intellectual Property, including any metabolites, prodrugs, salts, hydrates, solvates, esters, intermediates, polymorphs, isomers, analogues and derivatives of any CRT/ICR Existing Compounds.
 - "Losses" has the meaning given in Clause 7.1.
- "Materials" means any chemical or biological substances including any: organic or inorganic element or compound; nucleotide or nucleotide sequence including DNA and RNA sequences; gene; vector or construct including plasmids, phages, bacterial vectors, bacteriophages and viruses; host organism including bacteria, fungi, algae, protozoa and hybridomas; eukaryotic or prokaryotic cell line or expression system or any development strain or product of that cell line or expression systems; protein including any peptide or amino acid sequence, enzyme, antibody or protein conferring targeting properties and any fragment of a protein or a peptide enzyme or antibody; drug or pro-drug; assay or reagent; any other genetic or biological material or micro-organism or any transgenic animal; and any physical property rights relating to any of the foregoing.
- "Non-Compound Intellectual Property" means (a) that part of the CRT/ICR Existing Intellectual Property which is not Compound Intellectual Property, and (b) includes without limitation CRT/ICR Existing Intellectual Property that consists of, contains or relates to [***].
 - "Parties" means CRT, the ICR and the Company and "Party" shall mean any of them.
- "*Patents*" means any patent applications, patents, author certificates, inventor certificates, utility' models, together with any divisionals, renewals, continuations, continuations-in-part, extensions, re-examinations, reissues, substitutions, confirmations, registrations, revalidations, and additions of or to them, as well as any Supplementary Protection Certificate, or any like form of protection, anywhere in the world for each of the foregoing.

- "*Phase I Trial*" means a human clinical trial in which a Licensed Product is administered to human subjects at multiple dose levels with the primary purpose of determining safety, metabolism, pharmacokinetic and pharmacodynamic data of the Licensed Product and consistent with 21 CFR § 312.21(a).
 - "Programme" has the meaning given in the Collaboration and Option Agreement.
 - "Programme Intellectual Property" has the meaning given in the Collaboration and Option Agreement.
 - "Protein Degradation Product" has the meaning given in the Collaboration and Option Agreement.
 - "Senior Executive Team" means [***].
 - "Shareholders' Agreement" has the meaning given in the recitals.
- "Sub-Licensee" means a person to which a sub-licence is granted in accordance with Clause 2.2 l,, respect of the whole or any part of the rights granted under this Agreement.
- "Supplementary Protection Certificate" means a right based on a patent pursuant to which the holder of the right is entitled to exclude third parties from using, making, having made, selling or otherwise disposing or offering to dispose of, importing or keeping the product to which the right relates, such as supplementary protection certificates in Europe, and any similar right anywhere in the world.
 - "Technology Access Fee" means [***].
 - "Term" has the meaning set forth in the Collaboration and Option Agreement.
 - "Territory" means worldwide.
 - "Third Party" means a person other than a Party or an Affiliate of a Party.
 - "Third Party Service Provider" has the meaning given in the Collaboration and Option Agreement.
- "*Tobacco Party*" means: (i) any person who develops, sells or manufactures tobacco products; and/ or (ii) any person which makes the majority of its profits from the importation, marketing, sale or disposal of tobacco products. Furthermore, Tobacco Party shall include any person that is Controlled by or under common Control with any of the persons referred to in subsections (i) and/or (ii) of the immediately preceding sentence.
 - "Transaction Documents" means the Formation and Investment Agreement, Shareholders' Agreement and Collaboration and Option Agreement.

1.2 In this Agreement:

- (a) unless the context otherwise requires, all references to a particular Clause, paragraph or Schedule shall be a reference to that clause, paragraph or schedule, in or to this Agreement as it may be amended from time to time pursuant to this Agreement;
 - (b) the table of contents and headings are inserted for convenience only and shall be ignored in construing this Agreement;
- (c) unless the contrary intention appears, words importing the masculine gender shall include the feminine and vice versa and words in the singular include the plural and vice versa;
- (d) unless the contrary intention appears, words denoting persons shall include any individual, partnership, company, corporation, joint venture, trust association, organisation or other entity, in each case whether or not having separate legal personality;
- (e) reference to the words "include" or "including" are to be construed without limitation to the generality of the preceding words, and, for clarity, the words "include" or "including" shall be deemed to be followed by the phrase "without limitation" or like expression; and
- **(f)** reference to any statute or regulation includes any medication or re-enactment of that statute or regulation provided that the modification or re-enactment does not diminish the rights or extend the obligations of any Party.

2. LICENCES

- **2.1** Subject to the provisions of this Agreement (including the research rights retained by the ICR and CRT pursuant to Clause 2.5) and Clauses 8.5 and 20.2 of the Collaboration and Option Agreement, each of CRT and the ICR hereby grants to the Company:
- (a) an exclusive, fully-paid, irrevocable, perpetual licence, with the right to grant sub-licences as provided in Clause 2.2, under all of CRT's and the ICR's respective rights in and to the Compound Intellectual Property for all purposes, including to discover, research, develop, have developed, use, keep, make, have made, market, import, offer for sale, sell and otherwise dispose of Licensed Products in the Field in the Territory; and
- **(b)** an exclusive fully-paid, irrevocable, perpetual licence, with the right to grant sub-licences as provided in Clause 2.2, under all of CRT's and the ICR's respective rights in and to the Non-Compound Intellectual Property listed in Schedule 1 A(ii) and I B(ii), and the CRT/ICR Background Hit Intellectual Property listed in Schedule 2 Part B, to discover, research, develop, have developed, use, keep, make, have made, market, import, offer for sale, sell and otherwise dispose of Protein Degradation Products in the Field in the Territory;
- (c) a non-exclusive fully-paid irrevocable, perpetual licence, with the right to grant sub-licences as provided in Clause 2.2, under all of CRT's and the ICR's respective rights in and to the Non-Compound Intellectual Property to discover, research, develop, have developed, use, keep, make, have made, market, import, offer for sale, sell and otherwise dispose of Licensed Products other than Protein Degradation Products in the Field in the Territory; and

- (d) a non-exclusive fully-paid, irrevocable, perpetual licence, with the right to grant sub-licences as provided in Clause 2.2, under all of CRT's and the ICR's respective rights in and to the CRT/ICR Background Hit Intellectual Property listed in Schedule 2 Part A to discover, research, develop, have developed, use, keep, make, have made, market, import, offer for sale, sell and otherwise dispose of Protein Degradation Products in the Field in the Territory.
- **2.2** The Company and the Sub-Licensees shall be entitled **to** grant sub-licences **in** respect of the rights granted under this Agreement, provided that:
- (a) the Company shall ensure that there are included in the terms of any sublicence like obligations and undertakings on the part of the Sub-Licensee as are contained in this Agreement and are applicable to Sub-Licensees, including Clause 10 (confidentiality), Clause 3 (performance). Clause 7 (indemnity') and Clause 13.1 (CRT's and the ICR's rights on termination) and shall further ensure that all Sub-Licensees duly comply with the same;
 - **(b)** no sub-licence shall be granted to a Tobacco Party;
- (c) within [***] of the grant of any sub-licence, the Company shall provide CRT and the ICR with a summary of the terms of such sub-licence at the Company's expense, and the Company shall have the right to exclude from such summary any terms that are not reasonably necessary for CRT and the ICR to assess the compliance of such sub-license with Clause 2.2(a).

The foregoing obligations shall not apply in relation to contracts the Company or its Sub-Licensees enters into with Third Party' Service Providers, provided that such contracts relate to the provision of research, development and/or manufacturing services to the Company in connection with the Company's Licensed Products, and no sub-license under CRT/ICR Existing Intellectual Property is granted to such Third Party to: (i) research, develop or manufacture its own Licensed Products; or(ii) sell the Company's or its Sub-Licensees' Licensed Products.

- **2.3** Any breach of Clause 2.2 shall be deemed to be a material breach.
- **2.4** The grant of any sub-licence shall be without prejudice to the Company's obligations under this Agreement. Any act or omission of any Sub-Licensee which, if it were the act or omission of the Company would be a breach of any of the provisions of this Agreement, will be deemed to be a breach of this Agreement by the Company who will be liable to CRT and the ICR accordingly.
- **2.5** CRT and the ICR retain rights from the licences under Clause 2.1(a) and (b) under the CRT/ICR Existing Intellectual Property and the CRT/ICR Background Hit Intellectual Property listed in Schedule 2 Part B solely to carry out non-commercial academic research and teaching and to conduct any and all activities allocated to CRT and/or the ICR under the Programme (in each case, subject always to [***]).
- **2.6** No Party may use or practice the CRT/ICR. Existing Intellectual Property except as expressly permitted by this Agreement. Except as explicitly set forth in this Agreement, no Party shall be deemed by estoppel or implication to have granted any other Party any license or other right to any intellectual property of such Party and each Party reserves all rights not otherwise expressly granted hereunder.

3. PERFORMANCE

The Company shall use Commercially Reasonable Efforts to conduct the Programme in accordance with the terms and conditions of the Collaboration and Option Agreement.

4. CONSIDERATION

In consideration of the rights and licences granted under this Agreement to the Company by each of CRT and the ICR, the Company shall:

- (a) issue common shares of the Company to each of CRT and the ICR pursuant to the provisions of the Formation and Investment Agreement;
- **(b)** pay to CRT [***] percent ([***]%) of the Technology Access Fee within [***] of the registration of the FORMATION (as defined in the Formation and Investment Agreement) in the appropriate commercial register; and
- **(c)** pay to the ICR [***] percent ([***]%) of the Technology Access Fee within [***] of the registration of the FORMATION (as defined in the Formation and Investment Agreement) in the appropriate commercial register. Such payment shall be made directly to the ICR and sent to [***]; Account name: [***]; Account number: [***]; Sort code: [***].

Payment of the Technology Access Fee in accordance with Clauses 4(b) and 4(c) shall be subject to the payment terms set out in clauses 10.1, 10.3, 10.4, 10.5 and 10.6 of the Collaboration and Option Agreement, provided that such payment shall be made in [***] and clauses 10.3, 10.4, 10.5 and 10.6 of the Collaboration and Option Agreement shall apply to payments to both CRT and the ICR.

5. INTELLECTUAL PROPERTY PROTECTION, PROCEEDINGS AND COSTS

- **5.1** Subject to Clauses 5.2, 5.3 and 5.4, the Company shall have the first right, but not the obligation, for preparing, filing, prosecuting and maintaining (including any patent interference, reexamination, *inter partes* review, post-grant review, reissue, revocation, opposition and appeal proceedings) any CRT/ICR Existing Patents in the joint names of CRT and ICR, and in the name of the Company to the extent such CRT/ICR Existing Patent includes claims for inventions that are not within the CRT/ICR Existing Intellectual Property, at Company's own cost and expense with the aim of maximizing duration and scope thereof, which, during the Term, will be in accordance with the strategy agreed by the JSC. CRT and the ICR shall provide the Company all reasonable assistance and cooperation requested in the CRT/ICR Existing Patent preparation, filing, prosecution and maintenance efforts provided in this Clause 5.1, including providing any reasonably necessary' and suitably limited powers of attorney and assignments and executing any other required documents or instruments for such filing, prosecution and maintenance.
- **5.2** The Company shall keep CRT and the ICR reasonably informed as to the prosecution and maintenance status of any such CRT/ICR Existing Patents and shall at CRT's request promptly provide CRT and ICR with a copy of all material submissions made to or responses received from the relevant patent offices and all material correspondence to and responses received from the relevant patent agent in relation to the CRT/ICR Existing Patents in each applicable country of the Territory. Company shall notify CRT and ICR at least [***] prior to any restriction of scope of any of the CRT/ICR Existing Patents. At Company's sole discretion, Company shall be at liberty to instruct its patent agent to copy CRT and the ICR into correspondence sent by the patent agent in which case Company shall be deemed to have complied with this Clause 5.2.

- **5.3** If Company is aware of and elects not to file a patent application claiming a patentable invention that is not commercially sensitive to Company (as valuable CRT/ICR Existing Know How which the Company believes reasonable is best protected through remaining confidential prior to any first publication by the ICR), Company shall promptly notify' CRT and the ICR of such decision and CRT (and if CRT declines, the ICR) shall have the right (but not the obligation) to file such an application. If CRT (or if CRT declines, the ICR) elects to exercise such right by notice in writing to the Company, CRT (or if CRT declines, the ICR) shall thereafter be solely responsible for the (expense of filing, prosecuting and maintaining the corresponding Patent, which shall be excluded from the licence granted under Clause 2.1.
- **5.4** If Company elects to stop the prosecution or maintenance of any part of the CRT/ICR Existing Patents, Company shall notify CRT and ICR in writing of such decision, but no later than [***] prior to the expiration of any applicable time bars. During the aforementioned [***] notice period. Company shall retain the responsibility for the prosecution and maintenance of the CRT/ICR Existing Patent in question. On expiry of such notice period, unless Company demonstrates [***] (the "Commercial Delay Rationale"):
 - (a) the licence granted pursuant to Clause 2.1 shall terminate in respect of the CRT/ICR Existing Patents identified in such notice;
- **(b)** Company shall, at CRT's (of if CRT declines, ICR's) request, promptly transfer to CRT (or any person nominated by CRT, or if CRT declines, ICR) any and all documents and information in Company's control relating to such CRT/ICR Existing Patents; and
- (c) CRT (or if CRT declines, ICR) shall be free to prosecute or abandon such CRT/ICR Existing Patents at its sole discretion and to grant rights thereunder to any person without further reference to Company.

Should CRT and/or ICR dispute the Commercial Delay Rationale set out in the notice, the matter may be referred for determination in accordance with Clause 25.

- **5.5** Each Party will notify the other, in writing as soon as reasonably practicable after it becomes aware that any claim is made or threatened against CRT, the ICR, the Company or a Sub-Licensee (in the case of the latter, where this is made known to the Company) by any Third Party that the exercise by CRT, the ICR, the Company or a Sub-Licensee of the rights granted pursuant to this Agreement infringes any Patent or other rights of any Third Party.
- **5.6** In the event of the circumstances described in Clause 5.5 arising, the Company shall use Commercially Reasonable Efforts to take such steps as may be necessary in order to address as appropriate any such alleged infringement by the Company or a Sub-Licensee, and CRT and the ICR shall, at the Company's cost, give it all reasonable co-operation in this regard.

- **5.7** Each Party will notify the other in writing as soon as it becomes aware of unauthorised use of the CRT/ICR Existing Intellectual Property, and the Company shall have the sole right, but not the obligation to:
- (a) at its own cost and subject to Clause 5.8, bring proceedings in its own name or, if required by law, jointly with CRT and the ICR, for misappropriation of and/or unauthorized used the CRT/ICR Existing Intellectual Property; and
- **(b)** in any such proceedings settle any such claim, subject to the consent of CRT and the ICR, which shall not be unreasonably withheld, conditioned or delayed.
- **5.8** In any such proceedings, CRT and the ICR shall, at the Company's cost, promptly provide the Company with all documents and assistance as the Company may reasonably require, including joining in any such proceeding and/or providing any reasonably necessary' and suitably limited powers of attorney. The Company shall promptly provide CRT and the ICR with notice of such proceedings, keep CRT and the ICR regularly informed of progress and promptly provide CRT and the ICR with such information as CRT and/or ICR may reasonably require, including copies of all documents filed at court or other tribunal in such proceedings. Subject to Clause 12.6 of the Collaboration and Option Agreement, the Company shall be entitled to keep all monies recovered in such proceedings.
- 5.9 If the Company does not exercise its right to bring proceedings pursuant to Clause 5.7(a) within [***] of the written notification pursuant to Clause 5.7 then CRT (and if CRT declines, the ICR) may, but shall not be obliged, to bring such proceedings at its own cost. If necessary (including to recover damages), the Company shall join in such proceedings. After reimbursement of the Company's costs and expenses incurred in such proceedings, CRT or the ICR (as the case may be) shall be entitled to all monies recovered in such proceedings. In any such proceedings the Company, and the ICR or CRT (as the case may be), shall, at the CRT's or the ICR's costs (as the case may be) promptly provide CRT or the ICR (as the case may be) with all documents and assistance as CRT or the ICR (as the case may be) may reasonably require. CRT and/or the ICR (as the case may be) shall promptly provide the Company and CRT or the ICR (as the case may be) with notice of such proceedings, keep the Company and CRT or the ICR (as the case may be) regularly informed of progress and promptly provide the Company and CRT or the ICR (as the case may be) may reasonably require, including copies of all documents filed at court or other tribunal in such proceedings.

6. REPRESENTATIONS AND WARRANTIES

- 6.1 Each Party represents and warrants to the other Parties that:
 - (a) such Party has the full power and authority to enter into this Agreement and to carry out the provisions hereof; and
- **(b)** such Party has taken all necessary' action on its part to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder;

	(c) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, va	alid, binding obligat	ion,
enforceable ag	gainst it in accordance with the terms hereof;		

- (d) the execution, delivery and performance of this Agreement by such Party does not conflict in a material manner with any agreement or any provision thereof, or any instrument or written understanding to which it is a party or by which it is bound, nor violate any law or regulation of any court, governmental body or administrative or other agency having jurisdiction over such Party; and '
- (e) no government authorization, consent, approval, license, exemption of or filing or registration with any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, under any applicable laws, rules or regulations currently in effect, is or will be necessary' for the execution and delivery of this Agreement;
- (f) in respect of the subject-matter of this Agreement, it has not employed any individual or entity debarred by the FDA (or subject to a similar sanction of any equivalent Competent Authority outside the United States of America) or, to its knowledge, that is the subject of any FDA debarment investigation or proceeding (or similar proceeding of any equivalent Competent Authority outside the United States of America), it being understood that details of the subjects of FDA debarment investigations are not published and a Party would not make any particular enquiries of relevant individuals outside such Party's standard policies and procedures.
- **6.2** Each of CRT and the ICR represents and warrants to the Company that to the actual knowledge of its Senior Executive Team and of its representatives involved in the negotiation of this Agreement that, as of the Effective Date:
 - (a) such Party has sufficient legal title to grant the licences as purported to be granted pursuant to this Agreement as at the Effective Date;
 - **(b)** [***];
 - (c) [***]:
 - (d) [***];
 - (e) [***]; and
 - (f) [***];
 - (g) [***]; and
 - (h) [***].

Solely for the purposes of this Clause 6.2, "*Controlled*" means, with respect to any intellectual property, possession of the ability (whether through ownership or license) to grant the licenses or sublicenses as provided herein without violating the terms of any agreement or other arrangement with any third party.

6.3 Except as otherwise expressly set forth in this Agreement. NO PARTY MAKES ANY REPRESENTATION OR EXTENDS ANY WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY THAT EXERCISE OF ANY INTELLECTUAL PROPERTY RIGHT UNDER THIS AGREEMENT DOES NOT INFRINGE ANY INTELLECTUAL PROPERTY RIGHTS OF THIRD PARTIES, AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

6.4 No claim for breach of any of the representations or warranties in Clause 6 may be brought against CRT, the ICR or the Company after [***] has elapsed from the Services Expiry Date (as defined in the Collaboration and Option Agreement), after which date each Party shall be fully and completely discharged from any liability for breach of any of the representations and warranties in Clause 6. The Company's aggregate liability, and CRT's and the ICR's combined aggregate liability, for breach of one or more of the representations or warranties in this Clause 6 shall be subject to clause 13.4 of the Collaboration and Option Agreement.

7. INDEMNITY

7.1 The Company shall indemnify and hold harmless the Indemnified Parties from and against any and all liabilities, losses, damages, costs and expenses, including, without limitation, reasonable legal fees (collectively, "Losses") arising out of or resulting from any and all Third Party suits, claims, actions, proceedings, investigations or demands ("Claims") arising from or in connection with the exercise by the Company or a Sub-Licensee of the rights granted in Clause 2.1, or based upon or in connection with the development, manufacture or commercialization of any and all Licensed Products by or on behalf of the Company or any of its Sub-Licensees, except, in each case to the extent such Losses arise from (a) the gross negligence or willful misconduct of such Indemnitee: or (b) a material breach of any representation, warranty, covenant or obligations under this Agreement by CRT or the ICR.

7.2 Promptly after receipt by the Company of any claim or alleged claim or notice of the commencement of any action, administrative or legal proceeding, or investigation to which the indemnity provided for in this Clause 7 may apply, the Company shall give written notice to CRT and the ICR of such fact. In the event that any Indemnitee entitled to indemnification under Clause 7.1 is seeking such indemnification, such Indemnitee shall (a) give written notice to the Company of the Claim as soon as reasonably practicable after such Indemnitee receives notice of such Claim, (b) permit the Company to assume the direction and control of the defence of such Claim (including the sole right to settle it at the sole discretion of the Company, taking into consideration in good faith any reasonable concerns or objections raised by the Indemnitee; provided that such settlement does not impose any obligation on, or otherwise adversely affect, the Indemnitee or other Party), (c) cooperate as reasonably requested (at the expense of the Company) in the defence of such Claim, and (d) undertake all reasonable steps to mitigate any loss, damage or expense with respect to such Claim. If the Company fails to assume the defence thereof by providing the relevant Indemnitee notice of such election in writing within [***] after receipt by the Company of notice of such Claim, the Indemnitee may assume such defence and for the avoidance of doubt the indemnity in Clause 7.1 shall extend to the reasonable legal and other expenses consequently incurred in connection with such defence to the extent such Indemnitee is entitled to indemnification under Clause 7.1. The Company and the relevant Indemnitee will co-operate in good faith in the conduct of any defence, provide such reasonable assistance as may be required to enable any claim properly to be defended and the Party with conduct of the action shall provide promptly to the other Party copies of all correspondence and documents and notice in writing of the substance of all oral communicati

7.3 Should the Company assume conduct of the defence, the Indemnitee may retain separate legal advisers, at its sole cost and expense, save that if the Company wrongly denies the applicability of the indemnity or reserves its position in relation to the same, the indemnity in this Clause 7 shall extend to the Indemnified Party's reasonable costs and expenses so incurred:

8. INSURANCE

Reasonably prior to Commencement of a Phase 1 Trial, the Company shall put in place and thereafter maintain in accordance with the terms of this Clause 8, at its own cost, comprehensive product liability insurance and general commercial liability insurance. The Company shall ensure at CRT's and/or the ICR's request, that CRT and/or the ICR interest as co-assured be noted on the policy or policies. Within [***] of the Effective Date and of the beginning of each policy period, the Company shall, upon the reasonable request of CRT and/or the ICR, provide CRT and/or the ICR with a certificate evidencing the coverage required hereby, and the amount hereof and the noting of CRT and/or the ICR's interest. Such insurance shall be with a reputable insurance company and shall be maintained for not less than [***] following the expiration/termination of this Agreement for any reason or, if such coverage is of the "claims made" type, for [***] following the expiration or termination of this Agreement for any reason.

9. LIMITATION OF LIABILITY

9.1 EXCEPT AS SET FORTH IN CLAUSE 9.2, NO PARTY, NOR CRUK, NOR THEIR RESPECTIVE OFFICERS. EMPLOYEES AND AGENTS SHALL HAVE LIABILITY WHETHER UNDER STATUTE OR IN TORT (INCLUDING BUT NOT LIMITED TO NEGLIGENCE), CONTRACT OR OTHERWISE TO ANY OTHER PARTY AND/OR CRUK IN RESPECT OF ANY CONSEQUENTIAL, INDIRECT OR PUNITIVE DAMAGES IN CONNECTION WITH THIS AGREEMENT. INCLUDING, WITHOUT LIMITATION, LOSS OF GOODWILL, OPPORTUNITY, PROFIT OR CONTRACT; PROVIDED THAT, BECAUSE MONETARY DAMAGES MAY NOT ADEQUATELY COMPENSATE A PARTY FOR A BREACH OF THE CONFIDENTIALITY PROVISIONS OF CLAUSE 10 BY ANY OTHER PARTY AND REMEDIES UNDER CLAUSE 10.2 MAY BE INADEQUATE OR UNAVAILABLE WITH RESPECT TO SUCH A BREACH, THE FOREGOING LIMITATION ON LIABILITY WILL NOT LIMIT A PARTY'S RIGHT TO RECOVER FROM ANY OTHER PARTY CONSEQUENTIAL OR INDIRECT DAMAGES (BUT NOT PUNITIVE DAMAGES) SUSTAINED AS A RESULT OF BREACH BY SUCH OTHER PARTY OF THE CONFIDENTIALITY PROVISIONS OF CLAUSE 10, BUT TO THE EXTENT ANY SUCH DAMAGES ARISE AS A RESULT OF A BREACH OF CLAUSE 10, THE BREACHING PARTY'S LIABILITY FOR ANY SUCH DAMAGES SHALL BE LIMITED TO [***] AND CRT'S AND THE ICR'S LIABILITY FOR ANY SUCH DAMAGES SHALL BE LIMITED TO [***] AND CRT'S AND THE ICR'S LIABILITY FOR ANY SUCH DAMAGES SHALL BE SEVERAL AND NOT JOINT.

9.2 Nothing in this Agreement shall be construed as excluding or limiting the liability of any Party or CRUK or any of their respective officers, employees and agents to the other Party for death or personal injury of any person resulting from the negligence or willful misconduct of such persons.

10. CONFIDENTIALITY

10.1 The Parties acknowledge and agree that any information furnished by or on behalf of a Party to the other Parties pursuant, to this Agreement shall collectively constitute "Confidential Information" of such Party for purposes of the Collaboration and Option Agreement, including Clause 17 thereof, and the provisions of Clause 17 of the Collaboration and Option Agreement are hereby incorporated by reference in this Agreement, *mutatis mutandis*.

10.2 Given the nature of the Confidential Information and the competitive damage that a Party' would suffer upon unauthorized disclosure, use or transfer of its Confidential Information to any Third Party, the Parties agree that monetary damages would not necessarily be a sufficient remedy for any breach of this Clause 10. In addition to all other remedies, a Party shall be entitled to specific performance and injunctive and other equitable relief as a remedy for any breach or threatened breach of this Clause 10.

11. PUBLICATIONS

The Parties acknowledge and agree that any proposed publications of scientific papers containing Programme Intellectual Property shall be governed by, Clause 18 of the Collaboration and Option Agreement, and the provisions of Clause 18 of the Collaboration and Option Agreement are hereby incorporated by reference in this Agreement, *mutatis mutandis*.

12. TERM AND TERMINATION

- 12.1 This Agreement is binding upon the Parties as of the Execution Date, and Clauses 9-27 are effective as of the Execution Date, but no other provision of this Agreement shall be effective as of the Execution Date or ever become effective except as expressly provided in this Clause 12.1. The remainder of this Agreement (i.e., those provisions in addition to those that are effective as of the Execution Date) shall become effective on the date of both CRT's and ICR's receipt of payment of the Technology Access Fee pursuant to Clauses 4(b) and 4(c) the ("Effective Date") and shall remain in full force and effect until terminated in accordance with the provisions of this Clause 12.
- **12.2** This Agreement may be terminated by written agreement of the Parties, with the effects of any such termination as set forth Clause 13 and as otherwise agreed to in writing by the Parties.
- **12.3** If the Effective Date has not occurred within [***] of the Execution Date, then, effective as of 11:59 p.m., London time, on such [***] after the Execution Date, this Agreement shall automatically become null and void *ab initio*. In the event that CRT's and the ICR's subscription amounts are to be returned in accordance with section 2.9 of the Formation and Investment Agreement, [***] and this Agreement shall automatically terminate.

13. EFFECTS OF TERMINATION

13.1 The termination of this Agreement howsoever arising, will be without prejudice to the rights, obligations and duties of any Party accrued prior to termination. The following Clauses will continue to be enforceable notwithstanding termination: Clauses 2, 6.3, 7, 8, 9, 10, and 13-27.

14. FORCE MAJEURE

- **14.1** No Party shall be liable for any delay or failure in performance if such delays are caused by strike, riot, civil commotion, fire, acts of God or other circumstances beyond its reasonable control which circumstance that Party could not reasonably have been expected to take into account at the Effective Date and provided that:
 - (a) the Party so affected shall give prompt notice thereof to the other Parties;
 - **(b)** the suspension of performance is no greater scope than is required by the Force Majeure: and
 - (c) lack of funds shall not be interpreted as an event or circumstance beyond the reasonable control of a Party.
- **14.2** Subject to Clause 14.3 below, the Party giving such notice shall be excused from such of its obligations hereunder for as long as it continues to be so affected and shall perform its obligations as soon as such circumstances shall cease to affect its operations.
- **14.3** If such force majeure continues unabated for a period of at least [***], the Parties will meet to discuss in good faith what actions to take or what modifications should be made to this Agreement as a consequence of such force majeure in order to alleviate its consequences on the affected Parties.

15. ASSIGNMENT

- **15.1** This Agreement shall be binding upon and inure to the benefit of the Parties, their permitted successors and assigns. This Agreement shall only be assignable:
- (a) by any Party to any of Affiliate of such Party without the consent of the other Parties provided that such Affiliate is not a Tobacco Party;
 - (b) by any Party' with the written consent of the others; or
- (c) by any Party without the consent of the other Parties, to any successor to all or substantially all the assets of its business to which this Agreement relates provided that it is not a Tobacco Party.

Any purported assignment in violation of this Clause 15.1 shall be null and void.

16. NOTICES

All notices shall be in writing and sent by hand, facsimile, or post and shall be deemed to be properly served (i) if sent by hand on [***], when delivered at the relevant address and if delivered at any other time, on [***]; (ii) if sent by post, [***] after posting; (iii) if sent by facsimile on [***], when transmitted and if sent at any other time, on [***], provided a confirmatory copy is sent by post within [***] of transmission, and shall be sent to the following addresses or facsimile numbers as may be amended by the relevant Party in writing:

The Company: Monte Rosa Therapeutics AG

Aeschevnorstadt 36

4051 Basel Switzerland Facsimile: [***]

For the attention of: Chief Executive Officer

CRT: Cancer Research Technology Limited

Angel Building 407 St John Street London, EC IV 4AD

England
Facsimile: [***]

For the attention of: The Chief Executive Officer

The ICR: Institute of Cancer Research

15 Cotswold Road Belmont, Sutton Surrey SM2 5NG

England

Facsimile: [***]

For the attention of: Director of Enterprise

17. VARIATION

No variation, modification, amendment, extension, or release of any provision hereof shall be effective unless it is in writing, signed by the Parties.

18. ENTIRE AGREEMENT

18.1 Each of the Parties confirms that the Agreement (including all Schedules hereto), together with the Transaction Documents, represents the entire understanding, and constitutes the whole agreement, in relation to its subject matter and supersedes any previous agreement between the Parties with respect thereto. In the event of any conflict or discrepancies between the provisions of this Agreement (on the one hand) and the Articles (as defined in the Transaction Documents) or any description of the terms of this Agreement included in any statutory' report that may be required under Swiss law (on the other hand), the provisions of this Agreement shall prevail.

18.2 Each Party confirms that:

- (a) in entering into this Agreement it has not relied on any representation or warranty' or undertaking which is not contained in this Agreement; and
- **(b)** in any event, without prejudice to any liability for fraudulent misrepresentation or fraudulent misstatement, no Party shall be under any liability or shall have any remedy in respect of misrepresentation or untrue statement unless and to the extent that a claim lies under this Agreement or any Transaction Document.

19. FURTHER ASSURANCE

The Parties hereby undertake to do all such other acts and things, and execute and provide all such documents at the requesting Party 's cost as may be necessary or desirable to give effect to the purposes of this Agreement.

20. WAIVER

No waiver, release, relaxation, forbearance or indulgence by any Party' in enforcing any of the terms or conditions of this Agreement or the granting of time by any Party to any other shall prejudice, affect or restrict the rights and powers of such Party. The waiver of any breach of any term or any condition of this Agreement shall not be construed as a waiver of any subsequent breach of a term or condition of the same or of a different nature. Any waiver by a Party of a particular provision or right shall be in writing, shall be as to a particular matter and, if applicable, for a particular period of time and shall be signed by such Party.

21. SEVERABILITY

- **21.1** If the whole or any part of this Agreement is or becomes or is declared illegal, invalid or unenforceable in any jurisdiction for any reason (including by reason of the provisions of any legislation and/or by reason of any court of competent jurisdiction):
- (a) in the case of the illegality, invalidity or unenforceability of the whole of this Agreement it shall terminate only in relation to the jurisdiction in question; or
- **(b)** in the case of the illegality', invalidity or unenforceability of a part of this Agreement that part shall be severed from this Agreement in the jurisdiction in question and that illegality, invalidity or unenforceability shall not in any way whatsoever prejudice or affect the remaining parts of this Agreement which shall continue in full force and effect and in no circumstances shall sums paid by the Company to CRT or the ICR under this Agreement, if any, be repayable.
- **21.2** If in the reasonable opinion of any Party any severance under this Clause 21 materially affects the commercial basis of this Agreement, the Parties shall discuss, in good faith, ways to eliminate the material effect.

22. LAW AND JURISDICTION

This Agreement shall be governed by and construed in accordance with the laws of England and Wales, and the Parties agree, subject to Clause 25 below, to submit to the exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this Agreement (except in respect of disputes under Clause 10 where jurisdiction is nonexclusive).

23. EXECUTION

This Agreement may be executed in any one or more number of counterpart agreements each of which, when executed, shall be deemed to form part of and together constitute this Agreement. Facsimile signatures and signatures transmitted via PDF shall be treated as original signatures.

24. ANNOUNCEMENTS AND USE OF NAMES

- **24.1** Save as provided in Clause 24.2 no Party shall make, or procure or permit the making of, any press release or other public announcement in relation to this Agreement without first obtaining the written approval of the other Parties to any such release or announcement, which shall not unreasonably be withheld, conditioned or delayed.
- **24.2** Each Party agrees that it may make any announcement with respect to this Agreement or any ancillary matter as shall be required by law or the regulations of any stock exchange to which it is subject, without the other Parties' consent (which shall not unreasonably be withheld, conditioned or delayed) provided it has used reasonable endeavours in the time available to consult with the other Parties on the terms of any such announcement beforehand.
- **24.3** No Party shall use the name of the other (including in the case where the other is CRT, that of CRUK (or its successor)) other than as provided in Clause 24.1 and 24.2 without the prior written consent of such Party, which shall be at such Party's sole discretion. Notwithstanding the foregoing, (a) to the extent information regarding this Agreement has already been publicly disclosed in accordance with Clause 24.1 and 24.2, a Party may subsequently disclose the same information to the public without the consent of the other Parties, and (b) the Company shall also be permitted to disclose the terms of this Agreement, in each case under appropriate confidentiality provisions substantially equivalent to those of this Agreement, to any actual or potential acquirers, investors. Sub-Licensees, collaborators, and/or professional advisors.

25. DISPUTE RESOLUTION

It shall be a condition precedent to the commencement of any action in court or other tribunal (save an action for specific performance, injunctive or other equitable relief in accordance with Clause 10.2) in respect of any dispute relating to this Agreement that the Parties have sought to resolve the dispute by one or more Parties notifying it in writing for resolution to the Executive Officers of the Parties who shall meet (whether in person or via teleconference) within [***] of such notice to seek resolution in good faith.

26. NO THIRD PARTY BENEFICIARIES

Except for the Third Parties and the respective rights referred to in Clause 7 (Indemnity) and Clause 9 (Limitation of Liability) this Agreement is neither expressly nor impliedly made for the benefit of any party other than those executing it whether pursuant to the Contract (Rights of Third Parties) Act 1999 or otherwise. Notwithstanding the provisions of this Clause 26, the Parties shall be entitled to amend, suspend, cancel or terminate this Agreement or any part of it in accordance with Clause 17, without the consent of any Third Party including those referred to in this Clause 26.

27. LANGUAGE; AUTHORSHIP

This Agreement has been prepared in the English language, and the English language shall control its interpretation. In addition, all notices required or permitted to be given hereunder, and all written, electronic, oral of other communications among the Parties regarding this Agreement, shall be in the English language. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties, and no presumption or burden of proof shall arise or disfavoring any Party by virtue of the authorship of any provisions of this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized representatives as of the Effective Date.

CANCER RESEARCH TECHNOLOGY LIMITED

Director: /s/ Illegible
Print Name: Illegible
Date: April 19, 2018

THE INSTITUTE OF CANCER RESEARCH: ROYAL CANCER HOSPITAL

Authorized Signatory: /s/ Illegible

Print Name: Illegible

Date: April 19, 2018

MONTE ROSA THERAPEUTICS AG

By: /s/ Bradley Bolzon
Director: President of Board
Print Name: Bradley Bolzon
Date: April 26, 2018

[Signature page to License Agreement]

SCHEDULE 1

A - CRT/ICR EXISTING KNOW HOW [***]

B - CRT/ICR EXISTING MATERIALS

[***]

C – CRT/ICR EXISTING COMPOUNDS

[***]

SCHEDULE 2

CRT/ICR Background Hits

[***]

Certain identified information has been excluded from this exhibit because it is both not material and is the type that the registrant treats as private or confidential. Information that was omitted has been noted in this document with a placeholder identified by the mark "[***]".

COLLABORATION AND OPTION AGREEMENT

(1) CANCER RESEARCH TECHNOLOGY LIMITED

AND

(2) THE INSTITUTE OF CANCER RESEARCH

AND

(3) MONTE ROSA THERAPEUTICS AG

April 10, 2018

TABLE OF CONTENTS

	ANTER DE ETATION	Page
	INTERPRETATION	2
	JSC	12
-	PROGRAMME	15
	RESEARCH FUNDING	16
-	ICR INTERNAL RESEARCH	17
	OWNERSHIP OF INTELLECTUAL PROPERTY	21
7.	LICENCES	22
8.	PERFORMANCE	25
9.	CONSIDERATION	27
10.	. PAYMENT AND STATEMENT	28
11.	. ACCOUNTS	29
12.	. INTELLECTUAL PROPERTY PROTECTION, PROCEEDINGS AND COSTS	30
13.	. REPRESENTATIONS, WARRANTIES AND COVENANTS	32
14.	. INDEMNITY ·	33
15.	. INSURANCE	34
16.	. LIMITATION OFLIABILITY	35
17.	. CONFIDENTIALITY	35
18.	. PUBLICATIONS	37
19.	. TERM AND TERMINATION	38
20.	. EFFECTS OF TERMINATION	39
21.	. FORCE MAJEURE	40
22.	. ASSIGNMENT AND SUB-CONTRACTING	41
23.	. NOTICES	42
24.	. VARIATION	42
25.	. ENTIRE AGREEMENT	42
26.	. FURTHER ASSURANCE	43
27.	. WAIVER	43
28.	. SEVERABILITY	43
29.	. LAW AND JURISDICTION	44
30.	. EXECUTION	44

TABLE OF CONTENTS (continued)

	r age
31. ANNOUNCEMENTS AND USE OF NAMES	44
32. DISPUTE RESOLUTION	44
33. NO THIRD PARTY BENEFICIARIES	45
34. LANGUAGE; AUTHORSHIP	45

COLLABORATION AND OPTION AGREEMENT

THIS COLLABORATION AND OPTION AGREEMENT is entered into and made effective as of the 10th day of April, 2018 (the "Effective Date"), by and among:

- **(1) CANCER RESEARCH TECHNOLOGY LIMITED**, a company registered in England and Wales under number 1626049 with registered office at Angel Building, 407 St John Street, London, ECIV 4AD, England ("*CRT*");
- (2) THE INSTITUTE OF CANCER RESEARCH: ROYAL CANCER HOSPITAL, a company limited by guarantee and registered in England (registered number 00534147) and registered charity whose office is at 123 Old Brompton Road, London, SW17 9LN, England ("ICR"); and
- **(3) MONTE ROSA THERAPEUTICS AG** (in formation), a company to be organized under the laws of Switzerland, with registered office at Aeschenvorstadt 36, 4051 Basel, Switzerland ("*Company*"). CRT, the ICR and the Company are each referred to herein by name or as a "*Party*" or, collectively, as the "*Parties*".

WHEREAS:

- **(A)** CRT is an oncology focused technology transfer and development company, wholly owned by Cancer Research UK ("*CRUK*"), a company limited by guarantee (registered in England and Wales under number 4325234) and a charity (registered in England under number 1089464 and registered in Scotland under number SC041666) of Angel Building, 407 St John Street, London, EC1V 4AD, United Kingdom;
- **(B)** The ICR is a registered charity and a college of the University of London, and is dedicated to the pursuit of scientific and clinical research into the understanding, diagnosis and treatment of cancer;.
 - (C) The Company is engaged in the research and development of protein degradation-based therapeutics;
- **(D)** As a condition to and concurrent with the execution and delivery of this Agreement, the Parties are entering into that certain (a) License Agreement (as may be amended, the "License Agreement"), pursuant to which CRT and the ICR grant the Company certain rights under their intellectual property rights existing as of the Effective Date in the field of cereblon (CRBN)-mediated protein degradation, and (b) Formation and Investment Agreement (as may be amended, the "Formation and Investment Agreement"), and the related Shareholders' Agreement (as may be amended, the "Shareholders' Agreement"), by and among the Company and its shareholders, including CRT and the ICR, pursuant to which the Company issues common shares of the Company as consideration for the rights granted to it under the License Agreement; and
- **(E)** As a condition to the execution and delivery of the Formation and Investment Agreement, Shareholders' Agreement and the License Agreement, the Company wishes to collaborate with CRT and the ICR, and CRT and the ICR are willing to collaborate, on the research and development in the field of CRBN-mediated protein degradation on terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the foregoing, the covenants and premises contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

1. INTERPRETATION

- 1.1 In this Agreement the following words and expressions shall have the following meanings:
- "*Affiliate*" means, with respect to a Party, any person that, whether now or in the future, Controls, is Controlled by or is under common Control with a Party. For the avoidance of doubt, Versant Ventures shall not be deemed an Affiliate of Company.
- "Affordable Price" means in relation to a Product: (i) a determination by the UK Pricing Authority that such Product should be used within the NHS; and/or (ii) approval by the UK Pricing Authority of the price proposed by the company or its Sub-Licensee in relation to sales of that Product in the United Kingdom (or one or more constituent countries thereof).
- "Agreement" means this Collaboration and Option Agreement and each of the Schedules attached hereto, as amended from time to time in accordance with the provisions of this Collaboration and Option Agreement.
- "Business Day" means a day other than a Saturday, Sunday, or any public holiday in England or any date on which commercial banks located in Switzerland, and/or New York, New York, U.S.A. are authorized or required by law to close.
 - "Claims" has the meaning given in Clause 14.1.
- "Commencement" means, in relation to a clinical trial, the date upon which administration of a Product to the first human subject has occurred, whether such subject is a healthy volunteer or a patient.
 - "Commercially Reasonable Efforts" means [***].
 - "Company Compound Intellectual Property" means [***].
 - "Company Non-Compound Intellectual Property" means [***].
- "Competent Authority" means any local or national agency, authority, department, inspectorate, minister, ministry official or public or statutory person (whether autonomous or not) of or of any government of any country having jurisdiction over the Agreement or any of the Parties or over the development or marketing of medicinal products including, the FDA, the European Medicines Agency, the European Commission and the European Court of Justice.
 - "Compound" means a CRT/ICR Existing Compound or a Programme Compound.
- "Compound Library" means (a) the CRT/ICR Existing Compound Library or any other compound library generated by or on behalf of the Company and/or the ICR pursuant to the conduct of Programme, but (b) excluding any compounds comprised in any CRT/ICR Background Library, other than Duplicate Compounds which remain Programme Compounds pursuant to Clause 7.6(b).

"Confidential Information" means all CRT/ICR Existing Know How, CRT/ICR Existing Materials, Programme Materials, Programme Know How and research and development plans, and all other information, in tangible or non-tangible form (including oral disclosure), including information relating to the customers, suppliers, business partners, clients, finances, business plans, technology, research and products (in each case actual or prospective) of a Party, the terms of this Agreement, and any other technical or business information (whether or not marked as confidential). CRT/ICR Existing Know How and CRT/ICR Existing Materials shall be deemed the Confidential Information of CRT and ICR. Programme Materials and Programme Know How shall be deemed the Confidential Information of the Company.

"Control" means the possession (directly or indirectly) of fifty percent (50%) or more of the voting stock or other equity interest of a subject entity with the power to vote, or the power in fact to control the management decisions of such entity through the ownership of securities or by contract or otherwise and "Controlling", "Controls", "Controlled by" and "under common Control with" as used with respect to any party shall be construed accordingly.

"CRBN" means cereblon.

"CRT/ICR Background Hit Compound" means [***].

"CRT/ICR Background Library" means [***].

"CRT/ICR Background Library Intellectual Property" means [***].

"CRT/ICR Existing Compound" has the meaning given in the License Agreement.

"CRT/ICR Existing Compound Library" has the meaning given in the License Agreement.

"CRT/ICR Existing Intellectual Property" has the meaning given in the License Agreement.

"CRT/ICR Existing Know How" has the meaning given in the License Agreement.

"CRT/ICR Existing Materials" has the meaning given in the License Agreement.

"CRT/ICR Reviewers" means independent persons nominated by CRT or Cancer Research UK for the purpose of monitoring and reviewing work funded by Cancer Research UK and/or providing scientific advice.

"CTU" has the meaning given in the License Agreement.

"Currency" means United States dollars.

"Data Package" means an information package prepared by the ICR comprising pre- clinical data and other information generated in a Non-Degradation Program (including all compounds discovered or developed pursuant to the conduct of such program), including each of the following to the extent available: (a) copies of all material data with respect to each such program, including material data from discovery, research and pre-clinical programs; (b) a reasonably detailed description of the intellectual property rights owned or controlled by CRT and/or the ICR with respect thereto; (c) copies of all reports prepared by or for ICR's Cancer Therapeutic Unit solely for such program; (d) any known risks or uncertainties relating to such program; and (e) copies of any written correspondence with Competent Authorities with respect to each such program.

"Disclosing Party" has the meaning given in Clause 17.1.

"Duplicate Compound" has the meaning given in Clause 7.6.

"Effective Date" has the meaning given in the preamble.

"Encumbrance Date" shall have the meaning set forth in Clause 7.6(b).

"Executive Officers" means an authorised executive officer of the Company, the Chief Executive Officer of CRT and the Director of Enterprise of the ICR or such other authorised officer of a Party as may be substituted from time to time upon the giving of written notice to the other Party.

"European Union" means European Union and/or (in the event that the United Kingdom is not part of the European Union) the United Kingdom.

"Expert" means a suitably qualified independent expert appointed by agreement among the Parties. However, in the event that the Parties are unable to reach agreement within [***] of either Party seeking in writing to the other to appoint such expert, each Party shall appoint one (1) expert and such experts shall together select a third expert within [***] after the last to occur of their respective appointments to resolve the dispute, who shall be the sole "Expert" for such dispute. If the two experts are unable to agree on a third expert within this period they shall submit two (2) names to the President for the time being of the Association of the British Pharmaceutical Industry (or any successor body thereto), who shall select an individual from the names submitted.

"Extended Exclusivity Period" means any period during which one of the following subsists in respect of a Product in a given country in the Territory: Orphan Drug Designation, paediatric designation or other exclusivity (excluding a Patent) granted by a Competent Authority beyond the expiry of the relevant Patent, which, in each case, confers exclusive marketing rights with respect to such Product in such country.

"FDA" means the United States Food and Drug Administration or any successor to it.

"Field" means the means the treatment, prevention and/or diagnosis of any and all diseases, disorders or conditions.

"First Commercial Sale" means, with respect to a Product and a country in the Territory, the first sale or other disposition for monetary value of such Product to a Third Party, after all relevant Regulatory Authorisations for the sale or other disposition for monetary value of such Product in such country have been obtained in respect of the relevant region or country.

"Force Majeure" means in relation to a Party any event or circumstance which is beyond the reasonable control of that Party, which event or circumstance that Party could not reasonably be expected to have taken into account at the Effective Date and which results in or causes the failure of that Party to perform any or all of its obligations under this Agreement including act of God, lightning, fire, storm, flood, earthquake, strike, lockout or other industrial disturbance, war, terrorist act, blockade, revolution, riot, insurrection, civil commotion, public demonstration, sabotage, act of vandalism, explosion, provided that lack of funds shall not be interpreted as a cause beyond the reasonable control of that Party.

"Formation and Investment Agreement" has the meaning given in the recitals.

"FTE" means a qualified full-time individual's work time dedicated by the ICR to work under the Programme, or in the case of less than a full-time dedicated individual, a full-time equivalent person year, based upon the ICR's typical employment contracts for a full-time individual in place at that time. As an example, under ICR's typical employment contract as of the Effective Date, a full-time individual's work time is based upon a total of [***] of work per week excluding allowances under standard ICR employment contracts for annual leave (e.g. holiday allowance, bank holidays) and other leave (e.g. sick leave, parental leave). Any individual who devotes less than the full-time individual work time set out in the ICR's typical employment contracts in place at that time shall be treated as an FTE on a pro-rata basis based upon the actual number of hours worked by that individual under this Agreement.

"ICR Internal Research" means the screening and research activities conducted by or on behalf of the ICR as described in Clause 5.1.

"ICR Internal Research Improvements" means [***].

"ICR Internal Research Program Hits" means any and all compounds from any Compound Library that are identified as having activity as a result of the screening conducted pursuant to the ICR Internal Research (as outlined in the research plan for each specific screening project in the ICR Internal Research).

"*ICR Internal Target*" means a target identified by or on behalf of the ICR as the biological target of an ICR Internal Research Program Hit as a result of screening (phenotypic or otherwise) the Compound Library and (if relevant) target deconvolution_ work conducted by the ICR under the ICR Internal Research (as outlined in the research plan for each specific ICR screening project that is provided by the ICR to the JSC).

"ICR Research Target IP" means [***].

"Indemnified Parties" means CRT, CRUK, the ICR and their respective officers, employees and agents and "Indemnitee" shall mean any of them.

"Indication" means [***].

"JSC" shall have the meaning given in Clause 2.1.

"Know How" means any technical, business or financial information and other information, of any type whatsoever, in any tangible or intangible form, which is not in the public domain as of the Effective Date, including, know-how, trade secrets, ideas, concepts, inventions, discoveries, data, formulae, specifications, information relating to Materials (including biological and chemical structures and functions as well as methods for synthesising chemical compounds), procedures for experiments and tests, results of experimentation and testing (including pharmacological, biological, chemical, biochemical, clinical test data and data resulting from non-clinical studies), results of research and development including laboratory records and data analyses. Information in a compilation or a compilation of information may be Know How notwithstanding some or all of its individual elements are in the public domain.

"*Lead Series Criteria*" means in relation to each project conducted pursuant to the ICR Internal Research, the project-specific criteria, which shall be determined by the JSC in accordance with Clause 2.1(n).

"License Agreement" has the meaning given in the recitals.

"Losses" has the meaning given in Clause 14.1.

"Major Markets" means the United Kingdom, the United States, France, Italy, Spain, Germany and Japan and "Major Market" shall mean any one of them.

"Materials" means any chemical or biological substances including any: organic or inorganic element or compound; nucleotide or nucleotide sequence including DNA and RNA sequences; gene; vector or construct including plasmids, phages, bacterial vectors, bacteriophages and viruses; host organism including bacteria, fungi, algae, protozoa and hybridomas; eukaryotic or prokaryotic cell line or expression system or any development strain or product of that cell line or expression systems; protein including any peptide or amino acid sequence, enzyme, antibody or protein conferring targeting properties and any fragment of a protein or a peptide enzyme or antibody; drug or pro-drug; assay or reagent; any other genetic or biological material or micro-organism or any transgenic animal; and any physical property rights relating to any of the foregoing.

"Member" shall have the meaning given in Clause 2.2(a).

"Milestone Events" has the meaning given in Clause 9.1.

"Milestone Payments" has the meaning given in Clause 9.1.

"Net Sales" means the gross amount invoiced on account of sales of a Product by the Company or any of its Affiliates or Sub-Licensees in th Territory (but not including sales between the Company, its Affiliates or Sub-Licensees where the Product is intended for resale) less the following deductions directly relating to such sales of Product:
(a) [***];
(b) [***];

(b) [***]; (c) [***]; (d) [***]; (e) [***]; (f) [***]; and (g) [***].

For purposes of this definition, the Product shall be considered "sold" and "deductions" allowed when recorded as invoiced in the Company's, its Affiliate's or Sub-Licensee's financial statements prepared in accordance with the relevant accounting standards. Net Sales shall exclude [***].

"NHS" means the National Health Service in England and Wales (or any successor organisation thereto).

"Non-Degradation Program" means a drug discovery program with respect to one or more ICR Internal Targets of the corresponding ICR Internal Research Program Hits aimed at identifying, discovering and developing a Non-Protein Degradation Product.

"Non-Protein Degradation Product" means any product that 1s not a Protein Degradation Product.

"Notice of Interest" has the meaning given in Clause 5.2(c)(ii).

"Oncology Indication" means [***].

"Option A Program" has the meaning given in Clause 5.2(a).

"Parties" means CRT, the ICR and the Company and "Party" shall mean any of them.

"*Patents*" means any patent applications, patents, author certificates, inventor certificates, utility models, together with all divisionals, renewals, continuations, continuations-in-part, extensions, re-examinations, reissues, substitutions, confirmations, registrations, revalidations and additions of or to them, as well as any Supplementary Protection Certificate, or any like form of protection anywhere in the world for each of the foregoing.

"Payment Schedule" means the payment schedule set out in Schedule 1.

"Phase I Trial" means a human clinical trial in which a Product is administered to human subjects at multiple dose levels with the primary purpose of determining safety, metabolism, pharmacokinetic and pharmacodynamic data of the Product and consistent with 21 CFR § 312.21(a).

"*Phase II Trial*" means a clinical trial of a Product in human patients, the principal purpose of which are to make a preliminary determination that the Product is safe for its intended use, to determine its optimal dose, and to obtain sufficient information about the Product's efficacy to permit the design of Phase III Trials, and consistent with 21 CFR 312.21(b).

"Phase III Trial" means a human clinical trial of a Product, which trial is designed: (a) to establish that the Product is safe and efficacious for its intended use; (b) to define warnings, precautions and adverse reactions that are associated with the Product in the dosage range to be prescribed; and (c) consistent with 21 CFR § 312.21(c). Any Phase II Trial that is adapted to be a larger scale trial and intended as a pivotal trial for the purpose of obtaining Regulatory Authorisation of a Product, shall be deemed a Phase ill Trial.

"*Pre-Clinical Candidate*" means any compound that is nominated by the !CR as a pre-clinical candidate, considering the Pre-Clinical Candidate Criteria, and is approved by the JSC as a "Pre-Clinical Candidate".

"*Pre-Clinical Candidate Criteria*" means in relation to each project conducted pursuant to the ICR Internal Research, the project-specific criteria which shall be determined by the JSC (in line with the guideline criteria set forth in Schedule 3), as may be amended from time to time by the JSC.

"*Price Approval*" means, in those countries in the Territory where a Competent Authority may approve or determine pricing and/or pricing reimbursement for pharmaceutical products, such approval or determination.

"*Product*" means any product: (i) the manufacture, use or sale of which is covered by one or more Valid Claims of any Programme Patents; (ii) containing or comprising a Programme Compound and/or an Unencumbered CRT/ICR Background Hit Compound; and/or (iii) discovered, developed and/or generated using or incorporating any part of the Programme Intellectual Property or (to the extent licensed pursuant to Clause 7.6(a)) the CRT/ICR Background Library Intellectual Property. For the purposes of Clause 8, 19.2(t), and 20.2 only, "Product" shall also mean: (a) any product containing or comprising a CRT/ICR Existing Compound; and/or (b) discovered, developed and/or generated using or incorporating any part of the CRT/ICR Existing Intellectual Property.

"Programme" means the programme of research set out in Schedule 2 as may be amended from time to time in accordance with this Agreement.

"Programme Compound" means (a) any compound that is discovered, developed and/or generated by or on behalf of the Company and/or the ICR pursuant to the conduct of the Programme and/or (b) any salt, free acid, free base, clathrate, solvate, hydrate, hemihydrates, anhydride, ester, chelate, conformer, congener, crystal form, crystal habit, polymorph, amorphous solid, homolog, isomer, stereoisomer, enantiomer, racemate, prodrug, isotopic or radiolabeled equivalent, metabolite, conjugate, complex, mixture, serum, solution, lyophilized material, or other formulation of any such compound; but for the avoidance of doubt excludes CRT/ICR Background Hit Compounds other than Duplicate Compounds which remain in the Compound Library as referred to in part (b) of the definition of Compound Library.

- "Programme Intellectual Property" means Programme Patents and Programme Know How and Programme Materials but for the avoidance of doubt excludes (a) CRT/ICR Background Hit Compounds, other than Duplicate Compounds which remain in the Compound Library as referred to in part (b) of the definition of Compound Library and (b) that part of the Know How in the CRT/ICR Background Library Intellectual Property that is the identity of any compound (e.g. as may be identified by chemical structure) within the CRT/ICR Background Library.
- "Programme Know How" means any and all Know How developed and/or generated by or on behalf of the Company and/or the ICR pursuant to conduct of the Programme, including the chemical structures of the Programme Compounds and the data generated by the ICR from screening of any CRT/ICR Background Library using the [***], but for the avoidance of doubt excludes that part of the Know How in the CRT/ICR Background Library Intellectual Property that is the identity of any compound (e.g. as may be identified by chemical structure) within the CRT/ICR Background Library.
- "*Programme Materials*" means any and all Materials developed and/or generated by or on behalf of the Company and/or the ICR pursuant to conduct of the Programme, including the physical stocks of Programme Compounds.
- "Programme Patents" means any and all Patents claiming patentable inventions made or conceived by or on behalf of the Company and/or the ICR pursuant to the conduct of the Programme.
- "*Progress Report*" means a detailed written report produced by the ICR in respect of: (i) the progress of a Non-Degradation Program; (ii) the progress of any applications for Regulatory Authorisation and (where relevant) Price Approvals for products developed under such Non-Degradation Program; and (iii) the progress of and plans for marketing and sale of products developed under such Non-Degradation Program.
 - "Project Compound Intellectual Property" has he meaning given in Clause 5.2(a)(ii).
 - "Project Non-Compound Intellectual Property" has the meaning given in Clause 5.2(a)(ii).
- "Protein Degradation Product" means any Product designed or intended to have a primary mechanism of action through CRBN-mediated protein degradation.
- "Quarter" means any of the three-monthly periods commencing on the first day of any of the months of January, April, July, and October in any year and "Quarterly" has a corresponding meaning.
 - "Receiving Party" has the meaning given in Clause 17.1.
- "*Regulatory Authorisations*" means all authorisations, approvals, clearances, and licences of a Competent Authority that may be required in any country of the Territory prior to commercial sale of the relevant Product in the Field, including any necessary variations thereto, but excluding any Price Approvals.

- "Researchers" has the meaning given in Clause 3.2.
- "Reviewers" has the meaning given in Clause 17.1
- "Screening Platform" means [***].
- "Selected Programme Patents" has the meaning given in Clause 12.3.
- "Services Expiry Date" means the date that is [***] after the Effective Date; provided that if the [***] Researcher has not been appointed by such date, then such date shall be extended to the earlier to occur of (a) the date that the [***] Researcher is appointed or (b) the date that is [***] after the Effective Date.
 - "Services Term" has the meaning given in Clause 3.3.
 - "Shareholders' Agreement" has the meaning given in the recitals.
 - "Sponsor" shall have the meaning set out in Directive 2001/20/EC, as the same may be amended.
- "Sub-Licensee" means any Third Party to whom the Company grants a license or sublicense of rights under any Company Compound Intellectual Property or Company Non-Compound Intellectual Property, excluding licenses granted to any Third Party Service Provider in connection with the provision of research, development and/or manufacturing services.
 - "Substrates" means, [***].
- "Supplementary Protection Certificate" means a right based on a patent pursuant to which the holder of the right is entitled to exclude third parties from using, making, having made, selling or otherwise disposing or offering to dispose of, importing or keeping the product to which the right relates, such as supplementary protection certificates in Europe, and any similar right anywhere in the world.
 - "Term" means the term of this Agreement determined in accordance with Clause 18.1.
 - "Territory" means worldwide.
 - "Third Party" means a person other than a Party or an Affiliate of a Party.
- "Third Party Service Provider" means a Third Party who provides research, development and/or manufacturing services to the ICR, Company or its Sub-Licenses in connection with the Products, including contract research, manufacturing or sales organisations, universities and hospitals. However, a Tobacco Party may not act as a Third Party Service Provider.
- "Tobacco Party" means: (i) any person who develops, sells or manufactures tobacco products; and/ or (ii) any person which makes the majority of its profits from the importation, marketing, sale or disposal of tobacco products. Furthermore, Tobacco Party shall include any person that is Controlled by or under common Control with any of the persons referred to in (i) and/or (ii).

- "Transaction Documents" means the Formation and Investment Agreement, Shareholders' Agreement and License Agreement.
- "UK Pricing Authority" means any supra-national, national or regional government department, authority, agency or entity (including a non-departmental public body or similar entity) with responsibility for evaluating the cost effectiveness of medicinal products in the United Kingdom (or one or more constituent countries thereof) or otherwise determining whether the NHS (or constituent parts thereof) should purchase medicinal products.
 - "Unencumbered CRT/ICR Background Hit Compound" shall have the meaning set forth in Clause 7.6(a).
- "Valid Claim" means a claim of any Programme Patent which has not expired, been withdrawn, abandoned or surrendered or been refused, revoked or held invalid in a decision rendered by a court or other governmental agency of competent jurisdiction in the relevant country or territory from which no appeal has been or can be taken, including through opposition, reexamination, reissue or disclaimer.
 - "VAT" means value added tax.
 - "Versant Ventures" means Versant Ventures Capital VI L.P.
 - "Year" means a calendar year.
 - **1.2** In this Agreement:
- (a) unless the context otherwise requires, all references to a particular Clause, paragraph or Schedule shall be a reference to that clause, paragraph or schedule, in or to this Agreement as it may be amended from time to time pursuant to this Agreement;
 - (b) the table of contents and headings are inserted for convenience only and shall be ignored in construing this Agreement;
- (c) unless the contrary intention appears, words importing the masculine gender shall include the feminine and vice versa and words in the singular include the plural and vice versa;
- (d) unless the contrary intention appears, words denoting persons shall include any individual, partnership, company, corporation, joint venture, trust association, organisation or other entity, in each case whether or not having separate legal personality;
- (e) reference to the words "include(s)" or "including" are to be construed without limitation to the generality of the preceding words, and, for clarity, the words "include(s)" or "including" shall be deemed to be followed by the phrase "without limitation" or like expression; and

(f) reference to any statute or regulation includes any modification or re-enactment of that statute or regulation, provided that the modification or re-enactment does not diminish the rights or extend the obligations of any Party.

2. JSC

- **2.1** As soon as practicable following the Effective Date, the Parties shall establish a joint steering committee ("*JSC*") to provide a forum for the coordination, communication and oversight of the Parties' activities under this Agreement. Except as otherwise provided herein, the JSC shall have the authority to:
 - (a) the extent that the same is not set out in Schedule 2, oversee the conduct of the Programme by the ICR;
 - **(b)** monitor progress against any agreed milestones and the timetable of the Programme;
 - (c) the extent that the same is not set out in Schedule 2, allocate the work under the Programme;
 - **(d)** promote the due performance of the Programme;
- (e) advise and assist in the resolution of any scientific or technical difficulties which are experienced in the performance of the Programme, including assessment of the Programme for purposes of Section 19.2(g);
 - (f) review the results of the Programme with a view to identifying any patentable inventions;
 - (g) consider opportunities for publications and patent filing;
 - **(h)** discuss the patent strategy for Programme Patents;
 - (i) serve as the initial forum to discuss and seek to resolve any issue or dispute arising between the Parties under the Agreement;
- (j) review periodic reports, prepared by the ICR in accordance with the schedule agreed between the Parties (in the first instance, Quarterly), summarising in reasonable detail the results of the Programme during such period, and send copies of such reports to the Parties;
- (k) subject to Clause 3.7, review, discuss and propose amendments to Schedule 2 from time to time as may be necessary or desirable to give effect to this Agreement;
- (I) receive updates from the Company from time to time regarding the research activities undertaken by Ridgeline, or any other Third Party, on behalf of the Company and contributing to, related to, adding to the delivery of and/or in support of the Programme, or associated with the subject-matter of the Programme, except to the extent CRT and/or the ICR notifies the Company that it has a conflict with respect to any particular Ridgeline or other Third Party activity.

- (m) review and approve the ICR's proposed screens of any Compound Library in connection with the ICR Internal Research, proposed progression routes, and evaluate the output of such screening and subsequent hit validation and deconvolution studies;
- (n) review and approve the Lead Series Criteria for each project conducted pursuant to the ICR Internal Research, in line with the guideline criteria set forth in Schedule 3, as adapted for such project;
 - (o) review and approve the Pre-Clinical Candidate Criteria for each project conducted pursuant to the ICR Internal Research;
- (p) review and approve each Pre-Clinical Candidate nominated by the ICR, taking into account the Pre-Clinical Candidate Criteria applicable to such nominated candidate;
- (q) receive updates from the Company from time to time regarding any research activities undertaken by or on behalf of Company in support of the Programme;
- (r) receive and discuss summary reports from the Company regarding the research, development and commercialization of all Products at least once every [***], including a summary in reasonable detail of the progress of development of each Product against the respective development plan and the progress of any applications for Regulatory Authorisation; and
 - (s) perform any obligations that are expressly delegated to it under this Agreement;
 - Parts (a), (b), (c), (d), (e), (f), (j) (k) and (I) shall apply for the duration of the Services Term only.
 - **2.2** The JSC shall be established and run by the Parties as follows:
- (a) The JSC shall comprise of [***] appointed by of the Company, [***] appointed by the ICR, and [***] appointed by CRT (collectively, the "*Members*"). The initial members of the JSC shall be as follows:

COMPANY MEMBER	CRT/ICR MEMBER
[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

- **(b)** The Parties, through the JSC, may change the number of Members. Each Party shall be entitled to remove any Member appointed by it and to appoint any person to fill a vacancy arising from the removal or retirement of such Member. The Parties shall give each other, prior written notice of any proposed changes in the identity of their Members.
- **(c)** The Parties shall use reasonable endeavours to ensure that their appointed Members are of a level of expertise and seniority to make decisions arising within the scope of the JSC's responsibilities.
- (d) The JSC shall meet as soon as practicable after the Effective Date and thereafter shall hold regular meetings for the duration of the Service Term [***] and thereafter [***], unless otherwise agreed to by the Parties, and at any time upon the reasonable request of a Party, in each case at dates and times to be mutually agreed:
- **(e)** The JSC shall meet in person, by teleconference or by video-teleconference, as agreed by the JSC. During the Services Term, in-person meetings shall be at the ICR unless otherwise agreed by the Parties. Each Party shall solely bear all travel and subsistence costs incurred by such Party's Members in connection with their attending each the meeting of the JSC.
 - (f) At least [***] written notice of each proposed meeting of the JSC shall be given to each Member, by the Company.
- **(g)** The quorum for meetings of JSC shall be at least one (1) Member appointed by of each of the Company, CRT and the ICR. Members may be represented at any meeting by another Member designated in writing by the absent Member. Each Party may invite a reasonable number of nonmember, non-voting representatives of such Party, who are subject to written confidentiality obligations commensurate in scope to the provisions of Clause 17, to attend meetings of the JSC.
- **(h)** CRT and the ICR shall initially appoint from the Members a chairman to chair meetings, who shall be [***]. In the event [***] resigns as the chairman or is no longer affiliated with CRT and the ICR, then the ICR shall appoint an ICR employee as replacement chairman, and shall be entitled to remove any chairman appointed by it and to appoint any person to fill a vacancy arising from the removal or retirement of such chairman, in each case, subject to the consent of the Company, not to be unreasonably withheld. The chairman shall have no casting vote, except as expressly provided in Clause 2.2(i).
- (i) The JSC shall have only the powers expressly assigned to it in Clause 2.1, and shall have no power to amend, modify, or waive compliance with this Agreement. Decisions of the JSC shall be made by unanimous agreement of the Members present, with the Company's appointed Members collectively having one (1) vote, and CRT's- and ICR's appointed Members collectively having one (1) vote. No vote of the JSC may be taken unless at least one (1) of each Party's representatives is present for the vote. If the JSC cannot reach consensus with regard to any matter within its decision-making authority within [***] after such matter has been brought to the JSC's attention, then [***].

(j) The chairman, or the chairman's delegate, shall be responsible for: (a) preparing JSC meeting agendas reasonably in advance of JSC meetings, which JSC meeting agendas will include all agenda items reasonably requested by any Member for inclusion therein; (b) sending invitations and a JSC meeting agenda along with appropriate information for such agenda to all members of the JSC at least [***] before the next scheduled meeting of the JSC; (c) preparing and circulating minutes within [***] after each meeting of the JSC setting forth, among other things, a description, in reasonable detail, of the discussions at the meeting and a list of any actions, decisions, or determinations approved by the JSC. Such minutes shall be effective only after being approved by all Members. Definitive minutes of all JSC meetings shall be finalized no later than [***] after the meeting to which the minutes pertain. The minutes of each meeting of the JSC shall be retained in a suitable minute book, which shall be kept by the Company and be available at all reasonable times on Business Days for inspection by the Parties and the Members.

(k) The JSC will be disbanded upon expiration or termination of this Agreement.

3. PROGRAMME

3.1 Subject to the terms and conditions of this Agreement, and as further provided in this Clause 3, the ICR shall provide research services to the Company in accordance with the Programme, which shall describe the research activities to be conducted by the Parties, responsibilities of each Party, the number of FTEs to be provided by the ICR to conduct such activities, and will contain a good faith estimate of all costs and expenses necessary to perform the Programme, including a budget for all costs to be funded by Company and, solely in the case of any outsourcing costs included in such budget, incurred by the ICR in conducting such activities. For the avoidance of doubt, the Company shall not be responsible for any other costs beyond those set forth in Clause 4.1. As among the Parties, the ICR shall be responsible for all costs it incurs in conducting the Programme (as set forth in Schedule 2 as of the Effective Date) that are not intended to be funded by the Company pursuant to Clause 4.1, provided that, for the avoidance of doubt, in no event shall the ICR be obliged to commit any additional cost or resource for the duration of the Programme beyond the allocation of FTEs from the CTU, as set forth in Schedule 2.

3.2 In consideration for the payments to be made pursuant to Clause 4.1, the ICR shall engage the researchers set forth under the section entitled "Researchers" (the "*Researchers*") in Schedule 2 for the period of time set forth therein, to be appointed as soon as practicable after the Effective Date, to work at the ICR as employees of the ICR, to carry out under the supervision of the [***], or the chairman of the JSC in the event [***] is no longer affiliated with CRT and the ICR, those parts of the Programme allocated to the ICR hereunder. The Company shall provide the ICR with such assistance with the selection and recruitment of the Researchers as the ICR may reasonably request, and the ICR shall promptly notify the Company of the appointment of each Researcher, including the effective date of each such appointment.

- 3.3 The Programme shall commence on the Effective Date and continue until the Services Expiry Date unless earlier terminated as provided in this Clause 3 .3 or upon termination of this Agreement (the "Services Term"). Notwithstanding the foregoing, as soon as reasonably practicable following the [***] of the Effective Date, the JSC, subject to approval by the Board of Directors of the Company, shall determine whether the Programme shall proceed or shall terminate, effective as of the date of such approval by the Board of Directors of the Company. In the event of the termination of the Programme in accordance with the immediately preceding sentence, the Company shall pay any outstanding amounts to the ICR that would have been payable by the Company, if any, with respect to each Researcher who was appointed prior to the date of notice of such termination until expiry of the [***] of the date each such Researcher's contract at the ICR began or, if an individual Researcher is already an employee of the ICR the date such Researcher starts conducting work on the Programme, together with any other costs that the ICR has incurred or committed to which would have been reimbursable by the Company to the ICR but for such decision to terminate, and which cannot be cancelled by the ICR under applicable law. For clarity, if the [***] period applicable to a particular Researcher expired before such termination, no amounts shall be payable by the Company to the ICR with respect to such Researcher as a result of the preceding sentence. During such remaining period of time, the Researchers shall wind down the Programme and conduct such activities as agreed to by the JSC and/or the Parties.
- **3.4** During the Services Term, each of the ICR and the Company shall use Commercially Reasonable Efforts to conduct all activities allocated to it under the Programme.
- **3.5** Each of the Company and the ICR shall keep the JSC fully informed of the identity and qualifications of those of its employees engaged in carrying out the Programme.
- **3.6** Each of the ICR and the Company shall keep or cause to be kept complete, current, accurate, organised and legible records of the results achieved under the Programme, and all data and other Know How resulting from such activities. Such records shall fully and properly reflect all work done and results achieved in the performance of the Programme appropriate for regulatory and patent purposes. Each Party shall have the right to review and copy such records maintained by the other Parties at reasonable times during normal business hours, and in any event no more than [***], and to obtain access to the originals to the extent necessary or useful for regulatory and patent purposes, to the extent such Party has the right to conduct regulatory and patent activities under this Agreement.
- **3.7** Unless otherwise agreed between the Parties, no Party shall be under an obligation to provide additional resource or facilities over and above those set out in this Agreement. The Programme may only be amended by the prior written agreement of the Parties.

4. RESEARCH FUNDING

- **4.1** As a contribution to the ICR's costs of carrying out its allotted tasks under the Programme, the Company shall make payments to the ICR in accordance with the Payment Schedule. All payments shall be paid [***] in advance with the first such amount being due on the Effective Date. Such payments shall be made directly to the ICR and sent to [***]; Account name: [***]; Account number: [***]; Sort code: [***].
 - **4.2** The ICR shall apply the payments received from the Company pursuant to Clause 4.1 solely for the purpose of carrying out the Programme.

5. ICR INTERNAL RESEARCH

5.1 Subject to the terms and conditions of this Agreement, and as further provided in this Clause 5, following the [***] of the Effective Date, the ICR may propose to the JSC that the ICR conduct [***] or more screens of any Compound Library; *provided* that any such screens shall not overlap and shall be materially different from screens undertaken or planned to be undertaken by the Company, as determined by the JSC.

(a) By way of example:



(b) After receipt of JSC approval, if any, the ICR may conduct such screen in accordance with the parameters and a written work plan approved by the JSC (the "*ICR Internal Research*"). In the event the Parties dispute any matters to be determined or approved by the JSC pursuant to this Clause 5.1, the Parties shall first seek to resolve such dispute in accordance with Clause 32.2. In the event such dispute is not resolved in accordance with Clause 32.2, the Parties shall refer such dispute to an Expert for resolution in accordance with Clause 32.1.

5.2 The ICR shall promptly notify the JSC of any ICR Internal Research Program Hits and ICR Research Target IP, and the Company and the ICR, acting through the JSC and subject to approval by the Board of Directors of the Company, shall jointly assess any ICR Internal Research Program Hits and ICR Research Target IP to determine the progression route, which shall be selected from the following two (2) options.

(a) Option A.

(i) The Company may elect to progress a drug discovery program with respect to such ICR Interna Research Program Hits and any corresponding ICR Internal Target in which the Company would conduct research and development activities to develop and select a Pre-Clinical Candidate (and if mutually agreed, undertake phase I tria1(s)) exclusively in collaboration with the ICR, with the allocation of such research and development activities between the ICR, the Company and appropriate third parties as mutually agreed, and the Company would have exclusive commercialization rights, in each case in accordance with the terms of clause 5.2(a)(ii) below (each, an "Option A Program"). The research and development costs of such Option A Program shall be funded by or on behalf of the Company, or shared with the ICR, as mutually agreed to by the Company and the ICR in writing.

(ii) The Company, CRT and the ICR shall negotiate in good faith and enter into a definitive development and license agreement with respect to such Option A Program, the terms and conditions of which shall provide for the following, unless otherwise agreed in writing by the Company and the ICR: (1) all intellectual property rights developed in connection with research and development activities conducted in collaboration by CRT, the ICR and the Company under such Option A Program shall be jointly owned by the Company, CRT and the ICR, excluding the ICR Internal Research Program Hits, which are subject to Clause 6.1; (2) CRT and the ICR shall grant to the Company (a) a worldwide, exclusive, sub-licensable license under their rights in the applicable Project Compound Intellectual Property, and (b) a worldwide, sub-licensable license under their rights in the applicable Project Non-Compound Intellectual Property and ICR Research Target IP, in each case for all purposes in the Field, which license may be exclusive or non-exclusive as to specific Project Non-Compound Intellectual Property and ICR Research Target IP as agreed, on commercially reasonable and customary terms, taking into account [***]; (3) the Company shall have the first right to manage the prosecution, filing and maintenance of such Project Compound Intellectual Property and Project Non-Compound Intellectual Property (subject to CRT and the ICR having step-in rights in the event that the Company does not file, or continue to prosecute and maintain any such intellectual property rights) and the commercialization strategy with respect thereto; (4) as between the Parties, all intellectual property rights developed by or on behalf of the Company outside of the research and development activities conducted in collaboration by CRT, the ICR and the Company under such Option A Program shall be solely owned and managed by the Company; (5) the Company shall grant to CRT and the ICR a worldwide, non-exclusive, sub-licensable licence under its rights (including rights granted under (2) above) in such Project Non-Compound Intellectual Property and ICR Research Target IP for all purposes, subject to [***], in each case with details taking the Parties' interests into account to be negotiated in the definitive agreement for the Option A Program; and (6) during the term of the research and development collaboration period between the ICR and the Company and for [***] thereafter, CTU shall not conduct another drug discovery project against the relevant ICR Internal Target (save that for the avoidance of doubt nothing shall prohibit the CTU from continuing any drug discovery project against the relevant ICR Internal Target where such project was initiated prior to the identification of the ICR Internal Target pursuant to the ICR Internal Research). "Project Compound Intellectual Property" means, with respect a particular Option A Program, that part of the intellectual property referred to in (1) above that consists of, contains or relates directly and solely to any such Option A Program compound (such Option A Program compounds to be defined in the definitive agreement), but for the avoidance of doubt excludes the ICR Internal Research Program Hits, and the intellectual property referred to in paragraph (b) of the definition of Project Non-Compound Intellectual Property. "Project Non-Compound Intellectual Property" means, with respect a particular Option A Program, (a) that part of the intellectual property referred to in (1) above which is not Project Compound Intellectual Property, and (b) includes without limitation that part of the intellectual property referred to in (1) above that consists of, contains or relates to targets, biomarkers, assays, protocols and synthetic routes, insights regarding biochemical interactions, formulations and processes for manufacture or use which have applicability outside the chemical matter comprising Project Compound Intellectual Property, but excludes the ICR Internal Research Program Hits.

(b) Option B. If the Company elects not to progress a drug discovery program as an Option A Program as provided in Clause 5.2(a), then
the ICR may elect to progress a drug discovery program with respect to any ICR Internal Target corresponding to such ICR Internal Research Program
Hits in accordance with Clause 5.2(c), if:

- (i) [***];
- (ii) [***]; and
- (iii) the aim is to develop a Non-Protein Degradation Product.

In the event the Parties dispute whether a progressable target has been identified under the Programme, solely for the purposes of this Clause 5.2(b), the Parties shall first seek to resolve such dispute in accordance with Clause 32.2. In the event such dispute is not resolved in accordance with Clause 32.2, the Parties shall refer such dispute to an Expert for resolution in accordance with Clause 32.1.

- (c) Subject to the terms and conditions of this Clause 5.2(c), in the event the conditions of Clause 5.2(b) are met, the ICR may elect to progress a Non-Degradation Program with respect to such ICR Internal Target, at its sole cost (as between the Parties). For clarity, in no event will any Non-Degradation Program include the aim of developing any Protein Degradation Product. All intellectual property rights developed in connection with such Non-Degradation Program shall by jointly owned by CRT and the ICR in equal and undivided shares, without regards to the inventorship thereof, excluding the ICR Internal Research Program Hits, which are subject to Clause 6.1. Each Party hereby assigns and agrees to assign such of its interest in and to such intellectual property rights as is necessary to effect such vesting.
- (i) The ICR shall provide the JSC and the Company with a Progress Report with respect to each Non-Degradation Program at least once every [***]. Such Progress Report shall summarise in reasonable detail the results of the Non-Degradation Program during such period, unless agreed otherwise by the Parties.
- (ii) As soon as reasonably practicable after the JSC determines that a Non-Degradation Program satisfies the Lead Series Criteria, the ICR shall provide written notice of the same to the Company, and within [***] of such notice deliver to the Company a Data Package with respect to such Non-Degradation Program. The Company shall have the right to request, whilst at all times acting reasonably, any additional information required to make such Data Package complete within [***] of such notice, and the ICR shall provide such requested information to the Company to the extent it is available within [***] of such request. Within [***] after receipt of each such complete Data Package, the Company will have the right to deliver written notice to the ICR and CRT of its interest in progressing such Non-Degradation Program by conducting research and development activities to develop and select a Pre-Clinical Candidate (and if mutually agreed, undertake phase I trial(s)) exclusively in collaboration with the ICR, with the allocation of such research and development activities between the ICR, the Company and appropriate third parties as mutually agreed, and the Company would have exclusive commercialization rights, in each case in accordance with the terms of this Clause 5.2(c)(ii) (each, a "Notice of Interest") and, if the Company provides the Notice of Interest, the Parties shall promptly enter into good faith negotiations with respect to a definitive development and license agreement (including the grant by CRT and the ICR to the Company of rights to intellectual property rights developed in connection with such Non-Degradation Program") in accordance with the terms outlined in Clause 5.2(a)(ii), provided that (a) references to "Option A Program" shall be replaced by "Non-Degradation Program", and (b) it is understood that the part of the intellectual property referred to in part (1) of such Clause 5.2(a)(ii) that is developed prior to the effective date of such definitive development and license agreement shall be join

(iii) If the Company does not provide a Notice of Interest with respect to a Non-Degradation Program within the applicable [***] period, then the ICR shall promptly provide written notice of the same, including the complete Data Package for such Non-Degradation Program, to Versant Ventures. Within [***] after delivery of such notice, Versant Ventures will have the right to deliver written notice to the ICR and CRT of its interest in such Non-Degradation Program and, if Versant Ventures delivers such notice, CRT, the ICR and Versant Ventures shall, as soon as reasonably practicable after delivery of such notice, enter into good faith arms-length negotiations with respect to a definitive development and license agreement among CRT, the ICR and a new company to be funded by Versant Ventures (including the grant by CRT and the IC to the such new company of rights to intellectual property rights developed in connection with such Non-Degradation Program) in accordance with the terms outlined in Clause 5.2(a)(ii), provided that (a) references to "Option A Program" shall be replaced by "Non-Degradation Program", (b) references to "Company" shall mean such new company, and (c) it is understood that the part of the intellectual property referred to in part (1) of such Clause 5.2(a)(ii) that is developed prior to the effective date of such definitive development and license agreement shall be jointly owned by CRT and the ICR only.

(iv) If the Company does not provide a Notice of Interest with respect to a Non-Degradation Program in accordance with Clause 5.2(c)(ii) and Versant Ventures does not provide a written notice of its interest to the ICR with respect to such Non-Degradation Program in accordance with Clause 5.2(c)(iii), then the ICR shall have the right to continue pursuing such Non-Degradation Program at its sole cost (as between the Parties) and, subject to Clause 5.2(c)(v), the ICR and CRT shall be free to further discover, develop and commercialise such Non-Degradation Program (including by granting licences to any Third Parties) without any further obligation to Company. CRT and the ICR shall share with Company a percentage (%) of net revenues they receive from their commercialisation of such Non-Degradation Program, such percentage to be agreed between the Parties at or prior to the time of commercialisation to reflect the ICR's initial use of the Compound Library.

(v) In the event that CRT and/or the ICR has not entered into a commercial collaboration, partnership or licence with a Third Party in relation to a Non-Degradation Program that was not licensed to either the Company or Versant Ventures in accordance with Clause 5.2(c)(ii) or 5.2(c) (iii), as soon as reasonably practicable after the JSC determines that a pre-clinical development candidate identified under such Non-Degradation Program satisfies the Pre-Clinical Candidate Criteria; the ICR shall provide written notice of the same to the Company, and within [***] of such notice deliver to the Company a Data Package with respect to such Pre-Clinical Candidate. The Company shall have the right to request, whilst at all times acting reasonably, any additional information required to make such Data Package complete within [***] of such notice, the ICR shall provide such requested information to the Company to the extent it is available within [***] of such request. Within [***] after receipt of each such complete Data Package, the Company will have the right to deliver written notice to the ICR and CRT of its interest in such Pre-Clinical Candidate and, if the Company provides such notice, the Parties shall promptly enter into good faith negotiations with respect to the grant of rights by CRT and the ICR to the Company does not provide such a notice with respect to a Non-Degradation Program in accordance with this Clause: (A) CRT and the ICR shall be entitled to further discover, develop and commercialise (including granting licences to any Third Parties) without any further obligation to Company; and (B) CRT and the ICR shall share with Company a percentage (%) of net revenues they receive from their commercialisation of such Non-Degradation Program, such percentage to be agreed between the Parties at or prior to the time of commercialisation to reflect the ICR's initial use of the Compound Library.

5.3 The Company shall have no right to progress arty drug discovery program with respect to any target identified through the ICR Internal Research for the purposes of developing a product without the prior written approval of CRT and the ICR, except that the foregoing shall not limit or restrict the Company, directly or indirectly, from progressing any drug discovery program with respect to any target (i) that is identified as part of the Programme, (ii) that is part of any drug discovery program for which the Company has exercised its right to obtain rights to such program pursuant to Sections 5.2(a) or 5.2(c) and as permitted under the terms subsequently agreed in respect of such drug discovery program in accordance with those Clauses, or (iii) that is independently identified by or on behalf of the Company without the use of the CRT/ICR Existing Intellectual Property. In the event CRT and the ICR provide written approval for any drug discovery program that requires such approval under the preceding sentence independent of Clause 5.2(a), the Parties sha negotiate in good faith a definitive development agreement with respect to such drug discovery program on commercially reasonable and customary terms, which, for clarity, may include [***].

5.4 Subject to the rights and obligations set forth in this Clause 5 with respect to any target identified through the ICR Internal Research, nothing shall limit or restrict the ICR, CRT or CRUK, directly or indirectly, from progressing a drug discovery program outside of this Agreement with respect to any target (i) where such drug discovery program has been initiated prior to identification of such target under the Programme; or (ii) that is independently identified by or on behalf of the ICR, CRT, CRUK or a collaborative partner of any of them without the use of the Programme Intellectual Property or in breach of this Agreement. Nothing shall limit or restrict the Company, directly or indirectly, from progressing a drug discovery program outside of this Agreement with respect to any target that is independently identified by or on behalf of the Company or a collaborative partner of any of them without the use of the CRT/ICR Existing Intellectual Property, Programme Intellectual Property or, as applicable, Project Compound Intellectual Property or Project Non-Compound Intellectual Property, or in breach of this Agreement.

6. OWNERSHIP OF INTELLECTUAL PROPERTY

6.1 Subject to any licenses granted to CRT and/or the ICR hereunder, as among the Parties, all rights in and to the Programme Intellectual Property and ICR Internal Research Program Hits shall vest solely in the Company, without regards to the inventorship thereof. Each of CRT and the ICR hereby assigns to the Company (or if assignment is not permitted, waives enforcement of or grants to the Company an exclusive, irrevocable, perpetual, worldwide, fully paid license, with the right to sublicense through multiple tiers of sublicense under) all right, title and interest CRT and the ICR, as applicable, may have in and to Programme Intellectual Property and ICR Internal Research Program Hits arid shall cooperate fully, including by providing any reasonably necessary and suitably limited powers of attorney and assignments and executing all other papers and instruments, and requiring its employees and subcontractors, to execute such reasonably necessary and suitably limited powers of attorney and assignments and all other papers and instruments, to effectuate the ownership of the Company and to enable the Company to file, prosecute, defend and enforce any Patents in such Programme Intellectual Property in any country.

6.2 All rights in and to the ICR Research Target IP shall vest jointly in CRT and the ICR and the Company in equal and undivided shares, without regards to the inventorship thereof. Each Party hereby assigns and agrees to assign such of its interest in and to the ICR Research Target IP (if any) as is necessary to effect such vesting. For the avoidance of doubt, irrespective of the terms of Clause 5, the Company and CRT hereby grant the ICR a non-exclusive licence to the ICR Research Target IP for the purpose of undertaking any academic non-commercial research.

6.3 Subject to any licences granted to the Company hereunder, as among the Parties, the CRT and the ICR shall retain all rights in, title to and ownership of CRT/ICR Background Library Intellectual Property.

7. LICENCES

7.1 The Company shall be entitled to grant sub-licences in respect of the rights granted under this Agreement, provided that no sub-licence shall be granted to a Tobacco Party. The grant of any sub-licence shall be without prejudice to the Company's obligations under this Agreement. Any act or omission of any such Sub-Licensee which, if it were the act or omission of the Company would be a breach of any of the provisions of this Agreement, will be deemed to be a breach of this Agreement by the Company who will be liable to CRT and the ICR accordingly, unless the Company terminates such sub-licence or cures any such breach.

7.2 Subject to the terms and conditions of this Agreement, for the duration of the Services Term, the Company hereby grants to the ICR under the Programme Intellectual Property and its rights in the CRT/ICR Existing Intellectual Property (it being agreed that such grant with respect to the CRT/ICR Existing Intellectual Property is not subject to the obligations relating to sub-licenses under Clause 2.2 of the License Agreement), a non-exclusive, fully paid- up, non-transferrable, non-sublicensable (other than to Third Party Service Providers) license, solely for the ICR's performance of its allotted tasks under the Programme. The grant of any sub-licence by the ICR to any Third Party Service Provider shall be without prejudice to the ICR's obligations under this Agreement. Any act or omission of any such Third Party Service Provider which, if it were the act or omission of the ICR would be a breach of any of the provisions of this Agreement, will be deemed to be a breach of this Agreement by the ICR who will be liable to the Company accordingly, unless the ICR terminates such sub-licence to such Third Party Service Provider or cures any such breach.

7.3 Subject to the terms and conditions of this Agreement, the Company hereby grants to the ICR a non-exclusive, fully paid-up, non-transferrable, non-sublicensable license to screen any Compound Library and to use the ICR Internal Research Program Hits (if applicable), and its rights in the ICR Research Target IP (including for the avoidance of doubt, the identity of the ICR Internal Targets as the biological target(s) of an ICR Internal Research Program Hit(s)) solely for the ICR's performance of the ICR Internal Research (which in the case of the ICR Internal Research Program Hits will solely be for the purpose of the ICR Internal Research up to the point of notification to the JSC pursuant to Clause 5.2), or clarity, the ICR shall (a) not practice such license prior to the [***] of the Effective Date, and (b) practice such license only in accordance with Clause 5.

7.4 The Company hereby grants to each of the CRT and the ICR a non-exclusive, sub-licensable to Third Party Service Provider(s), fully paid up, perpetual, irrevocable, worldwide, licence to use the Programme Intellectual Property comprised within the Company Non-Compound Intellectual Property, for non-commercial academic research and teaching purposes. In addition, the Company acknowledges that nothing herein prohibits the ICR from purchasing or synthesizing any Programme Compound that is commercially available from a commercial reagents supplier for academic research purposes, and that the ICR shall be permitted to use such Programme Compound in accordance with the terms under which it is supplied to the ICR (or made available to academic researchers) by such supplier (if applicable).

7.5 CRT and/or the ICR shall not use or deal with the Programme Intellectual Property except as expressly permitted by this Agreement.

7.6 The Parties acknowledge that the CTU may conduct screening of the CRT/ICR Background Library pursuant to the Programme and that any CRT/ICR Background Hit Compounds may be encumbered by Third Party rights. The Parties further acknowledge that any particular compound identified by the CTU for inclusion in the Compound Library during the assembly of the Compound Library (and not as result of screening of, or utilizing, the CRT/ICR Background Library pursuant to the Programme) may also be a member of the CRT/ICR Background Library as of the date such compound was or is identified by the CTU for inclusion in the Compound Library (each, a "*Duplicate Compound*"). CRT and the ICR shall promptly notify the Company in the event any Duplicate Compound is identified after the Effective Date. Each CRT/ICR Background Hit Compound (including those that are deemed a Duplicate Compound) shall be subject to appropriate internal encumbrance checks by the ICR to determine whether there are any Third Party encumbrances which may prevent or limit use of such CRT/ICR Background Hit Compounds within the Programme.

(a) Except in the case of Duplicate Compounds (which are addressed in Clause 7.6(b)), in the event that a CRT/ICR Background Hit Compound is subject to a Third Party encumbrance, CRT and the ICR shall promptly notify the Company, which notice shall describe such encumbrance and include reasonable evidence supporting the existence and description of such encumbrance (to the extent not otherwise prohibited by any confidentiality obligations of CRT and the ICR), and such CRT/ICR Background Hit Compound shall not be available to the Programme. To the extent not prohibited by a Third Party encumbrance existing as of the date that any such CRT/ICR Background Hit Compound has been identified: (i) the ICR shall disclose the structure of each such CRT/ICR Background Hit Compound (each an "Unencumbered CRT/ICR Background Hit Compound") to the Company; (ii) CRT and the ICR each hereby grants to the Company an irrevocable, perpetual (subject to Clause 8.5 and 20.2), non-exclusive, worldwide licence, with the right to grant sub-licences through multiple tiers, under all of CRT's and the ICR's respective rights in and to the CRT/ICR Background Library Intellectual Property covering such Unencumbered CRT/ICR Background Hit Compound to research, develop, have developed, use, keep, make, have made, market, import, offer for sale, sell and otherwise dispose of Protein Degradation Products in the Field. The ICR and CRT (in each case to the extent applicable to such Party's activities on behalf of the CTU only) shall not commercialise, or enter into an agreement after the Effective Date to enable any Third Party to commercialise, any of the Unencumbered CRT/ICR Background Hit Compounds whose primary mode of action is as a CRBN-binder. For the avoidance of doubt, no chemical structure disclosure or any other compound identifier disclosure shall be made by the ICR to the Company in relation to any CRT/ICR Background Hit Compounds.

(b) In the case of each Duplicate Compound, in the event that the CRT/ICR Background Hit Compound is subject to a Third Party encumbrance (provided that, the CTU registers such Duplicate Compound made or acquired by the CTU for inclusion of such compound in the Compound Library during the assembly of the Compound Library on the CTU's CRBN project entry under CRT's internal compound database, and upon registration identifies that such encumbrance was in place prior to: (i) in the case of such Duplicate Compound being synthesized internally by the CTU, the date of completion of such Duplicate Compound synthesis (being such Duplicate Compound isolated and characterized to prove it has the correct identity and purity, as recorded in the CTU's electronic lab notebook); or (ii) in the case of such Duplicate Compound being synthesized externally by a Third Party on the CTU's behalf, or purchased by the CTU from an external supplier, the date at which such Duplicate Compound is received at the ICR as recorded in the goods received record (in each case, the date described in clause (i) or (ii), the "Encumbrance Date")), CRT and the ICR shall promptly notify the Company, which notice shall describe such encumbrance and include reasonable evidence supporting the existence and description of such encumbrance (to the extent not otherwise prohibited by any confidentiality obligations of CRT and the ICR), and such CRT/ICR Background Hit Compound shall no longer be deemed a Programme Compound and shall not be available to the Programme. In the event that CRT and the ICR notify the Company that there is a possibility of such encumbrance being worked around such that a particular encumbered CRT/ICR Background Hit Compound might be available to work on under the Programme under certain conditions, the Parties shall discuss at the time and may agree the terms and conditions under which such CRT/ICR Background Hit Compound may be made available to work on under the Programme ([***]). To the extent not prohibited by a Third Party encumbrance existing as of the Encumbrance Date upon registering such Duplicate Compound in accordance with the CTU procedures described above: (i) subject to clause (ii), such Duplicate Compound shall remain a Programme Compound; and (ii) such Duplicate Compound shall also remain a part of any CRT/ICR Background Library and available for CRT and the ICR to use for any purpose, provided that the ICR and CRT (in each case to the extent applicable to such Party's activities on behalf of the CTU only) shall not commercialise, or enter into any agreement after the Effective Date to enable any Third Party to commercialise, any of such Duplicate Compounds whose primary mode of action is as a CRBN-binder. For the avoidance of doubt, the commercialisation by ICR and CRT of a Duplicate Compound that is directed against a target other than CRBN binding and has only residual or minimal binding to CRBN such that such residual or minimal binding is not necessarily required for its main therapeutic activity and/or has not been pursued in development shall be permitted.

7.7 CRT and the ICR each hereby grants to the Company a non-exclusive, fully paid up, sub-licensable through multiple tiers, world ide, licence to use the ICR Internal Research Improvements for all purposes.

7.8 Subject to the terms and conditions 0£ this Agreement, the Company and CRT hereby grants to the ICR a non-exclusive, sublicensable through multiple tiers, worldwide license to (a) use the ICR Internal Targets corresponding to the ICR Internal Research Program Hits and its rights in the ICR Research Target IP for the purpose of enabling the ICR and CRT to conduct the Non-Degradation Program pursuant to Clause 5.2(c) if neither the Company nor Versant Ventures obtains rights to such Non-Degradation Program as provided therein, and (b) use its rights in the ICR Research Target IP for the purpose of enabling the ICR and CRT to commercialise the Non-Degradation Program pursuant to Clause 5.2(c) if neither the Company nor Versant Ventures obtains rights to such Non-Degradation Program as provided therein. CRT and the ICR each hereby grants to the Company a non-exclusive, fully paid up, sub-licensable through multiple tiers, to its rights in the ICR Research Target IP for the purpose of enabling the Company to conduct the Programme and develop, manufacture and commercialize the Products discovered, developed and/or generated pursuant to any of the drug discovery projects which were initiated by the ICR pursuant to the Programme.

7.9 Except as explicitly set forth in this Agreement, no Party shall be deemed by estoppel or implication to have granted any other Party any license or other right to any intellectual property of such Party and each Party reserves all rights not otherwise expressly granted hereunder.

8. PERFORMANCE

- **8.1** The Company shall use Commercially Reasonable Efforts at all times during the Term to:
 - (a) develop one or more Products suitable for use in human clinical trials;
- **(b)** pursue Regulatory Authorisation and (where applicable) Price Approvals in at least [***] Major Markets for each Product in clinical development;
- (c) introduce and commercialise each Product in a Major Market where Regulatory Authorisation and, where applicable, Price Approval for such Product has been obtained; and
- (d) without prejudice to the generality of the foregoing, develop, pursue Regulatory Authorisation and (where applicable) Price Approvals, and commercialise (where Regulatory Authorisation and, where applicable, Price Approval for such Product has been obtained) at least one Product with an application in an Oncology Indication.
- **8.2** In the event that a Product is launched or ready to be launched in the United Kingdom, the Company will use Commercially Reasonable Efforts to cause such Product to be made available throughout the United Kingdom at an Affordable Price. In the event that the Company determines, through its exercise of Commercially Reasonable Efforts, that the best means of making a Product available throughout the United Kingdom at an Affordable Price is through a public assistance programme funded by the Company, the Parties shall discuss in good faith a proportionate adjustment to the royalty rates set forth in Clause 9.2(d).

- **8.3** The Company shall keep CRT, and IGR fully informed of the progress made towards reaching the Milestone Events specified in Clause 9.1 and shall notify CRT, the ICR and the JSC promptly of the achievement of each Milestone Event.
- **8.4** If Company or any of its Affiliates or Sub-Licensees intends to undertake a Phase I Trial of any Product, Company shall notify the ICR and/or CRT, providing particulars in reasonable detail of proposed investigation. At the ICR's and/or CRT's request, Company shall enter into negotiations to allow the ICR and the Royal Marsden NHS Foundation Trust and CRT through CRUK to conduct and/or participate in the Phase I Trial, subject to agreement in good faith of terms acceptable to the Company, the ICR, Royal Marsden NHS Foundation Trust and/or CRT (or CRUK) (as applicable). To the extent that the development of Products by the Company and/or its Sub-Licensees is to involve further clinical studies in the United Kingdom, CRT through CRUK and/or the ICR shall be entitled to participate in such clinical studies with the Company as the Sponsor, subject to agreement in good faith of terms acceptable to the Company, CRT through CRUK and/or the ICR. The terms and conditions under which such clinical studies are to be conducted shall be set out in a separate agreement from this Agreement which shall define, among other things, the clinical studies to be performed, the timetable to which they shall be carried out and the resources and the finances which the Company will provide to enable the studies to be completed in an expeditious manner.

8.5 In the event CRT and the ICR .have reasonable concerns about the Company's efforts with respect to a specific drug discovery project (including any abandonment thereof), which originated from the conduct of the Programme and has reached lead series identification, it shall notify the JSC of such concern and the JSC shall discuss the matter in good faith, including whether the stage of development and/or resources allocated to such project is reasonably justified as part of Company's global strategy to support the further development and commercialisation of Products, and potential solutions to address the concerns of CRT and the ICR, such as whether CRT and the ICR may assist with the development of such project (which could include the Parties collaborating on further research and development activities on a shared risk/reward basis, for example, following identification of a pre-clinical candidate, the Company collaborating with CRT and/or the ICR on initial or further clinical studies to develop in an Oncology Indication (which may involve CRT and/or the ICR conducting such a proof of concept study with CRUK, the ICR, and/or Royal Marsden NHS Foundation Trust)), or receive a handback of such project to enable CRT and the ICR or (as directed by CRT and the ICR) another entity to continue to develop such project, in each case only if any such action will not adversely affect any part of the Programme or the further development and commercialisation of any Product.

9. CONSIDERATION

In consideration of the rights granted under this Agreement, the Company shall pay to CRT:

- 9.1 the following payments ("Milestone Payments") within [***] of the first occurrence of each of the following events ("Milestone Events"):
 - (a) [***];
 - **(b)** [***];
 - (c) [***]; and
 - (d) [***].

For clarity, each Milestone Event set forth in Clauses 9.1(a) and 9.1(c) shall be payable only one (1) time, for the first achievement of such Milestone Event and each Milestone Event set forth in Clauses 9.1(b) and 9.1(d) shall be payable only one (1) time for a particular Product, regardless of how many times such Milestone Event is achieved with regards to a particular Product. Upon the occurrence of each Milestone Event with respect to a particular Product, any Milestone Event listed before it in this Clause 9.1, which is otherwise applicable to such Product and which has not occurred, shall be deemed to have occurred and the corresponding payment made to CRT. A Milestone Event may be triggered by the actions of the Company, its Affiliates, their Sub-Licensees or any person acting on behalf of the Company or its Sub-Licensees.

- **9.2** royalties on a Product by Product, and country by country basis until the later of:
- (a) the date when the manufacture, use, offer for sale, sale or importation of a Product is no longer covered by a Valid Claim in the country of sale, use or manufacture;
- **(b)** the expiry of [***] from the First Commercial Sale of such Product in the relevant country by the Company or its Affiliate or Sub-Licensee;
 - (c) the expiry of any Extended Exclusivity Period in the relevant country; and
 - (d) at the following royalty rates, which shall apply to the respective tiers of aggregate Net Sales of each Product in a given Year:

Product
First Product for which Net Sales occur

First Product for which Net Sales occur

First Product for which Net Sales in excess of [***] per cent ([***])%, on that portion of Net Sales of such Product which is equal to or exceeds [***] occurs

Other Products

[***] per cent ([***])%, on that portion of Net Sales of such Product which is less than [***]

Other Products

[***] per cent ([***])%, on that portion of Net Sales of such Product which is equal to or exceeds [***]

in each case within [***] of the end of each [***] in which the relevant Net Sales are invoiced by the Company, its Affiliates or their respective Sub-Licensees.

9.3 For the purposes of this Clause 9, a Product shall be a different from another Product if such Product consists of, comprises or contains a different Compound, as compared to any other Product.

10. PAYMENT AND STATEMENT

10.1 All payments due to CRT under this Agreement shall be made in the Currency in cleared funds to the account of "Cancer Research Technology Limited":

Sort code: [***]
BIC No: [***]
Dollar Account No: [***]
IBAN No: [***]
Account Name: [***]
Address: [***]

or such other account details as CRT may notify to the Company from time to time.

- **10.2** Where Products are sold by the Company, its Affiliates or Sub-Licensees in a currency other than the Currency, the rate of exchange to be used for converting such other currency into the Currency shall be a recognized independent third party rate of exchange which is the same as that used throughout the accounting system of the Company, its Affiliate or Sub-Licensees, as applicable, for such Year.
 - **10.3** All costs of transmission or currency conversion shall be borne by the Company.
- 10.4 All payments to CRT and the ICR under this Agreement are expressed to be exclusive of value added tax howsoever arising. It is the joint understanding of the Parties that, for VAT purposes, and provided that the Company remains solely established in Switzerland, the place of supply of the services supplied by CRT or the ICR to the Company shall be deemed to be in Switzerland and outside the scope of UK VAT according to article 44 of Council Directive 2006/1 12/EC and the UK implementing legislation as well as article 8(1) of the Swiss VAT Act 2010. Based on this understanding, each of CRT and the ICR shall not be entitled to charge either UK VAT or Swiss VAT on any payments under this Agreement to the Company. Conversely, the Company shall account for Swiss VAT on the purchase of services from abroad under the reverse charge regime according to article 45(1)(a) of the Swiss VAT Act 2010. Any change in the UK VAT legislation (whether it be after the UK leaves the European Union or otherwise) that requires CRT or the ICR to account or and pay UK VAT to HMRC on payments received by CRT or the ICR under this Agreement, shall require the Company to pay such VAT to CRT or the ICR in addition to those payments against receipt of a valid VAT invoice or invoices from CRT or the ICR (as applicable). Each Party shall be solely responsible for the payment of all taxes imposed on any income it receives which arises directly from the efforts of the Parties under this Agreement.

- 10.5 All sums payable under this Agreement shall be paid without deduction or deferment in respect of any claims whatsoever and of any taxes except any tax which the Company is required by law to deduct or withhold. If the Company is required by law to make any such tax deduction or withholding, the Company shall give reasonable assistance to CRT or the ICR (as applicable) to claim exemption from or (if that is not possible) a recovery of or credit for the deduction or withholding under any applicable double taxation or similar agreement from time to time in force, and shall promptly give CRT or the ICR (as applicable) proper evidence as to the deduction or withholding and payment over of the tax, including any value added tax, deducted or withheld.
- 10.6 Where CRT or the ICR does not receive payment of any undisputed sums due to it by the due date except where such dispute is resolved in CRT's or the ICR's favour under Clause 11.2, interest shall accrue both before and after any judgment on the sum due and owing to CRT at the rate equivalent to an annual rate of [***] percent ([***]%) over the then current [***], or, if lower, the highest rate permitted under applicable law, until the full amount is paid to CRT or the ICR, without prejudice to CRT's or the ICR's right to receive payment on the due date.
- **10.7** Within [***] after the end of each [***], the Company, shall send to CRT a written statement detailing in respect of that [***] (including a nil report if appropriate):
 - (a) any Milestone Payments which became due to CRT;
- **(b)** the aggregate Net Sales in respect of that [***] for each Product in each country of the Territory and the total royalties payable to CRT in respect of Net Sales;
 - (c) the type and value of deductions made in the calculation of Net Sales by each Product, and country;
 - (d) any currency conversions, showing the rates used;
 - (e) any further information necessary for the calculation of Net Sales of Products and/or the royalties on Net Sales due to CRT; and
 - **(f)** the amount of the royalties due to CRT in respect of that [***].
 - **10.8** The Company shall give CRT prompt notice upon the occurrence of the achievement of any Milestone Event.

11. ACCOUNTS

- 11.1 The Company shall:
- (a) keep and notwithstanding termination of this Agreement, maintain and shall procure that each of its Sub-Licensees, and each of its Affiliates and their respective Sub-Licensees and subcontractors, as applicable, keep and maintain, for at least [***], complete and accurate accounts and records (including any underlying documents supporting such accounts and records) in sufficient detail to enable the amount of all sums payable under this Agreement to be determined; and

(b) during the Term and thereafter until the said period of [***] relevant to the accounts and records has expired, at the reasonable request of CRT (subject to Clause 11.2) at the expense of CRT, permit or procure permission for an independent, chartered accountant nominated by CRT, and reasonably acceptable to the Company, to inspect the accounts and records kept pursuant to Clause 11.1(a) to confirm any payments due hereunder for a period covering not more than the preceding [***]. Such audits shall not be performed more frequently than [***], except in the event of a dispute under Clause 11.3 below, and no more frequently than once with respect to records covering any given period. Such audits may be exercised upon reasonable prior written notice to the other Party and shall be conducted during regular business hours. Subject to receiving not less than [***] written notice, the Company shall at the request of CRT assemble in one location all such relevant accounts and records of the Company, Affiliates and Sub-Licensees. All information disclosed in such audit shall be used only for the purpose of verifying payments or compliance with this Agreement and shall be treated as Confidential Information of the Party subject to audit, subject to the obligations of this Agreement.

11.2 CRT shall bear the full cost of such audit; provided that if, following any inspection pursuant to Clause 11.1(b), CRT's nominated accountant certifies to CRT that the payments in respect of any Quarter or Year fall short of the sums which were properly payable in respect of that Quarter or Year under this Agreement, CRT shall send a copy of the certificate to the Company and the Company shall (subject to Clause 11.3) within [***] of the date of receipt the certificate pay the shortfall to CRT and, if the shortfall exceeds [***] per cent ([***]%) of the sum properly payable, the Company shall also reimburse to CRT the reasonable costs and expenses of CRT in making the inspection.

11.3 If within [***] of the date of receipt by CRT of any certificate produced pursuant to Clause 11.2, Company notifies the other Party in writing that it disputes the certificate, then dispute shall be referred for resolution by Expert in accordance with Clause 32.1. For the duration of the Service Term and [***] thereafter, the ICR shall keep and maintain complete and accurate accounts and records (including any underlying documents supporting such accounts and records) of any outsourcing costs that ICR has incurred in the course of conducting the Programme in sufficient detail to enable the amount of all such sums reimbursed by Company under this Agreement to be determined. At Company's request (not more than [***] during the Service Term and for [***] after), the ICR shall provide the Company with proof of the ICR having incurred any such outsourcing costs.

12. INTELLECTUAL PROPERTY PROTECTION, PROCEEDINGS AND COSTS

12.1 Subject to Clause 12.2, 12.3 and 12.4, the Company shall have the first right, but not the obligation, for preparing; filing, prosecuting and maintaining (including any patent interference, reexamination, *inter partes* review, post-grant review, reissue, revocation, opposition and appeal proceedings) any Programme Patents, which, during the Term, will be in accordance with the strategy agreed by the JSC. CRT and the ICR shall provide the Company all reasonable assistance and cooperation requested in the Programme Patent preparation, filing, prosecution and maintenance efforts provided in this Clause 12.1, including providing any reasonably necessary and suitably limited powers of attorney and assignments and executing any other required documents or instruments for such filing, prosecution and maintenance.

- **12.2** The Company shall keep CRT and the ICR reasonably informed as to the prosecution and maintenance status of the Programme Patents. At Company's sole discretion, Company shall be at liberty to instruct its patent agent to copy CRT and the ICR into correspondence sent by the patent agent in which case Company shall be deemed to have complied with this Clause 12.2.
- **12.3** The following shall apply in respect of any part of the Programme Patents which directly relates to a Product discovered, developed and/or generated in a specific drug discovery project which was initiated by the ICR pursuant to the conduct of the Programme (the "Selected Programme Patents"):
- (a) If Company elects to stop the prosecution and maintenance of any part of the Selected Programme Patents, Company shall notify CRT and ICR in writing of such decision (and (if applicable) any associated Commercial Delay Rationale (as defined below)), but no later than [***] prior to the expiration of any applicable time bars. During the aforementioned [***] notice period, Company shall retain the responsibility for the prosecution and maintenance of the Selected Programme Patent in question. On expiry of such notice period unless Company demonstrates to [***] (the "Commercial Delay Rationale"):
- (i) Company shall, at CRT's (of if CRT declines, ICR's) request, promptly provide copies to CRT (or any person nominated by CRT, or if CRT declines, ICR) any and all documents and information in Company's control relating to such Selected Programme Patents; and
 - (ii) CRT (or if CRT declines, ICR) shall be free to prosecute or abandon such Selected Programme Patents at its sole discretion.

Should CRT and/or ICR dispute the Commercial Delay Rationale set out in the notice, the matter may be referred for determination in accordance with Clause 32.

- **12.4** Each Party will notify the other, in writing as soon as reasonably practicable after it becomes aware that any claim is made or threatened against CRT, the ICR, the Company or a Sub-Licensee (in the case of the latter, where this is made known to the Company) by any Third Party that the exercise by CRT, the ICR, the Company or such Sub-Licensee of the rights granted pursuant to this Agreement infringes any Patent or other rights of any Third Party.
- 12.5 In the event of the circumstances described in Clause 12.5 arising, (a) the Company shall use Commercially Reasonable Efforts to take such steps as may be necessary in order to address as appropriate any such alleged infringement by the Company, its Affiliates or its Sub-Licensees, and CRT and the ICR shall, at the Company's cost, give it all reasonable co-operation in this regard, and (b) each of CRT and the ICR shall use Commercially Reasonable Efforts take such steps as may be necessary in order to address as appropriate any such alleged infringement by CRT and/or the ICR, and the Company shall, at cost of CRT and the ICR, give them all reasonable co-operation in this regard.

- **12.6** Each Party will notify the other in writing as soon as it becomes aware of any infringement or suspected infringement by a third party of any of the Programme Patents or any unauthorised use of the Programme Know How and/or Programme Materials and the Company shall have the sole right, but not the obligation to:
- (a) at its own cost, and subject to Clause 8, bring proceedings in its own name for infringement of the Programme Patents or misappropriation of and/or unauthorized used the Programme Know How and/or Programme Materials; and
- **(b)** in any such proceedings settle any claim for infringement of the Programme Patents or misappropriation of and/or unauthorized used the Programme Know How and/or Programme Materials.
- 12.7 In any such proceedings, CRT and the ICR shall, at the Company's cost, promptly provide the Company with all documents and assistance as the Company may reasonably require, including joining in any such proceeding and/or providing any reasonably necessary and suitably limited powers of attorney. The Company shall promptly provide CRT and the ICR with notice of such proceedings, keep CRT and the ICR regularly informed of progress and promptly provide CRT and the ICR with such information as CRT and/or ICR may reasonably require, including copies of all documents filed at court or other tribunal in such proceedings. [***].

13. REPRESENTATIONS, WARRANTIES AND COVENANTS

- **13.1** Each Party represents and warrants to the other Parties that:
 - (a) such Party has the full power and authority to enter into this Agreement and to carry out the provisions hereof;
- **(b)** such Party has taken all necessary action on its part to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder;
- (c) this Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, valid, binding obligation, enforceable against it in accordance with the terms hereof;
- (d) the execution, delivery and performance of this Agreement by such Party does not conflict in a material manner with any agreement or any provision thereof, or any instrument or written understanding to which it is a party or by which it is bound, nor violate any law or regulation of any court, governmental body or administrative or other agency having jurisdiction over such Party;
- (e) no government authorization, consent, approval, license, exemption of or filing or registration with any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, under any applicable laws, rules or regulations currently in effect, is or will be necessary for the execution of this Agreement; and
- (f) in respect of the subject-matter of this Agreement, it will not employ any individual or entity debarred by the FDA (or subject to a similar sanction of any equivalent Competent Authority outside the United States of America) or, to its knowledge, that is the subject of any FDA debarment investigation or proceeding (or similar proceeding of any equivalent Competent Authority outside the United States of America), it being understood that details of the subjects of FDA debarment investigations are not published and a Party would not make any particular enquiries of relevant individuals outside such Party's standard policies and procedures.

- 13.2 Except as otherwise expressly set forth in this Agreement, NO PARTY MAKES ANY REPRESENTATION OR EXTENDS ANY WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING ANY WARRANTY THAT THEIR EXERCISE OF ANY INTELLECTUAL PROPERTY RIGHT UNDER THIS AGREEMENT DOES NOT INFRINGE ANY INTELLECTUAL PROPERTY RIGHTS OF THIRD PARTIES, AND EXPRESSLY DISCLAIMS ALL WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.
- 13.3 No claim for breach of any of the representations or warranties in Clause 13.1 may be brought against CRT, the ICR or the Company, as applicable, after [***] has elapsed from the Services Expiry Date, after which date CRT, the ICR and the Company shall each be fully and completely discharged from any liability for breach of any of the representations or warranties in Clause 13.1.
- **13.4** The Company's aggregate liability, and CRT's and the ICR's combined aggregate liability, for breach of one or more of the representations or warranties in Clause 13.1 and clause 6 of the License Agreement shall be limited to [***], provided that CRT's and the ICR's liability shall be several and not joint.
 - 13.5 Covenants of the Parties. The Company covenants to CRT and the ICR, and each of CRT and the ICR covenants to the Company, that:
- (a) all employees, agents or subcontractors of such Party or its Affiliates conduct activities under the Programme will be under the obligation to assign all right, title and interest in and to their inventions and discoveries, whether or not patentable, made, generated or developed in the course and as a result of such work to such Party as the sole owner thereof;
- **(b)** Such Party shall perform its activities pursuant to this Agreement in compliance (and shall ensure compliance by any of its subcontractors) in all material respects with all applicable laws, in each case as applicable under the applicable laws and regulations of the country and the state and local government wherein such activities are conducted.

14. INDEMNITY ·

14.1 The Company shall indemnify and hold harmless the Indemnified Parties from and against any and all liabilities, losses, damages, costs and expenses, including, without limitation, reasonable legal fees (collectively, "Losses") arising out of or resulting from any and all Third Party suits, claims, actions, proceedings, investigations or demands ("Claims") in connection with the exercise by the Company or a Sub-Licensee of any rights to the Programme Intellectual Property, or based upon or in connection with the development, manufacture or commercialization of any and all Products by or on behalf of the Company or any of its Affiliates or Sub-Licensees, except, in each case to the extent such Losses arise from (a) the gross negligence or willful misconduct of such Indemnitee; or (b) a material breach of any representation, warranty, covenant or obligations under this Agreement by CRT or the ICR.

14.2 Promptly after receipt by the Company of any claim or alleged claim or notice of the commencement of any action, administrative or legal proceeding, or investigation to which the indemnity provided for in this Clause 14 may apply, the Company shall give written notice to CRT and the ICR of such fact. In the event that any Indemnitee entitled to indemnification under Clause 14.1 is seeking such indemnification, such Indemnitee shall (a) give written notice to the Company of the Claim as soon as reasonably practicable after such Indemnitee receives notice of such Claim, (b) permit the Company to assume the direction and control of the defence of such Claim (including the sole right to settle it at the sole discretion of the Company, taking into consideration in good faith any reasonable concerns or objections raised by the Indemnitee; provided that such settlement does not impose any obligation on, or otherwise adversely affect, the Indemnitee or other Party), (c) cooperate as reasonably requested (at the expense of the Company) in the defence of such Claim, and (d) undertake all reasonable steps to mitigate any loss, damage or expense with respect to such Claim. If the Company fails to assume the defence thereof by providing the relevant Indemnitee notice of such election in writing within [***] after receipt by the Company of notice of such Claim, the Indemnitee may assume such defence and for the avoidance of doubt the indemnity in Clause 14.1 shall extend to the reasonable legal and other expenses consequently incurred in connection with such defence to the extent such Indemnitee is entitled to indemnification under Clause 14.1. The Company and the relevant Indemnitee will co-operate in good faith in the conduct of any defence, provide such reasonable assistance as may be required to enable any claim properly to be defended and the Party with conduct of the action shall provide promptly to the other Party copies of all correspondence and documents and notice in writing of the substance of all oral commun

14.3 Should the Company assume conduct of the defence, the Indemnitee may retain separate legal advisers, at its sole cost and expense, save that if the Company wrongly denies the applicability of the indemnity or reserves its position in relation to the same, the indemnity in this Clause 7 shall extend to the Indemnified Party' reasonable costs and expenses so incurred.

15. INSURANCE

Reasonably prior to Commencement of a Phase I Trial, the Company shall put in place and thereafter maintain in accordance with the terms of this Clause 15, at its own cost, comprehensive product liability insurance and general commercial liability insurance. The Company shall ensure at CRT's and/or the ICR's request, that CRT and/or the ICR interest as co-assured be noted on the policy or policies. Within [***] of the Effective Date and of the beginning of each policy period, the Company shall, upon the reasonable request of CRT and/or the ICR, provide CRT and/or the ICR with a certificate evidencing the coverage required hereby, and the amount hereof and the noting of CRT and/or the ICR's interest. Such insurance shall be with a reputable insurance company and shall be maintained for not less than [***] following the expiration/termination of this Agreement for any reason or, if such coverage is of the "claims made" type, for [***] following the expiration or termination of this Agreement for any reason.

16. LIMITATION OF LIABILITY

16.1 EXCEPT AS SET FORTH IN CLAUSE 16.2, NO PARTY, NOR CRUK, NOR THEIR RESPECTIVE OFFICERS, EMPLOYEES AND AGENTS SHALL HAVE LIABILITY WHETHER UNDER STATUTE OR IN TORT (INCLUDING BUT NOT LIMITED TO NEGLIGENCE), CONTRACT OR OTHERWISE TO ANY OTHER PARTY AND/OR CRUK IN RESPECT OF ANY CONSEQUENTIAL, INDIRECT OR PUNITIVE DAMAGES IN CONNECTION WITH THIS AGREEMENT, INCLUDING LOSS OF GOODWILL, OPPORTUNITY, PROFIT OR CONTRACT; PROVIDED THAT, BECAUSE MONETARY DAMAGES MAY NOT ADEQUATELY COMPENSATE A PARTY FOR A BREACH OF THE CONFIDENTIALITY PROVISIONS OF CLAUSE 17 BY ANY OTHER PARTY AND REMEDIES UNDER CLAUSE 17.6 MAY BE INADEQUATE OR UNAVAILABLE WITH RESPECT TO SUCH A BREACH, THE FOREGOING LIMITATION ON LIABILITY WILL NOT LIMIT A PARTY'S RIGHT TO RECOVER FROM ANY OTHER PARTY CONSEQUENTIAL OR INDIRECT DAMAGES (BUT NOT PUNITIVE DAMAGES) SUSTAINED AS A RESULT OF BREACH BY SUCH OTHER PARTY OF THE CONFIDENTIALITY PROVISIONS OF CLAUSE 17, BUT TO THE EXTENT ANY SUCH DAMAGES ARISE AS A RESULT OF A BREACH OF CLAUSE 17, THE BREACHING PARTY'S LIABILITY FOR ANY SUCH DAMAGES SHALL BE LIMITED TO [***] AND CRT'S AND THE ICR'S LIABILITY FOR ANY SUCH DAMAGES SHALL BE SEVERAL AND NOT JOINT.

16.2 Nothing in this Agreement shall be construed as excluding or limiting the liability of any Party or CRUK or any of their respective officers, employees and agents to the other Party for death or personal injury of any person resulting from the negligence or willful misconduct of such persons.

17. CONFIDENTIALITY

- **17.1** Pursuant to Clause 18, each Party (the "*Receiving Party*") shall keep, and it shall cause its respective directors, employees, agents and subcontractors to keep, secret and confidential all Confidential Information of any other Party which is disclosed to it by or on behalf of the other Party (the "*Disclosing Party*") and shall not use or disclose to any person whatsoever such Confidential I formation or any part of it, other than:
- (a) in the case of the- Company, disclosure to Sub-Licensees and Third Party Service Providers, subject to compliance with Clause 22.3, provided that each such person shall be bound by obligations of non-disclosure and non-use no less stringent than those in this Agreement, and as necessary in communications with Competent Authorities in the Territory relating to Products;
- **(b)** in the case of each Party, to its Affiliates and their respective directors, officers, employees or agents, in connection with the performance of its obligations or exercise of its rights under this Agreement, *provided* that each such person shall be bound by obligations of non-disclosure and non-use no less stringent than those in this Agreement;
- (c) in the case of the Company, to potential or actual acquirers, merger candidates or investors or venture capital firms, investment bankers or other financial institutions or investors, provided that in connection with such disclosure, the Company shall inform each disclose of the confidential nature of such Confidential Information and cause each disclose to treat such Confidential Information obligations of non-disclosure and non-use no less stringent than those in this Agreement;

- (d) in the case of CRT and ICR, to CRT/ICR Reviewers in accordance with Clause 17.4;
- **(e)** in the case of CRT and ICR, to any actual or potential royalty purchaser contemplated by Clause 22.2; *provided* that no chemical structures shall be disclosed to any or all such disclosees and each such disclosee shall be bound by written obligations of non-disclosure and non-use no less stringent than those in this Agreement; or
 - (f) to the extent expressly authorized by this Agreement or otherwise agreed to in writing,

In respect of any disclosures of any Confidential Information made by a Party to such Party's directors, officers, employees or agents, subcontractors or, in the case of the Company, Sub-Licensees, Third Party Service Providers, potential or actual acquirers, merger candidates or investors or venture capital firms, investment bankers or other financial institutions or investors, the Disclosing Party shall be responsible for any failure by such recipient to treat the disclosed Confidential Information as required under this Clause 17.

- **17.2** The provisions of Clause 17.1 shall not apply to any Confidential Information of the Disclosing Party which the Receiving Party can demonstrate by competent evidence:
 - (a) was in its possession (other than under an obligation of confidence to any other Party or to a Third Party) at the date of receipt;
 - (b) was generally available to the public or otherwise part of the public domain at the time of its disclosure to the Receiving Party;
- (c) became generally available to the public or otherwise enters the public domain after its disclosure and other than through any act or omission of the Receiving Party in breach of this Agreement or any other obligation of confidentiality owed to the Party communicating such information to the other; and/or
- **(d)** was developed independently by the Receiving Party, as evidenced by its contemporaneous written records, *provided* that Programme Materials and Programme Know How shall be deemed the Confidential Information of the Company and the Company the Disclosing Party and each of CRT and the ICR the Receiving Party with respect thereto.
- 17.3 Notwithstanding the provisions of Clause 17.1, the Receiving Party may disclose Confidential Information, without violating its obligations under this Agreement, to the extent the disclosure is required by a valid order of a court or other governmental body of competent jurisdiction or is otherwise required by law or regulation, *provided* that the Receiving Party shall give reasonable prior written notice to the Disclosing Party of such required disclosure and, at the Disclosing Party's request and expense, shall cooperate with the Disclosing Party's efforts to contest such requirement, to obtain a protective order requiring that the Confidential Information so disclosed be used only for the purposes for which the order was issued or the law or regulation required, and/or to obtain other confidential treatment of such Confidential Information.

17.4 CRT and ICR shall provide the Company, upon reasonable notice, the opportunity to review and comment on any Confidential Information proposed to be disclosed to CRT/ICR Reviewers and shall give due consideration to any comments received from the Company. CRT and ICR recognise the need to protect the confidentiality of the chemical structures of the Compound Library and pre-clinical candidates, and the Programme Know How that is the data associating that Unencumbered CRT/ICR Background Hit Compounds (where identified by chemical structure) bind CRBN. Prior to disclosing the chemical structure of any pre-clinical candidate (to the extent not already in the public domain), CRT and ICR shall, where practicable, limit the content of materials disclosed to CRT/ICR Reviewers to (in the following order of priority): [***]. CRT and ICR will use reasonable endeavours to ensure that any such disclosure to CRT/ICR Reviewers is made in confidence (whether under the terms of an appropriate confidential disclosure agreement with CRT, CRUK or otherwise which has a minimum confidentiality term of [***] from the date of disclosure).

17.5 Subject to Clause 17.4, the provisions of this Clause 17 shall remain in force for a period of [***] from the date of disclosure.

17.6 Given the nature of the Confidential Information and the competitive damage that a party would suffer upon unauthorized disclosure, use or transfer of its Confidential Information to any Third Party, the Parties agree that monetary damages would not necessarily be a sufficient remedy for any breach of this Clause 17. In addition to all other remedies, a Party shall be entitled to specific performance and injunctive and other equitable relief as a remedy for any breach or threatened breach of this Clause 17.

18. PUBLICATIONS

18.1 Any publication of Programme Intellectual Property generated by the ICR will be led by the ICR. Each proposed presentation or publication containing details of work or results of the Programme shall be sent to a reviewer appointed by each the Company, CRT and the ICR at the first meeting of the JSC (the "Reviewers") for review prior to submission to any Third Party for presentation or publication. Each Party shall have the right to replace its Reviewer upon written notice to the other Parties. The Reviewers shall review each such presentation or publication within [***] of its receipt. Whether or not such presentation or publication is to be permitted, with or without amendments, shall be subject to the unanimous, written consent of the Reviewers, taking into account the patentability of the proposed subject matter of disclosure and the value of such information as secret and confidential information. Notwithstanding the foregoing or anything to the contrary herein, at the request of the Company, any Confidential Information of Company which is not Programme Intellectual Property generated by or on behalf of the ICR contained within such proposed presentation or publication shall be removed prior to its submission for presentation or publication by the Party submitting such presentation or publication for review. The Reviewers may decide to postpone presentation or publication by up to [***] to allow for the filing of a patent application or the taking of such other measures as the Reviewers deem appropriate to establish and preserve any proprietary rights in the information in the proposed presentation or publication. If the Reviewers disagree whether such publication should be permitted the dispute shall be referred to the JSC and in the event of a continuing dispute, publication shall be permitted no later than [***] following receipt of the proposal by the Reviewers for review, provided that no scientific paper shall be restricted for publication on the basis that it contains information which is also

18.2 The Company and CRT acknowledge the importance of publications to the academic standing of the ICR. Accordingly, each of the Company, CRT and the ICR shall use Commercially Reasonable Efforts to facilitate publication of the results of the Programme. Publications shall make an appropriate acknowledgment of the respective contributions of the Parties, which may be by co-authorship of the publication as may be scientifically appropriate or customary or as otherwise agreed, on a publication by publication basis.

19. TERM AND TERMINATION

- **19.1** -This Agreement will become effective on the Effective Date and shall remain in full force and effect until terminated in accordance with the provisions of this Clause 19.
 - **19.2** Without prejudice to any other rights of the Parties, this Agreement may be terminated by notice in writing:
- (a) by the Company or by CRT and the ICR (acting together) if the Company, in the case of CRT and the ICR as the terminating Party, or if either CRT and/or the ICR, in the case of the Company as the terminating Party, is in material breach of any of its obligations under this Agreement, for the avoidance of doubt including those of Clause 8, and in the case of a remediable breach fails to remedy such breach within [***] of written notice to the breaching Party containing particulars of such breach and requiring it to be remedied and copied to any other non-breaching Party; provided, however, that if any breach other than a payment breach) is not reasonably curable within [***] and if the breaching Party has provided a reasonable plan for cure of such breach during such [***] period and is making a bona fide effort to cure such breach by diligently implementing such plan, such cure period will be extended for a time period to be agreed by the Parties (but in no event more than an additional [***]) in order to permit the breaching Party a reasonable period of time to cure such breach in accordance with such plan;
- **(b)** by CRT and the ICR (acting together) forthwith if a voluntary arrangement is proposed or approved or an administration order is made, or a receiver or administrative receiver is appointed of any of the Company's assets or undertakings or a winding-up resolution or petition is passed (otherwise than for the purpose of solvent reconstruction or amalgamation) or similar or equivalent action is taken against or by the Company by reason of its insolvency or in consequence of debt, in each case, which is not dismissed, discharged, bonded or stayed within [***] after the filing thereof or initiation of such action;
 - (c) by the Company forthwith if the circumstances described in Clause 19.2(b) apply to the ICR during the Services Term;
- (d) by the Company upon [***] written notice to CRT and the ICR if CRT and/or the ICR challenges or seeks to challenge the validity of the Programme Patents or any of them and each of CRT and the ICR shall forthwith in writing notify the Company of any decision to challenge the Programme Patents or any of them which it makes or of which it becomes aware;

- **(e)** by CRT and the ICR (acting together) forthwith upon written notice to the Company in the event of any transaction between the Company and a Tobacco Party in which a Tobacco Party acquires more than [***] of the voting equity securities of the Company;
- **(f)** by CRT and the ICR (acting together) upon [***] written notice to the Company if the Company permanently abandons all discovery, development and commercialization efforts for all Products (i) as indicated in a written notice by the Company to the ICR and CRT or (ii) as demonstrated by failure by the Company and its Affiliates and any Sub-Licensees to conduct any such efforts for a consecutive [***] period or longer, unless the Company provides reasonable evidence to CRT and the ICR within such [***] period showing that it or any of its Affiliates or Sub-Licensees has conducted any such efforts during the prior [***] period; or
- **(g)** by the Company upon [***] written notice to CRT and the ICR if the JSC determines that the continuation of the Programme or development of Products would be commercially unreasonable, scientifically unviable, illegal, unethical or impossible; or
 - (h) for any and/or no reason, by the written agreement of the Parties.

20. EFFECTS OF TERMINATION

- **20.1** Upon the termination of this Agreement for any reason:
- (a) to the extent not already paid by the Company, the Company shall be obligated to make payment to the ICR of the amounts specified in Clause 3.3;
- **(b)** the Company shall, within [***] of notice of termination of this Agreement provide CRT with a final written statement detailing, in respect of the time elapsed since the last report under Clause 10.7, the matters set out in Clause 10.7;
- (c) except to the extent that a Party obtains or retains the right to use Confidential Information of another Party pursuant to the terms of this Agreement, each Party shall promptly return to the other Party, or delete or destroy, all relevant records and materials in such Party's possession or control containing Confidential Information of another Party; provided that such Party may keep one (1) copy of such materials for archival purposes only subject to continuing confidentiality obligations; and
- (d) all other rights and obligations of the Parties shall terminate forthwith, except for those rights and obligations that survive termination of this Agreement as otherwise provided in this Clause 20.

- 20.2 Upon termination of this Agreement by CRT and the ICR (acting together) pursuant to Clause 19.2(a), 19.2(b), 19.2(e) or 19.2(f), by the Company pursuant to 19.2(g) or on mutual agreement pursuant to Clause 19.2(h), effective as of such termination, at the written request of CRT or ICR, the Company shall, and it hereby does, automatically grant to CRT and the ICR (acting together), a right of first negotiation, exercisable within [***] after the termination date, upon commercially reasonable terms and conditions (taking into account CRT and the ICR's needs for drug development and commercialization), which may include revenue-sharing arrangements (which amongst any other relevant factors shall take into account each Party's contribution to the relevant intellectual property), to be negotiated in good faith by the Parties for up to one hundred and [***] from the date of exercise, or for up to an additional [***] period at the election of CRT and the ICR, to obtain a license, under the Company's rights in the Programme Intellectual Property, the CRT/ICR Existing Intellectual Property, the CRT/ICR Existing Compound Library, the ICR Internal Research Program Hits, ICR Research Target IP, if applicable, and any other intellectual property rights owned or in the control of the Company that are reasonably necessary for the development, manufacture or commercialization of any Product, to research, develop, make, have made, market, use and sell Products. If the Parties have not completed a licence agreement by the end of the initial one hundred and [***] period, at the election of CRT and the ICR, the Parties shall seek to resolve any dispute and finalise the licence agreement by escalating to the Executive Officers of the Parties who shall also meet in the first instance (whether in person or via teleconference) within [***] of CRT's or the ICR's communication of such election to the Company to seek resolution in good faith.
- **20.3** Upon termination of this Agreement by the Company pursuant to Clause 19.2(a), 19.2(c) or 19.2(d), the licences granted to the ICR pursuant to Clause 7.3 shall terminate forthwith, provided that the ICR shall have the right to complete the ICR Internal Research with respect to any drug discovery program initiated by the ICR and approved by the JSC prior to the effective date of such termination, subject to the ICR's and the CRT's continued compliance with the terms and conditions of this Agreement, including Clause 5.
- **20.4** The termination of this Agreement howsoever arising, will be without prejudice to the rights and duties of any Party accrued prior to termination. The following Clauses will continue to be enforceable notwithstanding termination: Clauses 1, 6, 7.4, 7.5, 7.6, 7.7, 8, 9, 10, 11 (for the period specified therein), 12, 13.2, 14, 15 (for the period specified therein), 16, 17 (for the period specified therein), 20 (including any clauses referenced therein), 21, 22, 23, 24, 25, 27, 28, 29, 31, 32, 33 and 34.
- **20.5** In the event that Company were to decide that it wanted to destroy quantities of the Compound Library (i.e. the physical material), the Company shall use reasonable endeavors to notify CRT and the ICR and, at CRT's and/or ICR's request and expense, transfer such quantities of the Compound Library to the ICR.

21. FORCE MAJEURE

- **21.1** No Party shall be liable for any delay or failure in performance if such delays are caused by strike, riot, civil commotion, fire, acts of God or other circumstances beyond its reasonable control which circumstance that Party could not reasonably have been expected to take into account at the Effective Date and provided that:
 - (a) the Party so affected shall give prompt notice thereof to the other Parties;
 - (b) the suspension of performance is no greater scope than is required by the Force Majeure; and
 - (c) lack of funds shall not be interpreted as an event or circumstance beyond the reasonable control of a Party.

- **21.2** Subject to Clause 21.3 below, the Party giving such notice shall be excused from such of its obligations hereunder for as long as it continues to be so affected and shall perform its obligations as soon as such circumstances shall cease to affect its operations.
- **21.3** If such force majeure continues unabated for a period of at least [***], the Parties will meet to discuss in good faith what actions to take or what modifications should be made to this Agreement as a consequence of such force majeure in order to alleviate its consequences on the affected Parties.

22. ASSIGNMENT AND SUB-CONTRACTING

- **22.1** This Agreement shall be binding upon and inure to the benefit of the Parties, their permitted successors and assigns. This Agreement shall only be assignable:
- (a) by any Party to any of Affiliate of such Party without the consent of the other Parties provided that such Affiliate is not a Tobacco Party;
 - **(b)** by any Party with the written consent of the others; or
- (c) by any Party without the consent of the other Parties, to any successor to all or substantially all the assets of its business to which this Agreement relates provided that it is not a Tobacco Party.

Any purported assignment in violation of this Clause 22.1 shall be null and void.

- **22.2** Notwithstanding the foregoing, the rights of CRT to Milestone Payments and/or royalties under this Agreement (but, for the avoidance of doubt, not any obligations or duties of CRT) shall be assignable by CRT solely in connection with a transaction with an assignee concerning such rights of CRT's.
- **22.3** No Party may sub-contract its obligations under this Agreement to any Third Party without the prior written consent of the other Parties, *provided* that (i) the Company may sub-contract its obligations under this Agreement to an Affiliate of the Company or a Third Party Service Provider without the consent of the other Parties, and (ii) the ICR may sub-contract its obligations under this Agreement to a Third Party Service Provider without the consent of the other Parties. Each Party shall ensure that an appropriate written agreement is put in place with each permitted subcontractor, and/or in the case of the Company, Third Party Service Provider. Any act or omission of an Affiliate of a Party (and/or a Third Party Service Provider of the Company), which, if it were the act or omission of such Party (and/or the Company, as applicable) would be a breach of any of the provisions of this Agreement, will be deemed to be a breach of this Agreement by such Party who will be liable to the other Parties accordingly.

23. NOTICES

All notices shall be in writing and sent by hand, facsimile, or post and shall be deemed to be properly served (i) if sent by hand on [***], when delivered at the relevant address and if delivered at any other time, on [***]; (ii) if sent by post, [***] after posting; (iii) if sent by facsimile on [***], when transmitted and if sent at any other time, on [***], provided a confirmatory copy is sent by post within [***] of transmission, and shall be sent to the following addresses or facsimile numbers as may be amended by the relevant Party in writing:

For the attention of:

The Company: Monte Rosa Therapeutics AG

Aeschenvorstadt 36

4051 Basel Switzerland Facsimile: [***]

For the attention of: Chief Executive Officer

CRT: Cancer Research Technology Limited

Angel Building 407 St John Street London, EC1V 4AD

England
Facsimile: [***]

For the attention of: The Chief Executive Officer

The ICR: The Institute of Cancer Research: Royal Cancer Hospital

123 Old Brompton Road London, SW7 3RP United Kingdom

For the attention of: The Director of Enterprise

Versant Ventures Versant Ventures

Aeschenvorstadt 36

4051 Basel Switzerland Facsimile: [***]

For the attention of: Chief Executive Officer

24. VARIATION

No variation, modification, amendment, extension or release from any provision hereof shall be effective unless it is in writing, signed by the Parties.

25. ENTIRE AGREEMENT

25.1 Each of the Parties confirms that the Agreement (including all Schedules hereto), together with the Transaction Documents, represents the entire understanding, and constitutes the whole agreement, in relation to its subject matter and supersedes any previous agreement between the Parties with respect thereto.

25.2 Each Party confirms that:

- (a) in entering into this Agreement it has not relied on any representation or warranty or undertaking which is not contained in this Agreement; and
- **(b)** in any event, without prejudice to any liability for fraudulent misrepresentation or fraudulent misstatement, no Party shall be under any liability or shall have any remedy in respect of misrepresentation or untrue statement unless and to the extent that a claim lies under this Agreement or any Transaction Document.

26. FURTHER ASSURANCE

The Parties hereby undertake to do all such other acts and things, and execute and provide all such documents at the requesting Party's cost as may be necessary or desirable to give effect to the purposes of this Agreement.

27. WAIVER

No waiver, release, relaxation, forbearance or indulgence by any Party in enforcing any of the terms or conditions of this Agreement or the granting of time by any Party to any other shall prejudice, affect or restrict the rights and powers of such Party. The waiver of any breach of any term or any condition of this Agreement shall not be construed as a waiver of any subsequent breach of a term or condition of the same or of a different nature. Any waiver by a Party of a particular provision or right shall be in writing, shall be as to a particular matter and, if applicable, for a particular period of time and shall be signed by such Party.

28. SEVERABILITY

- **28.1** If the whole or any part of this Agreement is or becomes or is declared illegal, invalid or unenforceable in any jurisdiction for any reason (including by reason of the provisions of any legislation and/or by reason of any court of competent jurisdiction):
- (a) in the case of the illegality, invalidity or unenforceability of the whole of this Agreement it shall terminate only in relation to the jurisdiction in question; or
- **(b)** in the case of the illegality, invalidity or unenforceability of a part of this Agreement that part shall be severed from this Agreement in the jurisdiction in question and that illegality, invalidity or unenforceability shall not in any way whatsoever prejudice or affect the remaining parts of this Agreement which shall continue in full force and effect and in no circumstances shall sums paid by the Company to CRT or the ICR under this Agreement be repayable.
- **28.2** If in the reasonable opinion of any Party any severance under this Clause 28 materially affects the commercial basis of this Agreement, the Parties shall discuss, in good faith, ways to eliminate the material effect.

29. LAW AND JURISDICTION

This Agreement shall be governed by and construed in accordance with the laws of England and Wales. The Parties agree, subject to Clause 32 below, to submit to the exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this Agreement (except in respect of disputes under Clause 17 where jurisdiction is non-exclusive).

30. EXECUTION

This Agreement may be executed in any one or more number of counterpart agreements each of which, when executed, shall be deemed to form part of and together constitute this Agreement. Facsimile signatures and signatures transmitted via PDF shall be treated as original signatures.

31. ANNOUNCEMENTS AND USE OF NAMES

- **31.1** Save as provided in Clause 31.2 no Party shall make, or procure or permit the making of, any press release or other public announcement in relation to this Agreement without first obtaining the written approval of the other Parties to any such release or announcement, which shall not unreasonably be withheld, conditioned or delayed.
- **31.2** Each Party agrees that it may make any announcement with respect to this Agreement or any ancillary matter as shall be required by law or the regulations of any stock exchange to which it is subject, without the other Parties' consent (which shall not unreasonably be withheld, conditioned or delayed) provided it has used reasonable endeavours in the time available to consult with the other Parties on the terms of any such announcement beforehand.
- **31.3** No Party shall use the name of the other (including in the case where the other is CRT, that of CRUK (or its successor)) other than as provided in Clause 31.1 and 31.2 without the prior written consent of such Party, which shall be at such Party's sole discretion. Notwithstanding the foregoing, (a) to the extent information regarding this Agreement has already been publicly disclosed in accordance with Clause 31.1 and 31.2, a Party may subsequently disclose the same information to the public without the consent of the other Parties, and (b) the Company shall also be permitted to disclose the terms of this Agreement, in each case under appropriate confidentiality provisions substantially equivalent to those of this Agreement, to any actual or potential acquirers, investors, Sub-Licensees, collaborators, and/or professional advisors.

32. DISPUTE RESOLUTION

- **32.1** Insofar as this Agreement provides that a matter shall be resolved by Expert, the opinion of such expert (who shall act as an expert and not as an arbitrator) shall be final and binding on the Parties. In the event of a Party seeking an Expert under this Agreement, each Party shall make written submissions to the expert so appointed and to the other involved Party or Parties within [***] of the appointment. Each involved Party shall have [***] to respond to the other's submissions and the Expert shall be provided within a further [***].
- **32.2** It shall be a condition precedent to the commencement of any action in court or other tribunal (save an action for specific performance, injunctive or other equitable relief in accordance with Clause 17.6) in respect of any dispute relating to this Agreement and to the seeking of an Expert that the Parties have sought to resolve the dispute by one or more Parties notifying it in writing for resolution to the Executive Officers of the Parties who shall meet (whether in person or via teleconference) within [***] of such notice to seek resolution in good faith. If the Executive Officers fail to resolve the dispute within [***] of such meeting the Parties may, but shall not be obliged, to try in good faith to settle the matter by mediation in London under the rules of the Centre for Effective Dispute Resolution.

33. NO THIRD PARTY BENEFICIARIES

Except for the Third Parties and the respective rights referred to in Clause 5.2(c)(iii) and Clause 14 (Indemnity) and Clause 16 (Limitation of Liability) this Agreement is neither expressly nor impliedly made for the benefit of any party other than those executing it whether pursuant.to the Contract (Rights of Third Parties) Act 1999 or otherwise. Notwithstanding the provisions of this Clause 33, the Parties shall be entitled to amend, suspend, cancel or terminate this Agreement or any part of it in accordance with Clause 19, without the consent of any Third Party including those referred to in this Clause 33; *provided* that the prior written consent of Versant Ventures shall be required for any amendment, suspension, or cancellation of Clause 5.

34. LANGUAGE; AUTHORSHIP

This Agreement has been prepared in the English language, and the English language shall control its interpretation. In addition, all notices required or permitted to be given hereunder, and all written, electronic, oral or other communications among the Parties regarding this Agreement, shall be in the English language. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties, and no presumption or burden of proof shall arise or disfavoring any party by virtue of the authorship of any provisions of this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized representatives as of the Effective Date.

CANCER RESEARCH TECHNOLOGY LIMITED

Company Secretary: /s/ Illegible

Print Name: Illegible Date: April 4, 2018

THE INSTITUTE OF CANCER RESEARCH: ROYAL CANCER HOSPITAL

Authorized Signatory: /s/ Illegible

Print Name: Illegible Date: April 6, 2018

[Signature page to Collaboration and Option Agreement]

MONTE ROSA THERAPEUTICS AG (in formation) By: /s/ Illegible Print Name: Illegible Title: Director Date: April 6, 2018 By: /s/ Illegible Print Name: Illegible Title: Professor Date: April 6, 2018 By: /s/ Illegible Print Name: Illegible Title: Company Secretary, Cancer Research Technology Limited Date: April 4, 2018 By: /s/ Illegible Print Name: Illegible Title: Director, ICR Date: April 6, 2018

Versant Venture Capital VI, L.P. By: Versant Ventures VI GP, L.P. By: Versant Ventures VI, GP-GP, LLC

By: /s/ Bradley J. Bolzon

Print Name: Bradley J. Bolzon
Title: Managing Director

Date:

[Signature page to Collaboration and Option Agreement]

SCHEDULE 1

PAYMENT SCHEDULE

[***]

SCHEDULE 2

PROGRAMME

[***]

SCHEDULE 3

CRITERIA

[***]



The Institute of Cancer Research: Royal Cancer Hospital 123 Old Brampton Road London SW7 3RP United Kingdom For the attention of [***]

Monte Rosa Therapeutics AG Aeschenvorstadt 36 4051 Basel Switzerland For the attention of Alexander Mayweg, Member of the Board

25th February 2019

Dear Sirs,

Cancer Research Technology Angel Building 407 St John Street London EC1V 4AD United Kingdom

RE. Licence Agreement between Cancer Research Technology Limited ("CRT"), Institute of cancer Research: Royal Cancer Hospital ("ICR") and Monte Rosa Therapeutics AG ("Company") dated 26th April 2018 (the "License Agreement")

The Parties wish to modify the Licence Agreement pursuant to this letter of amendment ("Amendment 1"). Unless indicated otherwise, all capitalised terms used in this Amendment 1 shall have the meanings given to such terms in the License Agreement.

CRT, the ICR and the Company hereby agree to amend the Agreement, which shall be effective from the date of this letter as follows. For avoidance of doubt, the effective date as particularised under Clause *U.1* of the license Agreement remains unchanged.

1. Clause 1.1 shall be amended to include the following additional definitions:

```
"[***]" means [***].
"[***]" means [***].
```

"Protein Degradation Product Library" means [***].

"Protein Degradation Product HTS" means [***].

"Services Term" has the meaning given in the Collaboration and Option Agreement.

2. Clause 2.1(d) shall be amended by including "and" to read as follows:

"(d) a non-exclusive fully-paid ... in the Field in the Territory; and"

3. Clause 2.1 shall be amended after paragraph (d) to include the following:

"(e) a non-exclusive fully-paid, irrevocable, perpetual licence, with the right to grant sub-licences as provided in Clause 2.2, under all of CRT's and the ICR's respective rights in and to the Non-Compound Intellectual Property listed in schedule 1A(iii) to discover, research, develop, have developed, use, keep, make, have made, market import, offer for sale, sell and otherwise dispose of Licensed Products that are Protein Degradation Products in the Field in the Territory.

Registered address: Cancer Research Technology Ltd, Angel Building, 407 St John Street, London EC1V 4AD.
Registered In England (1626049). VAT registration number GB788 138678.
A wholly-owned subsidiary of Cancer Research UK, registered charity in England and Wales (1089464), Scotland (SC041666) and the Isle of Man (1103).

- 4. Clause 2 shall be amended after 2.6 to include the following:
 - "2.7 For the period of the Services Term only and-subject to Clause 2.8, the CTU shall not undertake any Protein Degradation Product HTS using the [***] outside of the Programme. For clarity, this clause shall terminate in the event any of the circumstances set forth in Clause 19.2 of the Collaboration and Option Agreement are invoked.
 - 2.8 For the avoidance of doubt during the Services Term CTU shall be able to, and each of the ICR and/or CRT .shall be entitled to itself, and/or enable any of its partners or licensees (including by way of granting sub-licenses to the Protein Degradation Desmoplasia assay Know How) to,: (i) conduct any high- throughput screen using the [***] outside of a Protein Degradation Product HTS ("Other HTS"); and/or (ii) use the [***] to progress, develop and commercialise any compound(s) that is identified from an Other HTS or any other research activity conducted outside of the Programme that did not involve a Protein Degradation Product HTS, without any restrictions or obligations to the Company.
- 5. Clause 13.1 shall be deleted and replaced with the following:
 - "13.1 The termination of this Agreement howsoever arising, will be without prejudice to the rights, obligations and duties of any Party accrued prior to termination. The following Clauses will continue to be enforceable notwithstanding termination: Clauses 2.1-2.6, 6.3, 7, 8, 9, 10, 11, and 13-27."
- 6. Schedule 1 Part A shall be amended after last bullet point to (ii) to include the following:
 - "(iii) [***]
 [***]."
- 7. Schedule 1 Part (A) shall be amended to include punctuation namely brackets for "i" and "ii" to read as follows:

```
"[***]"
"[***]"
```

Save as expressly varied by this letter of amendment to the licence Agreement, all other terms of the Licence Agreement remain the same. However, if there are any inconsistencies between the terms of this Amendment 1 and the provisions of the licence Agreement, then this Amendment 1 shall prevail.

This Amendment 1 shall be governed by and construed in accordance with the laws of England and Wales and the Parties agree to submit to the exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this Amendment 1.

Please sign, and add the date of signature, below confirming ICR and Company's agreement to be bound by the terms of this Amendment 1 and return three signed Amendment 1 to the undersigned.

Yours faithfully,					
/s/ Illegible For and on behalf of Cancer Research Technology Limited Name: Illegible Title:					
Countersigned by The Institute of Cancer Research: Royal Cancer Hospital					
Signature:	/s/ Illegible		Date:		
Name:	Illegible		Title:		
Countersigned by Monte Rosa Therapeutics AG					
Signature:	/s/ Illegible		Date:		
Name:	Illegible		Title:		
		3			

The Institute of Cancer Research: Royal Cancer Hospital 123 Old Brampton Road London SW7 3RP United Kingdom For the attention of [***]

Monte Rosa Therapeutics AG Aeschenvorstadt 36 4051 Basel Switzerland For the attention of Alexander Mayweg, Member of the Board

25th February 2019

Dear Sirs,



RE. Collaboration and Option Agreement between Cancer Research Technology Limited ("CRT"), Institute of Cancer Research: Royal Cancer Hospital ("ICR") and Monte Rosa Therapeutics AG ("Company") dated 10th April 2018 (the "Collaboration and Option Agreement")

The Parties entered into a License Agreement dated 26 April 2018 between CRT, ICR and the Company which was amended by Amendment 1 dated the same date as this COA Amendment 1 (as defined below). The Parties wish to modify the Collaboration and Option Agreement pursuant to this letter of amendment ("COA Amendment 1"). Unless indicated otherwise, all capitalised terms used in this COA Amendment 1 shall have the meanings given to such terms in the Collaboration and Option Agreement.

CRT, the ICR and the Company hereby agree to amend the Collaboration and Option Agreement, as of the date of this COA Amendment 1, as follows:

- 1. Section 1 shall be amended to include the following definition:
 - "[***]" has the meaning given in the License Agreement.
- 2. Schedule 2 shall be amended as follows:
 - a) Section titled "[***]" shall be amended to include the following:

"[***]

- [***]"
- b) Section titled "[***]" bullet point 7 shall be deleted and bullet point 8 shall be replaced from:

"…

• [***]

Registered address: Cancer Research Technology Ltd, Angel Building, 407 St John Street, London EC1V 4AD.
Registered In England (1626049). VAT registration number GB788 138678.
A wholly-owned subsidiary of Cancer Research UK, registered charity in England and Wales (1089464), Scotland (SC041666) and the Isle of Man (1103).

• [***]" to: "...

c) Point 5 of the Section titled "[***]" shall be deleted and replaced with the following:

"[***]"

d) Sub-section titled "[***]" of Section titled "Proposed Science Work Plan and Key Activities" shall be deleted and replaced with the following:

[***]	[***]
[***]	[***]
[***]	[***]
[***]	[***]

[***]

- e) Fourth Row labelled "[***]" under Timeline chart set forth under section titled "[***]:" shall be deleted as follows:

 [***]
- f) Sub-section titled "[***]" of Activity table set forth under Section titled "[***]" shall be deleted and replaced with the following:

 [***]

Save as expressly varied by this letter of amendment to the Collaboration and Option Agreement, all other terms of the Collaboration and Option Agreement remain the same. However, if there are any inconsistencies between the terms of this COA Amendment 1 and the provisions of the Collaboration and Option Agreement, then this COA Amendment 1 shall prevail.

This COA Amendment 1 shall be governed by and construed in accordance with the laws of England and Wales and the Parties agree to submit to the exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this COA Amendment 1.

Please sign, and add the date of signature, below conforming the and the Company's agreement to be bound by the terms of this COA Amendment d return three signed COA Amendment 1's to the undersigned.

/s/ Illegible March 6, 2019					
For and on behalf of					
Cancer Research Technology Limited					
Countersigned by The Institute of Cancer Research: Royal Cancer Hospital					
Signature: /s/ Illegible	Date: March 15, 2019				
N	Tide Disease				
Name: Illegible	Title: Director				
Countersigned by Monte Rosa Therapeutics AG					
Signature: /s/ Alexander Mayweg	Date: March 26, 2019				
Name: Alexander Mayweg	Title: Board Director				
	2				

Yours faithfully,



Cancer Research Technology 2 Redman Place London E20 1JQ United Kingdom T +44 (0)20 3469 6300 www.cancertechnology.com

The Institute of Cancer Research: Royal Cancer Hospital 123 Old Brompton Road London SW7 3RP United Kingdom For the attention of [***]

Monte Rosa Therapeutics AG Aeschenvorstadt 36 4051 Basel Switzerland For the attention of Alexander Mayweg, Member of the Board

20th January 2020

Dear Sirs,

RE. Collaboration and Option Agreement between Cancer Research Technology Limited ("CRT"), Institute of Cancer Research: Royal Cancer Hospital ("ICR") and Monte Rosa Therapeutics AG ("Company") dated 10th April 2018 and amended by COA Amendment 1 between CRT, ICR and Company dated 25th February 2019 (together the "Collaboration and Option Agreement")

The Parties wish to modify the Collaboration and Option Agreement pursuant to this letter of amendment ("COA Amendment 2"). Unless indicated otherwise, all capitalised terms used in this COA Amendment 2 shall have the meanings given to such terms in the Collaboration and Option Agreement.

CRT, the ICR and the Company hereby agree to amend the Collaboration and Option Agreement, as of the date of this COA Amendment 2, as follows:

1. Schedule 2 shall be deleted and replaced with the new Schedule 2 set forth in Annex 1 of this COA Amendment 2.

Save as expressly varied by this letter of amendment to the Collaboration and Option Agreement, all other terms of the Collaboration and Option Agreement remain the same. However, if there are any inconsistencies between the terms of this COA Amendment 2 and the provisions of the Collaboration and Option Agreement, then this COA Amendment 2 shall prevail.

This COA Amendment 2 shall be governed by and construed in accordance with the laws of England and Wales and the Parties agree to submit to the
exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this COA Amendment 2

Confidential

Please sign, and add the date of signature, below confirming the ICR's and the Company's agreement to be bound by the terms of this COA Amendment 2 and return three signed CO A Amendment 2's to the undersigned.

Yours faithfully,

/s/ Illegible

For and on behalf of

Cancer Research Technology Limited

Countersigned by The Institute of Cancer Research: Royal Cancer Hospital

Signature:	/s/ Illegible	Date:	March 2, 2020		
Name:	Illegible	Title:	Director		
Countersigned by Monte Rosa Therapeutics AG					
Signature:	/s/ Alexander Mayweg	Date:	March 2, 2020		
Name:	Alexander Mayweg	Title:	Board Member		

Annex 1

SCHEDULE 2

PROGRAMME

[***]

The Institute of Cancer Research; Royal Cancer Hospital 123 Old Brompton Road London SW7 3RP United Kingdom For the attention of [***]

Monte Rosa Therapeutics AG Aeschenworstadt 36 4051 Basel Switzerland For the attention of Markus Warmuth, Chief Executive Officer

18th June 2020

Dear Markus Warmuth and [***]

RE. Collaboration and Option Agreement between Cancer Research Technology Limited ("CRT"). Institute of Cancer Research: Royal Cancer Hospital ("ICR") and Monte Rosa Therapeutics AG ("Company") dated 10th April 2018 and amended by COA Amendment 1 between CRT, ICR and Company dated 25th February 2019 and COA Amendment 2 between CRT, ICR and Company dated 20th January 2020 (together the "Collaboration and Option Agreement")

The Parties wish to modify the Collaboration and Option Agreement pursuant to this letter of amendment ("COA Amendment 3") Unless indicated otherwise, all capitalised terms used in this COA Amendment 3 shall have the meanings given to such terms In the Collaboration and Option Agreement.

CRT, the ICR and the Company hereby agree to amend the Collaboration and Option Agreement, as of the date of this COA Amendment 3, as follows:

- The definition of "Services Expiry Date" in Clause 1.1 shall be deleted and replaced with the following'
 "Services Expiry Date" means 31st December 2020.
- 2. Clause 2.2(h) shall be deleted and replaced with the following:
 - "(h) The chairman of the meetings of the JSC shall alternate between each JSC meeting between the ICR Lead on the one hand, and the Company Lead on the other. The "ICR Lead" shall be [***] or, in the event that [***] resigns as the ICR Lead or is no longer affiliated with CRT and the ICR, then the ICR shall appoint an ICR employee as replacement ICR Lead, and shad be entitled to remove any ICR Lead appointed by it and to appoint any person to fill a vacancy arising from the removal or retirement of such ICR Lead, in each case, subject to the consent of the Company and CRT, not to be unreasonably withheld. The "Company Lead" shall be [***], or in the event that [***] is no longer affiliated with the Company, then the Company shall appoint a Company employee as a replacement Company Lead, and shall be entitled to remove any Company lead appointed by It and to appoint any person to fill a vacancy arising from the removal or retirement of such Company Lead, m each case, subject to the content of CRT and the ICR, not to be unreasonably withheld. The chairman shall have no casting vote, except as expressly provided in Clause 2.2(i)"

- 3. Clause 2.2(i) shall be deleted and replaced with the following:
 - "(i) The JSC shall have only the powers expressly assigned to it in Clause 2.1, and shall have no power to amend, modify, or waive compliance with this Agreement. Decisions of the JSC shall be made by unanimous agreement of the Members present, with the Company's appointed Members collectively having one (1) vote, and CRT's and ICR's appointed Members collectively having one (1) vote. No vote of the JSC may be taken unless at least one (1) of each Party's representatives is present for the vote. If the JSC cannot reach consensus with regard to any matter within its decision-making authority within [***] after such matter has been brought to the JSC's attention, then [***]."
- 4. For the purposes of Clause 2.2(j), the chairman's delegate for preparing and circulating minutes pursuant to Clausa 2.2(j)(c) shall be (i) the ICR Lead or the ICR Lead's delegate for JSC meetings chaired by the Company Lead, and (ii) the Company Lead or the Company Lead's delegate for JSC meetings chaired by the ICR Lead.
- 5. Clause 3.2 shall be deleted and replaced with Hie following:
 - "3.2 in consideration for the payments to be made pursuant to Clause 4.1, the ICR shall engage the researchers set forth under the section entitled "Researchers" (the "*Researchers*") in Schedule 2 for the period of time set forth therein, to be appointed as soon as practicable after the Effective Date, to work at the ICR as employees of the ICR, to carry out under the supervision of the ICR Lead, those parts of the Programme allocated to the ICR hereunder. The Company shall provide the ICR with such assistance with the selection and recruitment of the Researchers as the ICR may reasonably request, and the ICR shall promptly notify the Company of the appointment of each Researcher, including the effective date of each such appointment."

Save as expressly varied by this letter of amendment to the Collaboration and Option Agreement, all other terms of the Collaboration and Option Agreement remain the same. However, if there are any inconsistencies between the terms of this COA Amendment 3 and the provisions of the Collaboration and Option Agreement, then this COA Amendment 3 shall prevail.

This COA Amendment 3 shall be governed by and construed in accordance with the laws of England and Wales and the Parties agree to submit to the exclusive jurisdiction of the English courts in respect of any dispute arising out of or in connection with this COA Amendment 3.

Please sign, and add the date of signature, below confirming the ICR's and the Company's agreement to be bound by the terms of this COA Amendment 3 and return signed COA Amendment 3 to the undersigned.					
Yours faithfully,					
/s/ Illegible					
For and on behalf of Cancer Research Technology Limited					
Countersigned by The Institute of Cancer Research; Royal Cancer Hospital					
Signature: /s/ Illegible	Date:	July 23, 2020			
Name: Illegible	Position:	Director			
Countersigned by Monte Rosa Therapeutics AG					
Signature: /s/ Markus Warmuth	Date:	August 21, 2020			
Name: Markus Warmuth	Position:	CEO			